UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

(I.R.S. Employer Identification No.) 2910 Seventh Street, Berkeley, California 94710 (510) 204-7200 (Address of principal executive offices, including zip code) (Telephone Number) Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes 🗵 No 🗖 Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes 🗵 No 🗆 Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act). Large accelerated filer \Box Accelerated filer \boxtimes Non-accelerated filer \Box Smaller reporting company \Box (Do not check if a smaller reporting company) Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act of 1934). Yes 🗆 No 🗵 Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date. Class Outstanding at August 2, 2011 Common Shares, U.S. \$0.0075 par value 32,554,155

X QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the quarterly period ended June 30, 2011

or

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 For the transition period from to

Commission File No. 0-14710

XOMA Ltd.

(Exact name of registrant as specified in its charter)

52-2154066

Bermuda

(State or other jurisdiction of incorporation or organization)

XOMA Ltd. FORM 10-Q TABLE OF CONTENTS

PART I FINANCIAL INFORMATION

Condensed Consolidated Financial Statements (unaudited) Item 1. Condensed Consolidated Balance Sheets as of June 30, 2011 and December 31, 2010 1 Condensed Consolidated Statements of Operations for the Three and Six Months Ended June 30, 2011 and 2010 2 Condensed Consolidated Statements of Cash Flows for the Six Months Ended June 30, 2011 and 2010 3 Notes to Condensed Consolidated Financial Statements 4 Management's Discussion and Analysis of Financial Condition and Results of Operations Item 2. 14 Quantitative and Qualitative Disclosures About Market Risk Item 3. 22 Item 4. Controls and Procedures 23 PART II **OTHER INFORMATION** 23 Item 1. Legal Proceedings Item 1A. **Risk Factors** 24 Unregistered Sales of Equity Securities and Use of Proceeds 41 Item 2. Item 3. Defaults Upon Senior Securities 41 Item 4. 41 Reserved Item 5. Other Information 41 Item 6. Exhibits 42 **Signatures** 43

i

Page 1

PART I - FINANCIAL INFORMATION

ITEM 1. CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited)

XOMA Ltd. CONDENSED CONSOLIDATED BALANCE SHEETS (in thousands, except share and per share amounts)

	June 30, 2011 (unaudited)		2011 201	
ASSETS				
Current assets:				
Cash and cash equivalents	\$	51,157	\$	37,304
Trade and other receivables, net		11,059		20,864
Prepaid expenses and other current assets		807		712
Total current assets		63,023		58,880
Property and equipment, net		14,410		14,869
Other assets		2,165		503
Total assets	\$	79,598	\$	74,252
LIABILITIES AND SHAREHOLDERS' EQUITY				
Current liabilities:				
Accounts payable	\$	5,043	\$	3,581
Accrued liabilities		7,087		10,658
Deferred revenue		5,982		17,044
Warrant liabilities		1,395		4,245
Total current liabilities		19,507		35,528
Deferred revenue – long-term		8,665		1,086
Interest bearing obligations – long-term		27,031		13,694
Other long-term liabilities		257		353
Total liabilities		55,460		50,661
Shareholders' equity:				
Preference shares, \$0.05 par value, 1,000,000 shares authorized				
Series B, 8,000 designated, 0 and 2,959 shares issued and outstanding at June 30, 2011 and December 31, 2010, respectively		-		1
Common shares, \$0.0075 par value, 92,666,666 shares authorized, 32,200,731 and 28,491,318 shares outstanding at June 30, 2011				
and December 31, 2010, respectively		241		214
Additional paid-in capital		891,672		876,686
Accumulated comprehensive income		1		-
Accumulated deficit		(867,776)	_	(853,310)
Total shareholders' equity		24,138		23,591
Total liabilities and shareholders' equity	\$	79,598	\$	74,252

The accompanying notes are an integral part of these condensed consolidated financial statements.

(Note 1) The condensed consolidated balance sheet as of December 31, 2010 has been derived from the audited financial statements as of that date included in the Company's Annual Report on Form 10-K for the year ended December 31, 2010.



XOMA Ltd. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (unaudited) (in thousands, except per share amounts)

	Т	Three months ended June 30,			Six months ended June 30,		
	2011 2010		2010	2011		2010	
Revenues:							
License and collaborative fees	\$	6,039	\$ 150	\$ 11,86	6 \$	339	
Contract and other revenue		10,467	5,481	20,12	8	12,292	
Royalties		19	311	12	6	513	
Total revenues		16,525	5,942	32,12	0	13,144	
Operating expenses:							
Research and development		18,281	19,346	35,62	.8	36,933	
Selling, general and administrative		6,113	5,026	11,48	3	10,579	
Total operating expenses		24,394	24,372	47,11	1	47,512	
Loss from operations		(7,869)	(18,430)) (14,99	1)	(34,368)	
Other income (expense):							
Investment and interest income		18	6	2	8	9	
Interest expense		(634)	(90)) (1,16	6)	(177)	
Other income (expense)		355	2,950	1,67	8	(2,813)	
Net loss before taxes		(8,130)	(15,564)) (14,45	1)	(37,349)	
Provision for income tax expense		-	(16)) (1	5)	(16)	
Net loss	\$	(8,130)	\$ (15,580)) \$ (14,46	6) \$	(37,365)	
Basic and diluted net loss per common share	\$	(0.27)	\$ (0.93)) <u>\$ (0.4</u>	<u>9) </u> \$	(2.24)	
Shares used in computing basic and diluted net loss per common share		29,889	16,695	29,53	6	16,695	

The accompanying notes are an integral part of these condensed consolidated financial statements.

XOMA Ltd. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (unaudited) (in thousands)

	Six Months En	ided June 30,
	2011	2010
Cash flows from operating activities:		
Net loss	\$ (14,466)	\$ (37,365
Adjustments to reconcile net loss to net cash used in operating activities:	2,702	2.05
Depreciation	2,702	2,950
Common shares contribution to 401(k)	1,046	905
Share-based compensation expense	4,056	2,08
Accrued interest on interest bearing obligations Revaluation of warrant liabilities	495	164
	(2,850)	(1,69)
Warrant modification expense	-	4,50
Amortization of discount on long-term debt Unrealized loss on foreign currency exchange	661	
	1,876	
Unrealized gain on foreign exchange options	(156)	17
Other non-cash adjustments	11	12
Changes in assets and liabilities affecting cash: Receivables	9.805	(1.29)
Prepaid expenses and other assets	- ,	(1,285
Accounts payable and accrued liabilities	(1,601) (2,505)	(863)
Deferred revenue	(12,382)	(1,979
Other liabilities	(12,382) (25)	(1,979)
Net cash used in operating activities	(13,333)	(32,510
Cash flows from investing activities:		
Net purchase of property and equipment	(2,253)	(254
Net cash used in investing activities	(2,253)	(254
Cash flows from financing activities:		
Proceeds from issuance of long-term debt	20,102	
Proceeds from issuance of common shares	9,910	25,456
Payment for modification of warrants		(4,500
Net cash provided by financing activities	30,012	20,950
Net cash provided by miancing activities		20,930
Effect of exchange rate changes on cash	(573)	
Net increase in cash and cash equivalents	14,426	(11,808
Cash and cash equivalents at the beginning of the period	37,304	23,909
Cash and cash equivalents at the end of the period	\$ 51,157	\$ 12,10
Supplemental Cash Flow Information:		
Cash paid for income taxes	\$ 15	\$ 10
A	<u>\$ 15</u>	φ 10
Non-cash investing and financing activities:	¢ (0.000)	¢
Discount on long-term debt	<u>\$ (8,899)</u>	\$
Issuance and extinguishment of warrant liabilities	\$ -	\$ 1,76
Interest added to principal balance on Novartis note	\$ 170	\$ 164

The accompanying notes are an integral part of these condensed consolidated financial statements.

XOMA Ltd. NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (unaudited)

1. Description of Business

XOMA Ltd. ("XOMA" or the "Company"), a Bermuda company, is a biopharmaceutical company focused on the discovery, development and manufacture of therapeutic antibodies designed to treat inflammatory, autoimmune, infectious and oncological diseases. The Company's products are presently in various stages of development and are subject to regulatory approval before they can be commercially launched.

2. Basis of Presentation and Significant Accounting Policies

Basis of Presentation

The condensed consolidated financial statements include the accounts of XOMA and its subsidiaries. All intercompany accounts and transactions were eliminated during consolidation. The unaudited financial statements have been prepared in accordance with accounting principles generally accepted in the United States for interim financial information and with the instructions to Form 10-Q. These financial statements and related disclosures have been prepared with the assumption that users of the interim financial information have read or have access to the audited financial statements for the preceding fiscal year. Accordingly, these statements should be read in conjunction with the audited consolidated financial statements and related notes included in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2010, filed with the U.S. Securities and Exchange Commission ("SEC") on March 10, 2011.

In the opinion of management, the unaudited condensed consolidated financial statements include all adjustments, consisting only of normal recurring adjustments, which are necessary to present fairly the Company's consolidated financial position as of June 30, 2011, the consolidated results of the Company's operations for the three and six months ended June 30, 2011 and 2010, and the Company's cash flows for the six months ended June 30, 2011 and 2010. The interim results of operations are not necessarily indicative of the results that may occur for the full fiscal year or future periods.

Use of Estimates

The preparation of financial statements in conformity with accounting principles generally accepted in the United States requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue and expenses, and related disclosures. On an on-going basis, management evaluates its estimates including, but not limited to, those related to revenue recognition, long-lived assets, warrant liabilities, derivative instruments and share-based compensation. The Company bases its estimates on historical experience and on various other market-specific and other relevant assumptions that are believed to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ significantly from these estimates.

Concentration of Risk

Cash equivalents and receivables are financial instruments, which potentially subject the Company to concentrations of credit risk, as well as liquidity risk for certain cash equivalents such as money market funds. The Company has not encountered such issues during 2011.

The Company has not experienced any significant credit losses and does not generally require collateral on receivables. For the six months ended June 30, 2011, two customers represented 57% and 38% of total revenue and 54% and 45% of the accounts receivable balance.

For the six months ended June 30, 2010, two different customers represented 63% and 23% of total revenues. As of December 31, 2010, there were receivables outstanding from two customers representing 72% and 23% of the accounts receivable balance.

Recent Accounting Pronouncements

In June of 2011, Accounting Standards Codification Topic 220, *Comprehensive Income* was amended to increase the prominence of items reported in other comprehensive income. Accordingly, a company can present all nonowner changes in stockholders' equity either in a single continuous statement of comprehensive income or in two separate but consecutive statements. The Company plans to adopt this guidance as of January 1, 2012 on a retrospective basis and does not expect the adoption thereof to have a material effect on the Company's consolidated financial statements.

In May of 2011, Accounting Standards Codification Topic 820, *Fair Value Measurement* was amended to develop common requirements for measuring fair value and for disclosing information about fair value measurements in accordance with U.S. generally accepted accounting principles and International Financial Reporting Standards. The Company plans to adopt this guidance as of January 1, 2012 on a prospective basis and does not expect the adoption thereof to have a material effect on the Company's consolidated financial statements.

In March of 2010, Accounting Standards Codification Topic 605, *Revenue Recognition* was amended to define a milestone and clarify that the milestone method of revenue recognition is a valid application of the proportional performance model when applied to research or development arrangements. Accordingly, a company can make an accounting policy election to recognize a payment that is contingent upon the achievement of a substantive milestone in its entirety in the period in which the milestone is achieved. This guidance was adopted effective January 1, 2011 on a prospective basis and did not have a material effect on the Company's consolidated financial statements.

3. Condensed Consolidated Financial Statement Detail

Comprehensive Loss

Unrealized gain on the Company's available-for-sale securities is included in accumulated comprehensive income. Comprehensive loss and its components for the three and six months ended June 30, 2011 and 2010 was as follows (in thousands):

	Т	Three Months Ended June 30,			Six Months Ended June 30			June 30,
		2011		2010 2011		2010		
Net loss	\$	(8,130)	\$	(15,580)	\$	(14,466)	\$	(37,365)
Unrealized gain on available-for-sale securities		1		-		1		-
Comprehensive loss	\$	(8,129)	\$	(15,580)	\$	(14,465)	\$	(37,365)

Net Loss Per Common Share

Basic and diluted net loss per common share is based on the weighted average number of common shares outstanding during the period.

Potentially dilutive securities are excluded from the calculation of earnings per share if their inclusion is anti-dilutive. The following table shows the total outstanding securities considered anti-dilutive and therefore excluded from the computation of diluted net loss per share (in thousands):

	Three Months En	ded June 30,	Six Months Ended June 30,		
	2011 2010		2011	2010	
Options for common shares	3,833	1,478	3,711	1,223	
Convertible preference shares	17	254	135	254	
Warrants for common shares	1,607	1,607	1,607	1,607	
Total	5,457	3,339	5,453	3,084	

For the three and six months ended June 30, 2011 and 2010, all outstanding securities were considered anti-dilutive, and therefore the calculations of basic and diluted net losses per share were the same.

Cash and Cash Equivalents

At June 30, 2011, cash equivalents consisted of demand deposits, money market funds and treasury bonds with maturities of less than 90 days at the date of purchase. At December 31, 2010, cash equivalents consisted of demand deposits, money market funds and repurchase agreements with maturities of less than 90 days at the date of purchase.

The treasury bond held in cash and cash equivalents is classified as an available-for-sale security and is stated at fair value, with unrealized gains and losses, net of tax, if any, reported in other comprehensive income. The estimate of fair value is based on publicly-available market information. Realized gains and losses and declines in value judged to be other-than-temporary on available-for-sale securities are included in investment and interest income. The cost of investments sold is based on the specific identification method. Interest and dividends on securities classified as available-for-sale are also included in investment and interest income.

Cash and cash equivalent balances were recorded at fair value as follows as of June 30, 2011 and December 31, 2010 (in thousands):

		June 30, 2011					
	Cost Basis	Unrealized Gains	Unrealized Losses	Estimated Fair Value			
Cash	\$ 15.	548 \$	- \$	- \$ 15,548			
Cash equivalents	35	608	1	- 35,609			
Total cash and cash equivalents	\$ 51.	156 \$	1 \$	- \$ 51,157			
		Decen	nber 31, 2010				
	Cost Basis	Unrealized Gains	Unrealized Losses	Estimated Fair Value			
Cash	\$ 29,5	36 \$	- \$	- \$ 29,536			
Cash equivalents	7,7	68	-	- 7,768			
Total cash and cash equivalents	\$ 37,3	0.1 0	0	- \$ 37,304			

Foreign Exchange Options

The Company holds debt and may incur expenses denominated in foreign currencies, which exposes it to market risk associated with foreign currency exchange rate fluctuations between the U.S. dollar and the Euro. The Company is required to make principal and accrued interest payments in Euros on its \in 15.0 million loan from Les Laboratoires Servier ("Servier") (refer to Note 6 below). In order to manage its foreign currency exposure related to these payments, in May of 2011, the Company entered into two foreign exchange option contracts to buy \in 15.0 million and \in 1.5 million on January 2016 and January 2014, respectively. By having these option contracts in place, the Company's foreign exchange rate risk is reduced if the U.S. dollar weakens against the Euro. However, if the U.S. dollar strengthens against the Euro, the Company is not required to exercise these options, but will not receive any refund on premiums paid.

Upfront premiums paid on these foreign exchange option contracts totaled \$1.5 million. The fair values of these option contracts are re-valued at each reporting period and are estimated based on pricing models using readily observable inputs from actively quoted markets. The fair values of these option contracts are included in other assets on the condensed consolidated balance sheet and changes in fair value on these contracts are included in other income (expense) on the condensed consolidated statements of operations. The foreign exchange options were revalued at June 30, 2011 and had an aggregate fair value of \$1.7 million, and the Company recognized a gain of \$0.2 million related to the revaluation for the three and six months ended June 30, 2011.

Receivables

Receivables consisted of the following at June 30, 2011 and December 31, 2010 (in thousands):

	June 30, 2011	De	ecember 31, 2010
Trade receivables, net	\$ 8,257	\$	20,309
Other receivables	2,802		555
Total	\$ 11,059	\$	20,864

Property and Equipment

Property and equipment consisted of the following at June 30, 2011 and December 31, 2010 (in thousands):

	ne 30, 2011	December 31, 2010		
Furniture and equipment	\$ 32,827	\$	31,700	
Buildings, leasehold and building improvements	21,481		21,463	
Construction-in-progress	943		203	
Land	 310		310	
	55,561		53,676	
Less: Accumulated depreciation and amortization	 (41,151)		(38,807)	
Property and equipment, net	\$ 14,410	\$	14,869	

Depreciation expense was \$1.4 million and \$2.7 million for the three and six months ended June 30, 2011, respectively, compared with \$1.4 million and \$3.0 million, respectively, for the same periods in 2010.

Accrued Liabilities

Accrued liabilities consisted of the following at June 30, 2011 and December 31, 2010 (in thousands):

	ıne 30, 2011	December 31, 2010		
Accrued payroll and other benefits	\$ 2,709	\$	2,752	
Accrued management incentive compensation	2,253		4,982	
Accrued professional fees	709		1,020	
Accrued clinical trial costs	224		1,020	
Other	1,192		884	
Total	\$ 7,087	\$	10,658	

4. Fair Value Measurements

Fair value is defined as the price that would be received from selling an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. When determining the fair value measurements for assets and liabilities required to be recorded at fair value, the Company considers the principal or most advantageous market in which it would transact and assumptions that market participants would use when pricing the asset or liability, such as inherent risk, transfer restrictions and risk of nonperformance.

A fair value hierarchy was established which requires an entity to maximize the use of observable inputs and minimize the use of unobservable inputs when measuring fair value. A financial instrument's categorization within the fair value hierarchy is based upon the lowest level of input that is significant to the fair value measurement. The three levels of inputs that may be used to measure fair value are as follows:

- · Level 1: Quoted prices in active markets for identical assets or liabilities;
- Level 2: Inputs other than Level 1 that are observable, either directly or indirectly, such as quoted prices in active markets for similar assets or liabilities, quoted prices for identical or similar assets or liabilities in markets that are not active, or other inputs that are observable or can be corroborated by readily observable market data for substantially the full term of the assets or liabilities; or
- Level 3: Unobservable inputs that are supported by little or no market activity and that are significant to the fair value of the assets or liabilities.

The following tables set forth the Company's fair value hierarchy for its financial assets (cash equivalents) and liabilities measured at fair value on a recurring basis as of June 30, 2011 and December 31, 2010.



Financial assets and liabilities carried at fair value as of June 30, 2011 and December 31, 2010 were classified as follows (in thousands):

		Fair Value N	Aeasurements at June	30, 2011 Using		
		Quoted Prices in Active Markets for Identical Assets (Level 1)	Significant Other Observable Inputs (Level 2)	Significant Unobservable Inputs (Level 3)		Total
Assets: Treasury Bond ⁽¹⁾	\$	10,517	\$ -	\$ -	\$	10,517
Money market funds ⁽¹⁾	φ	25,092	р -	ۍ د	Ф	25,092
Foreign exchange options		25,092	1,657	-		1,657
Total	\$	35,609		\$	\$	37,266
Liabilities:						
Warrant liabilities	2		\$	\$ 1,395	\$	1,395
	Quoted Active M Identic	air Value Measur Prices in [arkets for al Assets	ements at December 3 Significant Other Observable Inputs (Level 2)	1, 2010 Using Significant Unobservable Inputs (Level 3)		Total
Assets:						
Repurchase agreements ⁽¹⁾	\$	1,428 \$		- \$ -	\$	1,428
Money market funds (1)		6,340				6,340
Total	\$	7,768 \$		- \$ -	\$	7,768
Liabilities:						
Warrant liabilities	<u>\$</u>	- \$		- \$ 4,245	\$	4,245

(1) Included in cash and cash equivalents

The fair value of the foreign exchange options at June 30, 2011 was determined using readily observable market inputs from actively quoted markets obtained from various third party data providers. These inputs, such as spot rate, forward rate and volatility have been derived from readily observable market data, meeting the criteria for Level 2 in the fair value hierarchy.

The fair value of the warrant liabilities at June 30, 2011 and December 31, 2010 was determined using the Black-Scholes Model, which requires inputs such as the expected term of the warrants, share price volatility and risk-free interest rate. These inputs are subjective and generally require significant analysis and judgment to develop.

The fair value of the warrant liabilities was estimated using the following range of assumptions at June 30, 2011 and December 31, 2010:

	June 30, 2011	December 31, 2010
Expected volatility	101.3 - 102.3%	93.5 - 94.9%
Risk-free interest rate	0.8%	2.0%
Expected term	3.5 - 3.6 years	3.9 - 4.1 years

The following table provides a summary of changes in the fair value of the Company's Level 3 financial liabilities for the six months ended June 30, 2011 (in thousands):

	Lia	Warrant abilities at ne 30, 2011
Balance at December 31, 2010	\$	4,245
Net decrease in fair value of warrant liabilities on revaluation		(2,850)
Balance at June 30, 2011	\$	1,395

For the three and six months ended June 30, 2011, the Company recognized net decreases of \$0.5 million and \$2.9 million, respectively, in the estimated fair value of the warrant liabilities resulting in recognized gains in the other income (expense) line of the condensed consolidated statements of operations.

For the three and six months ended June 30, 2010, the Company recognized net decreases of \$3.0 million and \$1.7 million, respectively, in the estimated fair value of the warrant liabilities resulting in recognized gains in the other income (expense) line of the condensed consolidated statements of operations.

5. Licensing, Collaborative and Other Arrangements

Servier

In December of 2010, the Company entered into a license and collaboration agreement with Servier, to jointly develop and commercialize XOMA 052 in multiple indications, which provided for a non-refundable upfront payment of \$15.0 million that was received by the Company in January of 2011. The upfront payment will be recognized over the estimated eight month period that the initial group of deliverables will be provided to Servier. The Company recognized \$5.5 million and \$11.0 million in revenue relating to this upfront payment during the three and six months ended June 30, 2011, respectively. In addition, the Company received a loan of €15.0 million, which was fully funded in January of 2011, with the proceeds converting to \$19.5 million at the date of funding (refer to Note 6 below). Also, the Company retains development and commercialization rights for Behcet's uveitis and other inflammatory and oncology indications in the U.S. and Japan, and an option to reacquire rights to diabetes and cardiovascular related diseases, as well as the first \$50.0 million of future XOMA 052 global clinical development and chemistry and manufacturing controls expenses and 50% of further expenses for the Behcet's uveitis indication, which is expected to advance into Phase 3 development in 2012. For the three and six months ended June 30, 2011, the Company recorded revenue of \$10.1 million and \$17.7 million, respectively, under this agreement, which included the revenue relating to the upfront payment.

Merck/Schering-Plough

In May of 2006, the Company entered into a fully funded collaboration agreement with Schering-Plough Research Institute, a division of Schering Corporation, now a subsidiary of Merck ("Merck/Schering-Plough") for therapeutic monoclonal antibody discovery and development. Under the agreement, Merck/Schering-Plough made up-front, annual maintenance and milestone payments to the Company, funded its research and development activities related to the agreement and would have paid royalties on sales of products resulting from the collaboration. During the collaboration, the Company discovered therapeutic antibodies against multiple targets selected by Merck/Schering-Plough using multiple human antibody phage display libraries, optimized antibodies through affinity maturation or other protein engineering, used the Company's proprietary HETM technology to humanize antibodies generated by hybridoma techniques, performed preclinical studies to support regulatory filings, developed cell lines and production processes and produced antibodies for initial clinical trials. Merck/Schering-Plough selected the first target at the inception of the agreement and, in December of 2006, exercised its right to initiate the additional discovery and development programs. In January of 2011, the Company successfully completed the contract services it had agreed to perform under the collaboration agreement with Merck/Schering-Plough.

6. Long-Term Debt and Other Financings

Long-Term Debt

Novartis Note

In May of 2005, the Company executed a secured note agreement with Novartis (then Chiron Corporation), which is due and payable in full in June of 2015. Under this note agreement, the Company borrowed semi-annually to fund up to 75% of the Company's research and development and commercialization costs under its collaboration arrangement with Novartis, not to exceed \$50.0 million in aggregate principal amount. Interest on the principal amount of the loan accrues at six-month LIBOR plus 2%, which was equal to 2.39% at June 30, 2011, and is payable semi-annually in June and December of each year. At the Company's election, the semi-annual interest payments can be added to the outstanding principal amount, in lieu of a cash payment, as long as the aggregate principal amount does not exceed \$50.0 million. The Company has made this election for all interest payments thus far. Loans under the note agreement are secured by the Company's interest in the collaboration with Novartis, including any payments owed to it thereunder.

At June 30, 2011 and December 31, 2010, the outstanding principal balance under this note agreement was \$13.9 million and \$13.7 million, respectively. Pursuant to the terms of the arrangement as restructured in November of 2008, the Company will not make any additional borrowings under the Novartis note. Due to the structure of the secured note agreement with Novartis and since there is no liquid market for this obligation, there is no practical method to estimate fair value of this long-term debt.

Servier Loan

In December of 2010, in connection with the license and collaboration agreement entered into with Servier (see footnote 5), the Company executed a loan agreement with Servier, which provided for an advance of up to \in 15.0 million. The loan was fully funded in January of 2011, with the proceeds converting to approximately \$19.5 million. The loan is secured by an interest in XOMA's intellectual property rights to all XOMA 052 indications worldwide, excluding certain rights in the U.S. and Japan. Interest is calculated at a floating rate based on a Euro Inter-Bank Offered Rate ("EURIBOR") and subject to a cap. The interest rate is reset semi-annually in January and July of each year. The interest rate for the initial interest period was 3.22%. The interest rate has been reset to 3.83% for six-month period from July 2011 through January 2012. Interest is payable semi-annually; however, the loan agreement provides for a deferral of interest payments over a period specified in the agreement. During the deferral period, accrued interest will be added to the outstanding principal amount for the purpose of interest calculation for the next six-month interest period. On the repayment commencement date, all unpaid and accrued interest shall be paid to Servier and thereafter, all accrued and unpaid interest shall be due and payable at the end of each six-month period. The loan matures in 2016; however, after a specified period prior to final maturity, the loan is to be repaid (i) at Servier's option, by applying up to a significant percentage of any milestone or royalty payments the Company's collaboration agreement and (ii) using a significant percentage of any upfront, milestone or royalty payments the Company receives from any third party collaboration or development partner for rights to XOMA 052 in the U.S. and/or Japan. In addition, the loan estored unrealized foreign exchange losses of \$0.5 million and \$2.1 million, respectively, related to the re-measurement of the loan as of June 30, 2011.

The loan has a stated interest rate lower than the market rate based on comparable loans held by similar companies, which represents additional value to the Company. The Company recorded this additional value as a discount to the face value of the loan amount, at its fair value of \$8.9 million. The fair value of this discount, which was determined using a discounted cash flow model, represents the differential between the stated terms and rates of the loan, and market rates. Based on the association of the loan with the collaboration arrangement, the Company recorded the offset to this discount as deferred revenue.

The loan discount is amortized under the effective interest method over the expected five-year life of the loan. The Company recorded non-cash interest expense of \$0.4 million and \$0.7 million during the three and six months ended June 30, 2011, respectively, resulting from the amortization of the loan discount. At June 30, 2011, the net carrying value of the loan was \$13.2 million.

The Company believes that realization of the benefit and the associated deferred revenue is contingent on the loan remaining outstanding over the five-year contractual term of the loan. If the Company were to stop providing service under the collaboration arrangement and the arrangement is terminated, the maturity date of the loan would be accelerated and a portion of measured benefit would not be realized. As the realization of the benefit is contingent, in part, on the provision of future services, the Company is recognizing the deferred revenue over the expected five-year life of the loan. The deferred revenue is amortized under the effective interest method, and the Company recorded \$0.4 million and \$0.7 million of related non-cash revenue during the three and six months ended June 30, 2011, respectively.

Interest Expense

Interest expense for the Novartis note and Servier loan are shown below (in thousands):

	Th	ree Months l	Endeo	l June 30,		Six Months E	June 30,	
		2011		2010		2011		2010
Interest expense					_			
Novartis note	\$	85	\$	83	\$	169	\$	165
Servier loan		538		-		980		-
Other		11		7		17		12
Total interest expense	\$	634	\$	90	\$	1,166	\$	177

Other Financings

ATM Agreements

In the third quarter of 2010, the Company entered into an At Market Issuance Sales Agreement (the "2010 ATM Agreement"), with Wm Smith & Co. and McNicoll, Lewis & Vlak LLC (the "Agents"), under which the Company could sell common shares from time to time through the Agents, as its agents for the offer and sale of the common shares, in an aggregate amount not to exceed the amount that could be sold under its registration statement on Form S-3 (File No. 333-148342) filed with the SEC on December 26, 2007. The Agents could sell the common shares by any method permitted by law deemed to be an "at the market" offering as defined in Rule 415 of the Securities Act of 1933, as amended (the "Securities Act"), including without limitation sales made directly on The NASDAQ Global Market, on any other existing trading market for the common shares or to or through a market maker. The Agents could also sell the common shares in privately negotiated transactions, subject to the Company's prior approval. The Company paid the Agents, collectively, a commission equal to 3% of the gross proceeds of the sales price of all common shares sold through them as sales agents under the 2010 ATM Agreement. From the inception of the 2010 ATM Agreement through May of 2011, the Company sold a total of 7,560,862 common shares incurred related to sales proceeds of \$34.0 million, including &21,386 common shares sold in 2011 for aggregate gross proceeds of \$4.4 million. Total offering expenses incurred related to sales under the 2010 ATM Agreement from inception to May of 2011 were \$1.0 million, including \$0.1 million incurred in 2011. In May of 2011, the 2010 ATM Agreement expired by its terms, and there will be no further issuances under this facility.

On February 4, 2011, the Company entered into an At Market Issuance Sales Agreement (the "2011 ATM Agreement"), with McNicoll, Lewis & Vlak LLC ("MLV"), under which the Company may sell common shares from time to time through MLV, as its agent for the offer and sale of the common shares, in an aggregate amount not to exceed the amount that can be sold under the Company's registration statement on Form S-3 (File No. 333-172197) filed with the SEC on February 11, 2011 and amended on March 10, 2011 and June 3, 2011, which was declared effective by the SEC on June 6, 2011. MLV may sell the common shares by any method permitted by law deemed to be an "at the market" offering as defined in Rule 415 of the Securities Act, including without limitation sales made directly on The NASDAQ Global Market, on any other existing trading market for the common shares or to or through a market maker. MLV may also sell the common shares in privately negotiated transactions, subject to the Company's prior approval. The Company will pay MLV a commission equal to 3% of the gross proceeds of the sales price of all common shares sold through them as sales agent under the 2011 ATM Agreement. From the inception of the 2011 ATM Agreement through June 30, 2011, the Company sold a total of 2,398,017 common shares under this agreement for aggregate gross proceeds of \$5.9 million, of which \$3.7 million of cash was received in June of 2011 with the remaining \$2.2 million of cash received in July of 2011. Total offering expenses incurred related to sales under the 2011 ATM Agreement from inception to June 30, 2011 were \$0.2 million.

7. Income Taxes

Income tax expense was not material for the three and six months ended June 30, 2011 or the comparable periods in 2010. The Company's effective tax rate will fluctuate from period to period due to several factors inherent in the nature of the Company's operations and business transactions. The factors that most significantly impact this rate include the variability of licensing transactions in foreign jurisdictions.

8. Share-Based Compensation

In December of 2010, the Board of Directors of the Company approved a company-wide grant of share options under the Company's 2010 Long Term Incentive and Share Award Plan ("LTIP") and, in the first quarter of 2011, the options for 1,430,840 shares became effective. 1,040,220 of these options were granted subject to shareholder approval of an increase in the number of shares available under the LTIP. On May 26, 2011, shareholder approval was obtained at the Company's annual general meeting of shareholders. A cumulative adjustment of \$1.3 million was recorded in the second quarter of 2011 to reflect the share-based compensation expense that would have been recorded from the closing share price and expected term on that date. The remaining assumptions included in the calculation were the same assumptions used for the second quarter option grants. A portion of the 2011 annual options granted include immediate vesting terms with the remainder of the options vesting monthly over two years for employees and one year for directors.



As of June 30, 2011, the Company had approximately 4,562,486 common shares reserved for future issuance under its share option plans and Employee Share Purchase Plan ("ESPP").

The following table shows total share-based compensation expense included in the condensed consolidated statements of operations for the three and six months ended June 30, 2011 and 2010 (in thousands):

	Tł	hree Months l	Ended	June 30,		June 30,		
		2011	2010		2011			2010
Research and development	\$	885	\$	629	\$	1,946	\$	1,086
Selling, general and administrative		1,399		490		2,110		995
Total share-based compensation expense	\$	2,284	\$	1,119	\$	4,056	\$	2,081

The valuation of share-based compensation awards is determined at the date of grant using the Black-Scholes Model. This model requires inputs such as the expected term of the option, expected volatility and risk-free interest rate. Further, the forfeiture rate also affects the amount of aggregate compensation. These inputs are subjective and generally require significant analysis and judgment to develop. While estimates of the expected term, volatility and forfeiture rate are derived primarily from the Company's historical data, the risk-free rate is based on the yield available on United States Treasury zero-coupon issues. The fair value of share-based awards was estimated based on the following weighted average assumptions for the three and six months ended June 30, 2011 and 2010:

	Three Months En	ded June 30,	Six Months End	ed June 30,
	2011	2010	2011	2010
Dividend yield	0%	0%	0%	0%
Expected volatility	87%	79%	87%	79%
Risk-free interest rate	1.76%	1.80%	1.87%	2.52%
Expected term	5.6 years	5.6 years	5.3 years	5.6 years

Share option activity for the six months ended June 30, 2011 was as follows:

	Options	Weighted Average Exercise Price	Weighted Average Remaining Contractual Life (in years)	Aggregate Intrinsic Value _(in thousands)
Options outstanding at December 31, 2010	2,331,450	\$ 25.36	7.71	\$ 99
Granted	1,806,040	5.24		
Forfeited, expired or cancelled	(85,988)	51.89		
Options outstanding at June 30, 2011	4,051,502	\$ 15.83	8.27	\$ 1
Options exercisable at June 30, 2011	2,308,248	\$ 22.33	7.63	\$

9. Share Capital

Series B Preference Shares

In December of 2003, the Company issued 2,959 Series B preference shares to Genentech, Inc. in repayment of \$29.6 million of the outstanding balance under a convertible subordinated debt agreement. Pursuant to the rights of the Series B preference shares, the holder of Series B preference shares was not entitled to receive any dividends on the Series B preference shares. The Series B preference shares ranked senior with respect to rights on liquidation, winding-up and dissolution of the Company to all classes of common shares. Upon any voluntary or involuntary liquidation, dissolution or winding-up of the Company, the holder of Series B preference shares would have been entitled to receive \$10,000 per Series B preference share (or \$29.6 million in the aggregate) before any distribution was made on the common shares. The holder of the Series B preference shares had no voting rights, except as required under Bermuda law.

The holder of Series B preference shares had the right to convert Series B preference shares into common shares at a conversion price equal to \$116.25 per common share, subject to adjustment in certain circumstances.

In April of 2011, the 2,959 Series B convertible preference shares were converted by Genentech into 254,560 common shares. The \$29.6 million liquidation preference associated with the Series B preference shares was eliminated as a result of this conversion.

10. Legal Proceedings, Commitments and Contingencies

On April 9, 2009, a complaint was filed in the Superior Court of Alameda County, California, in a lawsuit captioned Hedrick et al. v. Genentech, Inc. et al, Case No. 09-446158. The complaint asserts claims against Genentech, the Company and others for alleged strict liability for failure to warn, strict product liability, negligence, breach of warranty, fraudulent concealment, wrongful death and other claims based on injuries alleged to have occurred as a result of three individuals' treatment with RAPTIVA®. The complaint seeks unspecified compensatory and punitive damages. Since then, additional complaints have been filed in the same court, bringing the total number of pending cases to seventy five. The cases have been consolidated as a coordinated proceeding. All of the complaints allege the same claims and seek the same types of damages based on injuries alleged to have occurred as a result of the plaintiffs' treatment with RAPTIVA®. On July 15, 2011, the Court dismissed with prejudice one of the cases in this coordinated proceeding, White v. Genentech, Inc., et al, Case No. RG-09-484026. The Company's agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which the Company believes is applicable to these matters. The Company believes the claims against it to be without merit and intends to defend against them vigorously.

On August 4, 2010, a petition was filed in the District Court of Dallas County, Texas in a case captioned McCall v. Genentech, Inc., et al., No. 10-09544. The defendants filed a Notice of Removal to the United States District Court for the Northern District of Texas on September 3, 2010. The removed case is captioned McCall v. Genentech, Inc., et al., No. 3:10-cv-01747-B. The parties have fully briefed the plaintiff's Motion to Remand and are awaiting a final ruling from the Court. The petition asserts personal injury claims against Genentech, the Company, and others arising out of the plaintiff's treatment with RAPTIVA®. The petition alleges claims based on negligence, strict liability, misrepresentation and suppression, conspiracy, and actual and constructive fraud. The petition seeks compensatory damages and punitive damages in an unspecified amount. On June 6, 2011, the Court dismissed plaintiff's claims of negligent misrepresentation, fraud, and conspiracy. The Company's agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which the Company believes is applicable to this matter. The Company believes the claims against it to be without merit and intends to defend against them vigorously

On January 7, 2011, a complaint was filed in the United States District Court for the Northern District of Texas in a case captioned Massa v. Genentech, Inc., et al., No. 4:11CV70. On January 11, 2011, a complaint was filed in the United States District Court for the District of Massachusetts in a case captioned Sylvia, et al. v. Genentech, Inc., et al., No. 1:11-cv-10054-MLW. These two complaints allege the same claims against Genentech, the Company and others and seek the same types of damages as the complaints filed in the Superior Court of Alameda County, California referenced above. The Company's agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which the Company believes is applicable to these matters. The Company believes the claims against it to be without merit and intends to defend against them vigorously.

On April 8, 2011, four complaints were filed in the United States District Court for the Eastern District of Michigan. The cases are captioned: Muniz v. Genentech, et al., 5:11-cv-11489-JCO-RSW; Tifenthal v. Genentech, et al., 2:11-cv-11489-JCO-RSW; Tifenthal v. Genentech, et al., 2:11-cv-11489-JCO-RSW; Tifenthal v. Genentech, et al., 2:11-cv-11462-RHC-MKM. The complaints allege the same claims against Genentech, the Company and others and seek the same types of damages as the complaints filed in the Superior Court of Alameda County, California referenced above. All four cases have been transferred to the United States District Court for the Western District of Michigan. Genentech and the Company have filed Motions to Dismiss all four actions and are awaiting a decision from the Court. The Company's agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which the Company believes is applicable to this matter. The Company believes the claims against it to be without merit and intends to defend against them vigorously.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The accompanying discussion and analysis of our financial condition and results of operations are based upon our consolidated financial statements and the related disclosures, which have been prepared in accordance with accounting principles generally accepted in the United States. The preparation of these financial statements requires us to make estimates and assumptions that affect the reported amounts in our condensed consolidated financial statements and accompanying notes. On an on-going basis, we evaluate our estimates including, but not limited to, those related to terms of revenue recognition, long-lived assets, derivative instruments, warrant liabilities and share-based compensation. We base our estimates on historical experience and on various other market-specific and other relevant assumptions that we believe to be reasonable under the circumstances, the results of which form the basis for making judgments about the carrying values of assets and liabilities that are not readily apparent from other sources. Actual results may differ significantly from these estimates.

Overview

We are a leader in the discovery, development and manufacture of therapeutic antibodies designed to treat inflammatory, autoimmune, infectious and oncological diseases. Our proprietary development pipeline includes XOMA 052, an antibody that inhibits interleukin-1 beta ("IL-1 beta") which is expected to advance into Phase 3 development for the treatment of Behcet's uveitis in 2011 and Phase 2 development for cardiovascular disease in 2012; XOMA 3AB, a biodefense anti-botulism product candidate comprised of a combination, or cocktail, of antibodies; and preclinical antibody discovery programs in several indications, including autoimmune, cardio-metabolic, inflammatory, and oncological diseases. We have a fully integrated product development platform, extending from preclinical science, clinical development to scale-up development, and manufacturing.

In December of 2010, we entered into a license and collaboration agreement with Les Laboratoires Servier ("Servier"), to jointly develop and commercialize XOMA 052 in multiple indications. XOMA 052 is designed to inhibit the pro-inflammatory cytokine IL-1 beta that is believed to be a primary trigger of pathologic inflammation in multiple diseases.

Our biodefense initiatives currently include a \$65.0 million multiple-year contract funded by the National Institute of Allergy and Infectious Diseases ("NIAID"), a part of the National Institutes of Health ("NIH"), to support our ongoing development of anti-botulism antibody product candidates, of which the first, XOMA 3AB, is in a Phase 1 clinical trial. This contract is the third that NIAID has awarded us for the development of botulinum antitoxins and brings the program's total awards to nearly \$100.0 million. We also develop products with premier pharmaceutical companies including Novartis AG ("Novartis") and Takeda Pharmaceutical Company Limited ("Takeda").

We have a premier antibody discovery and development platform that incorporates a collection of antibody phage display libraries and proprietary Human EngineeringTM, affinity maturation, Bacterial Cell Expression ("BCE") and manufacturing technologies that enhance our ability and that of our collaboration and development partners to discover and develop new therapeutic antibodies. BCE is a key biotechnology for the discovery and manufacturing of antibodies and other proteins. To date, more than 50 pharmaceutical and biotechnology companies have signed BCE licenses, and a number of licensed product candidates are in clinical development. We continue to develop and commercialize additional antibody-related technologies including proprietary display technologies to enable antibody discovery and optimization. Our technologies have contributed to the success of marketed antibody products, including LUCENTIS[®] (ranibizumab injection) for wet age-related macular degeneration and CIMZIA[®] (certolizumab pegol) for rheumatoid arthritis and Crohn's disease.

Significant Developments in the First Six Months of 2011

XOMA 052

In December of 2010, we entered into an agreement with Servier to jointly develop and commercialize XOMA 052 in multiple indications, which provided for a non-refundable upfront payment of \$15.0 million that we received in January of 2011. In connection with this agreement, Servier will fully fund the first \$50.0 million of future XOMA 052 global clinical development and chemistry and manufacturing controls ("CMC") expenses, and 50% of further expenses, for the Behcet's uveitis indication, which is expected to advance into Phase 3 development in 2011.

Table of Contents

- In January of 2011, we received the full €15.0 million advance allowed under our loan agreement with Servier dated December 30, 2010, converting to U.S. dollar proceeds of approximately \$19.5 million.
- In March of 2011, we announced that our Phase 2b trial of XOMA 052 in Type 2 diabetes in 421 patients did not achieve the primary endpoint of reduction in hemoglobin A1c ("HbA1c") after six monthly treatments with XOMA 052 compared to placebo. Significant decreases were observed in C-reactive protein ("CRP"), a biomarker for the risk of heart attack, stroke and other cardiovascular diseases, in all dose groups versus placebo. In addition, significant improvements in high-density lipoprotein ("HDL"), or "good" cholesterol, were observed in two of four XOMA 052 dose groups versus placebo. XOMA 052 was well-tolerated in this trial, with no serious drug-related adverse events and a safety profile consistent with previous trials.
- In June of 2011, we announced top line trial results from our six-month Phase 2a trial in 74 patients where XOMA 052 was shown to be well-tolerated with no significant
 differences in adverse events between XOMA 052 and placebo. Evidence of biological activity was observed including a reduction in CRP. There were no differences in
 glycemic control between the drug and placebo groups as measured by HbA1c levels.

XOMA 3AB

In May of 2011, the National Institute of Allergy and Infectious Diseases ("NIAID"), part of the National Institutes of Health ("NIH"), informed us that it is initiating a Phase 1 trial of XOMA 3AB, a novel formulation of three antibodies designed to prevent and treat botulism poisoning. This double-blind, dose-escalation study in approximately 24 healthy volunteers is designed to assess the safety and tolerability, and determine the pharmacokinetic profile, of XOMA 3AB.

Preclinical Pipeline

In June of 2011, we announced our discovery of two new classes of fully-human monoclonal antibodies, XMetA and XMetS, which activate or sensitize the insulin receptor in vivo, each representing a distinct new therapeutic approach to the treatment of patients with diabetes. Studies of XMetA demonstrated that it reduced fasting blood glucose levels and improved glucose tolerance in a mouse model of diabetes. After six weeks of treatment, there was a statically significant reduction in HbA1c levels, a standard measure of average blood glucose levels over time, in mice treated with XMetA compared to a control group, and there was a statistically significant reduction in elevated non-HDL cholesterol levels. Studies of XMetS showed enhanced insulin sensitivity and statistically significant reduction in elevated non-HDL cholesterol in mice treated with XMetS as compared to a control group, and there was a statistically significant reduction in elevated non-HDL cholesterol in the XMetS as compared to a control group, and there was a statistically significant reduction in elevated non-HDL cholesterol in the XMetS as compared to a control group, and there was a statistically significant reduction in elevated non-HDL cholesterol levels. Studies of XMetS as compared to a control group, and there was a statistically significant reduction in elevated non-HDL cholesterol levels. These data were presented at the American Diabetes Association's 71st Scientific Sessions.

Financing-Related

- In the first half of 2011, we sold 821,386 common shares through Wm Smith & Co. ("Wm Smith") and McNicoll, Lewis & Vlak LLC ("MLV") under our At Market Issuance Sales Agreement dated October 26, 2010 (the "2010 ATM Agreement"), for aggregate gross proceeds of \$4.4 million, and 2,398,017 common shares through MLV under our At Market Issuance Sales Agreement dated February 4, 2011 (the "2011 ATM Agreement"), for aggregate gross proceeds of \$5.9 million, of which \$3.7 million of cash was received in June of 2011 with the remaining \$2.2 million of cash received in July of 2011.
- In April of 2011, the 2,959 Series B convertible preference shares previously issued to Genentech, Inc. were converted by Genentech into 254,560 common shares, and the associated liquidation preference of \$29.6 million was eliminated.
- In May of 2011, we entered into two foreign exchange options contracts in order to manage our foreign currency exposure relating to principal and interest payments on our €15.0 million loan from Servier. Upfront premiums paid on these contracts totaled \$1.5 million.

Results of Operations

Revenues

Total revenues for the three and six months ended June 30, 2011 and 2010, were as follows (in thousands):

	 Fhree Months H	Ended	l June 30,	Six Months Ended June 30,			
	2011		2010		2011	_	2010
License and collaborative fees	\$ 6,039	\$	150	\$	11,866	\$	339
Contract and other revenue	10,467		5,481		20,128		12,292
Royalties	 19		311		126		513
Total revenues	\$ 16,525	\$	5,942	\$	32,120	\$	13,144

License and Collaborative Fees

License and collaborative fee revenue includes fees and milestone payments related to the out-licensing of our products and technologies. License and collaborative fee revenue increased by \$5.9 million and \$11.5 million for the three and six months ended June 30, 2011, respectively, compared to the same periods in 2010. These increases were primarily due to \$5.5 million and \$11.0 million in revenue recognized in the three and six months ended June 30, 2011, respectively, related to the collaboration and loan agreements with Servier to jointly develop and commercialize XOMA 052 in multiple indications. The generation of future revenue related to license fees and other collaborative arrangements is dependent on our ability to attract new licensees to our antibody and bacterial cell expression technologies and new collaboration partners. In connection with the collaboration agreement with Servier, we expect to experience an increase in these revenues for 2011 compared to 2010.

Contract and Other Revenue

Contract and other revenue includes agreements where we provide contracted research and development and manufacturing services to our contract and collaboration partners, including NIAID and Servier. The following table shows the activity in contract and other revenue for the three and six months ended June 30, 2011 and 2010 (in thousands):

		Thre	e Mo	nths Ended Jun	1e 3(),	Six Months Ended June 30,						
	_	2011		2010		Increase (Decrease)		2011		2010		Increase (Decrease)	
NIAID	\$	5,368	\$	4,783	\$	585	\$	12,252	\$	8,294	\$	3,958	
Servier		4,592		-		4,592		6,674		-		6,674	
Takeda		296		266		30		584		3,026		(2,442)	
Other		211		432		(221)		618		972		(354)	
Total revenues	\$	10,467	\$	5,481	\$	4,986	\$	20,128	\$	12,292	\$	7,836	

The increase in revenue from our NIAID Contract No. HHSN272200800028C ("NIAID 3") was due to increased activity under the contract. XOMA 052 clinical development and CMC activity under the collaboration with Servier also contributed to the increase in contract revenue. Partially offsetting these increases was a decrease in revenue from our Takeda contracts in 2011 as a result of the cessation of certain Takeda programs in 2010.

Based on expected levels of revenue generating activity related to our NIAID 3 and Servier contracts, we expect contract and other revenue to increase in 2011 compared to 2010.

Royalties

Revenue from royalties decreased by \$0.3 million and \$0.4 million for the three and six months ended June 30, 2011, respectively, compared to the same periods in 2010, primarily due to the sale of our CIMZIA[®] royalty interest in the third quarter of 2010. Royalties earned from sales of CIMZIA[®] were \$0.3 million and \$0.5 million for the three and six months ended June 30, 2010. We will not receive any further royalties on sales of CIMZIA[®].

Research and Development Expenses

Biopharmaceutical development includes a series of steps, including *in vitro* and *in vivo* preclinical testing, and Phase 1, 2 and 3 clinical studies in humans. Each of these steps is typically more expensive than the previous step, but actual timing and the cost to us depends on the product being tested, the nature of the potential disease indication and the terms of any collaborative or development arrangements with other companies or entities. After successful conclusion of all of these steps, regulatory filings for approval to market the products must be completed, including approval of manufacturing processes and facilities for the product. Our research and development expenses currently include costs of personnel, supplies, facilities and equipment, consultants, third party costs and other expenses related to preclinical and clinical testing.

Research and development expenses were \$18.3 million and \$35.6 million for the three and six months ended June 30, 2011, respectively, compared with \$19.3 million and \$36.9 million for the same periods of 2010. The decreases of \$1.0 million and \$1.3 million for the three and six months ended June 30, 2011, respectively, as compared to the same periods in 2010, were primarily due to decreased spending on XOMA 052-related clinical trials as the Phase 2 clinical program in Type 2 diabetes is nearing completion. Partially offsetting this decrease was an increase in spending on NIAID 3 in the first two quarters of 2011 due to increased activity under the contract.

Salaries and related personnel costs are a significant component of research and development expenses. We recorded \$8.8 million and \$17.6 million in research and development salaries and employee-related expenses for the three and six months ended June 30, 2011, respectively, as compared with \$7.3 million and \$14.4 million for the same periods of 2010. The increases of \$1.5 million and \$3.2 million were primarily due to an increase in salaries and benefits of \$1.1 million and \$2.2 million for the three and six months ended June 30, 2011, respectively, and an increase in share-based compensation of \$0.3 million and \$0.9 million, respectively, for the same periods.

Our research and development activities can be divided into earlier stage programs and later stage programs. Earlier stage programs include molecular biology, process development, pilot-scale production and preclinical testing. Also included in earlier stage programs are costs related to excess manufacturing capacity, which we expect will decrease in 2011 due to the execution of the collaboration agreement with Servier, resulting in increased manufacturing capacity requirements. Later stage programs include clinical testing, regulatory affairs and manufacturing clinical supplies. The approximate costs associated with these programs for the three and six months ended June 30, 2011 and 2010 were as follows (in thousands):

	T	hree Months I	Ended	June 30,	Six Months Ended June			1 June 30,
		2011		2010		2011		2010
Earlier stage programs	\$	13,348	\$	12,566	\$	26,305	\$	23,740
Later stage programs		4,933		6,780		9,323		13,193
Total	\$	18,281	\$	19,346	\$	35,628	\$	36,933

Our research and development activities can also be divided into those related to our internal projects and those projects related to collaborative and contract arrangements. The approximate costs related to internal projects and collaborative and contract arrangements for the three and six months ended June 30, 2011 and 2010 were as follows (in thousands):

	Th	ree Months l	Ended	June 30,		June 30,		
		2011	2010		2011			2010
Internal projects	\$	12,051	\$	14,838	\$	20,937	\$	28,524
Collaborative and contract arrangements		6,230		4,508		14,691		8,409
Total	\$	18,281	\$	19,346	\$	35,628	\$	36,933

For the three and six months ended June 30, 2011, each of the two programs upon which we incurred the largest amount of expense (NIAID and XOMA 052) accounted for more than 20% but less than 30% of our total research and development expense. All remaining development programs accounted for less than 10% of our total research and development expense for the three and six months ended June 30, 2011. For the three and six months ended June 30, 2010, the program upon which we incurred the largest amount of expense (XOMA 052) accounted for more than 30% but less than 40%, and NIAID accounted for more than 20% but less than 30% of our total research and development expenses. All remaining development programs accounted for less than 30% of our total research and development expenses.

We expect our research and development spending in 2011 compared to 2010 will increase primarily due to increased activity associated with the expected initiation in 2011 of our Phase 3 clinical program for XOMA 052 for the Behcet's uveitis indication under our collaboration agreement with Servier.

Future research and development spending may be impacted by potential new licensing or collaboration or development arrangements, as well as the termination of existing agreements. Beyond this, the scope and magnitude of future research and development expenses are difficult to predict at this time.

Selling, General and Administrative Expenses

Selling, general and administrative expenses include salaries and related personnel costs, facilities costs and professional fees. Selling, general and administrative expenses were \$6.1 million and \$11.5 million for the three and six months ended June 30, 2011, respectively, compared with \$5.0 million and \$10.6 million for the same periods of 2010. The increases of \$1.1 million and \$0.9 million for the three and six months ended June 30, 2011, respectively, as compared to the same periods of 2010, were primarily due to increases in share-based compensation of \$0.9 million and \$1.1 million, respectively, and increases in professional fees of \$0.4 million and \$0.2 million, respectively, primarily relating to higher consulting service costs, partially offset by decreases due to our continued focus on cost control.

Other Income (Expense)

Interest expense was \$0.6 million and \$1.2 million for the three and six months ended June 30, 2011, respectively, compared to \$0.1 million and \$0.2 million for the same periods of 2010. The increases in interest expense of \$0.5 million and \$1.0 million for the three and six months ended June 30, 2011, respectively, as compared to the same periods of 2010, were primarily due to interest expense incurred as a result of the loan with Servier, which was funded in January of 2011. Refer to *Liquidity and Capital Resources: Servier Loan* below for further discussion of the loan with Servier.

Other income (expense) primarily consisted of gains (losses) on revaluation of warrant liabilities and unrealized and realized gains (losses). The following table shows the activity in other income (expense) for the three and six months ended June 30, 2011 and 2010 (in thousands):

	 Three	Mo	nths Ended Jun	e 30),		Six Months Ended June 30,					
	 2011		2010		Increase (Decrease)		2011	2010			Increase (Decrease)	
Other income (expense)												
Gain on warrant revaluation	\$ 459	\$	2,951	\$	(2,492)	\$	2,850	\$	1,691	\$	1,159	
Unrealized foreign exchange loss (1)	(257)		-		(257)		(1,874)		-		(1,874)	
Realized foreign exchange gain ⁽²⁾	(4)		(1)		(3)		556		(2)		558	
Unrealized gain on foreign exchange options	157		-		157		157		-		157	
Warrant modification expense	-		-		-		-		(4,500)		4,500	
Other	-		-		-		(11)		(2)		(9)	
Total other income (expense)	\$ 355	\$	2,950	\$	(2,595)	\$	1,678	\$	(2,813)	\$	4,491	

(1) Unrealized foreign exchange loss for the three and six months ended June 30, 2011 primarily relates to losses on the re-measurement of the ϵ 15 million Servier loan.

(2) Realized foreign exchange gain for the six months ended June 30, 2011 primarily relates to the conversion into U.S. dollars of th€15 million cash proceeds received from Servier in January of 2011.

Warrant Liabilities

In February of 2010, we issued warrants to purchase 1,260,000 of XOMA's common shares in connection with an underwritten offering. We have accounted for the warrants issued in February of 2010 as a liability at fair value. At December 31, 2010, the fair value of the warrant liabilities was \$3.5 million, estimated using the Black-Scholes Option Pricing Model (the "Black-Scholes Model"). We revalued the warrant liability at June 30, 2011 using the Black-Scholes Model and recorded decreases in the fair value of \$0.4 million and \$2.3 million for the three and six months ended June 30, 2011, respectively, as gains in the other income (expense) line of our condensed consolidated statement of operations. As of June 30, 2011, all of these warrants were outstanding and the fair value of the warrant liability was \$1.2 million.

In June of 2009, we issued warrants to certain institutional investors as part of a registered direct offering. The warrants represent the right to acquire an aggregate of up to 347,826 common shares over a five year period beginning December 11, 2009 at an exercise price of \$19.50 per share. We have accounted for the warrants issued in June of 2009 as a liability at fair value. At December 31, 2010, the fair value of the warrant liabilities was \$0.8 million, estimated using the Black-Scholes Model. We revalued the warrant liability at June 30, 2011 using the Black-Scholes Model and recorded decreases in the fair value of \$0.1 million and \$0.5 million for the three and six months ended June 30, 2011, respectively, as gains in the other income (expense) line of our condensed consolidated statement of operations. As of June 30, 2011, all of these warrants were outstanding and the fair value of the warrant liability was \$0.2 million.

Income Taxes

Income tax expense was not material for the three and six months ended June 30, 2011 and 2010.



Accounting Standards Codification Topic 740, *Income Taxes* provides for the recognition of deferred tax assets if realization of such assets is more likely than not. Based upon the weight of available evidence, which includes our historical operating performance and carry-back potential, we have determined that total deferred tax assets should be fully offset by a valuation allowance.

We did not have unrecognized tax benefits as of June 30, 2011 and do not expect this to change significantly over the next twelve months. We will recognize future interest and penalties accrued on any unrecognized tax benefits as a component of income tax expense. As of June 30, 2011, we have not accrued interest or penalties related to uncertain tax positions.

Liquidity and Capital Resources

Cash and cash equivalents at June 30, 2011 were \$51.2 million compared with \$37.3 million at December 31, 2010. Net cash used in operating activities was \$13.3 million for the six months ended June 30, 2011, compared with \$32.5 million for the same period in 2010. The decrease in cash used in operations for the six months ended June 30, 2011, as compared to same period of 2010, was primarily due to an increase in revenue receipts for license and collaborative fees, including the receipt of the \$15.0 million license fee as consideration for the collaboration with Servier, and the timing of collections of accounts receivable and payment of accounts payable.

In the first half of 2011, net accounts payable and accrued liabilities decreased by \$2.5 million primarily due to the payment in the period of employee bonuses for 2010.

Comparatively, for the six months ended June 30, 2010, trade and other receivables increased by \$1.3 million primarily a result of increased activity related to NIAID 3 and deferred revenue decreased by \$2.0 million primarily due to a decline in advance billings resulting from decreased activity under our collaboration contracts and the accelerated recognition of the remaining \$1.1 million of the unamortized balance in deferred revenue pertaining to a discontinued discovery and development program under our collaboration with Takeda.

Net cash used in investing activities was \$2.3 million for the six months ended June 30, 2011, compared with \$0.3 million for the same period of 2010. Cash used in investing activities for the six months ended June 30, 2011 and 2010 consisted of fixed asset purchases relating to CMC activity.

Net cash provided by financing activities was \$30.0 million for the six months ended June 30, 2011, compared with \$21.0 million for the same period of 2010. Cash provided by financing activities in the first half of 2011 related to proceeds received from the incurrence of the Servier loan for \$20.1 million and the issuance of common shares for \$9.9 million under the 2010 ATM Agreement and 2011 ATM Agreement. Cash provided by financing activities in the first half of 2010 related to proceeds of \$25.5 million received from the issuance of common shares, partially offset by \$4.5 million paid to the holders of warrants issued in June of 2009 upon modification of the terms.

Foreign Exchange Options

We hold debt and may incur expenses denominated in foreign currencies, which exposes us to market risk associated with foreign currency exchange rate fluctuations between the U.S. dollar and the Euro. We are required to make principal and accrued interest payments in Euros on our $\notin 15.0$ million loan from Servier. In order to manage our foreign currency exposure related to these payments, in May of 2011, we entered into two foreign exchange option contracts to buy $\notin 15.0$ million and $\notin 1.5$ million on January 2016 and January 2014, respectively. By having these option contracts in place, our foreign exchange rate risk is reduced if the U.S. dollar weakens against the Euro. However, if the U.S. dollar strengthens against the Euro, we are not required to exercise these options, but will not receive any refund on premiums paid.

Upfront premiums paid on these foreign exchange option contracts totaled \$1.5 million. The fair values of these option contracts are re-valued at each reporting period and are estimated based on pricing models using readily observable inputs from actively quoted markets. The fair values of these option contracts are included in other assets on the condensed consolidated balance sheet and changes in fair value on these contracts are included in other income (expense) on the condensed consolidated statements of operations. The foreign exchange options were revalued at June 30, 2011 and had an aggregate fair value of \$1.7 million, and we recognized a gain of \$0.2 million related to the revaluation for the three and six months ended June 30, 2011.

Servier Loan

In December of 2010, in connection with the license and collaboration agreement entered into with Servier, we executed a loan agreement with Servier, which provided for an advance of up to €15.0 million. The loan was fully funded in January of 2011, with the proceeds converting to approximately \$19.5 million. The loan is secured by an interest in XOMA's intellectual property rights to all XOMA 052 indications worldwide, excluding certain rights in the U.S. and Japan. Interest is calculated at a floating rate based on a Euro Inter-Bank Offered Rate ("EURIBOR") and subject to a cap. The interest rate is reset semi-annually in January and July of each year. The interest rate for the initial interest period was 3.22%. The interest rate has been reset to 3.83% for six-month period from July 2011 through January 2012. Interest is payable semi-annually; however, the loan agreement provides for a deferral of interest payments over a period specified in the agreement. During the deferral period, accrued interest will be added to the outstanding principal amount for the purpose of interest shall be due and payable at the end of each six-month period. The loan matures in 2016; however, after a specified period prior to final maturity, the loan is to be repaid (i) at Servier's option, by applying up to a significant percentage of any milestone or royalty payments ower development partner for rights to XOMA 052 in the U.S. and/or Japan. In addition, the loan becomes immediately due and payable upon certain customary events of default. At June 30, 2011, the outstanding principal balance under this loan was \$21.6 million. For the three and six months ended June 30, 2011, we recorded unrealized foreign exchange losses of \$0.5 million and \$2.1 million, respectively, related to the re-measurement of the loan as of June 30, 2011.

The loan has a stated interest rate lower than the market rate based on comparable loans held by similar companies, which represents additional value to us. We recorded this additional value as a discount to the face value of the loan amount, at its fair value of \$8.9 million. The fair value of this discount, which was determined using a discounted cash flow model, represents the differential between the stated terms and rates of the loan and market rates. Based on the association of the loan with the collaboration arrangement, we recorded the offset to this discount as deferred revenue.

The loan discount is amortized under the effective interest method over the expected five-year life of the loan. We recorded non-cash interest expense of \$0.4 million and \$0.7 million during the three and six months ended June 30, 2011, respectively, resulting from the amortization of the loan discount. At June 30, 2011, the net carrying value of the loan was \$13.2 million.

We believe that realization of the benefit and the associated deferred revenue is contingent on the loan remaining outstanding over the five-year contractual term of the loan. If we were to stop providing service under the collaboration arrangement and the arrangement is terminated, the maturity date of the loan would be accelerated and a portion of measured benefit would not be realized. As the realization of the benefit is contingent, in part, on the provision of future services, we are recognizing the deferred revenue over the expected five-year life of the loan. The deferred revenue is amortized under the effective interest method, and we recorded \$0.4 million and \$0.7 million of related non-cash revenue during the three and six months ended June 30, 2011.

ATM Agreements

In the third quarter of 2010, we entered into the 2010 ATM Agreement, with Wm Smith and MLV (the "Agents"), under which we could sell common shares from time to time through the Agents, as our agents for the offer and sale of the common shares, in an aggregate amount not to exceed the amount that can be sold under our registration statement on Form S-3 (File No. 333-148342) filed with the U.S. Securities and Exchange Commission (the "SEC") on December 26, 2007 and declared effective by the SEC on May 29, 2008. The Agents could sell the common shares by any method permitted by law deemed to be an "at the market" offering as defined in Rule 415 of the Securities Act of 1933, as amended (the "Securities Act"), including without limitation sales made directly on The NASDAQ Global Market, on any other existing trading market for the common shares or to or through a market maker. The Agents could also sell the common shares in privately negotiated transactions, subject to our prior approval. From the inception of the 2010 ATM Agreement through May of 2011, we sold a total of 7,560,862 common shares under this agreement for aggregate gross proceeds of \$34.4 million. Total offering expenses incurred related to sales under the 2010 ATM Agreement from inception to May of 2011 wer \$1.0 million, including \$0.1 million incurred in 2011. In May of 2011, 2010 ATM Agreement expired by its terms, and there will be no further issuances under this facility.

On February 4, 2011, we entered into an At Market Issuance Sales Agreement (the "2011 ATM Agreement"), with MLV, under which we may sell common shares from time to time through the MLV, as our agent for the offer and sale of the common shares, in an aggregate amount not to exceed the amount that can be sold under our registration statement on Form S-3 (File No. 333-172197) filed with the SEC on February 11, 2011 and amended on March 10, 2011 and June 3, 2011, which was declared effective by the SEC on June 6, 2011. MLV may sell the common shares by any method permitted by law deemed to be an "at the market" offering as defined in Rule 415 of the Securities Act, including without limitation sales made directly on The NASDAQ Global Market, on any other existing trading market for the common shares or to or through a market maker. MLV may also sell the common shares in privately negotiated transactions, subject to our prior approval. From the inception of the 2011 ATM Agreement through June 30, 2011, we sold a total of 2,398,017 common shares under this agreement for aggregate gross proceeds of \$5.9 million, of which \$3.7 million of cash vas received in June of 2011. Total offering expenses incurred related to sales under the 2011 ATM Agreement from inception to June 30, 2011 were \$0.2 million. From July 1, 2011 through August 2, 2011, 353,424 additional common shares were sold through MLV for aggregate gross proceeds of \$0.8 million. Total offering expenses related to these sales from July 1, 2011 to August 2, 2011 were approximately \$25,000.

Net proceeds received during the first half of 2011 from the Servier loan, 2010 ATM Agreement and 2011 ATM Agreement were used to continue development of our XOMA 052 product candidate and for other working capital and general corporate purposes. As of August 2, 2011, there were approximately \$93.3 million of gross proceeds available for issuance pursuant to the above-mentioned registration statement.

* * *

We have incurred significant operating losses and negative cash flows from operations since our inception. At June 30, 2011, we had an accumulated deficit of \$867.8 million, cash and cash equivalents of \$51.2 million and working capital of \$43.5 million. During the remainder of 2011, we expect to continue using our cash and cash equivalents to fund ongoing operations. Additional licensing, antibody discovery and development collaboration agreements, government funding and financing arrangements may positively impact our cash balances. Based on our cash reserves and anticipated spending levels, revenue from collaborations including the XOMA 052 collaboration agreement with Servier, funding from the loan agreement with Servier, biodefense contracts and licensing transactions and other sources of funding that we believe to be available, we estimate that we have sufficient cash resources to meet our anticipated net cash needs through the next twelve months. Any significant revenue shortfalls, increases in planned spending on development programs or more rapid progress of development programs than anticipated, as well as the unavailability of anticipated sources of funding, could shorten this period. If adequate funds are not available, we will be required to delay, reduce the scope of, or eliminate one or more of our product development programs and further reduce personnel-related costs. Progress or setbacks by potentially competing products may also affect our ability to raise new funding on acceptable terms.

Critical Accounting Estimates

Critical accounting estimates are those that require significant judgment and/or estimates by management at the time that the financial statements are prepared such that materially different results might have been reported if other assumptions had been made. We consider certain accounting policies including, but not limited to, revenue recognition, long-lived assets, derivative instruments, warrant liabilities and share-based compensation to be critical policies. There have been no significant changes in our critical accounting estimates during the six months ended June 30, 2011, as compared with those previously disclosed in our Annual Report on Form 10-K for the fiscal year ended December 31, 2010, filed with the SEC on March 10, 2011.

Forward-Looking Information and Cautionary Factors That May Affect Future Results

Certain statements contained herein related to the sufficiency of our cash resources, the timing of initiation of clinical trials and the amounts of certain revenues and certain costs in comparison to prior years, or that otherwise relate to future periods, are forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. These statements are based on assumptions that may not prove accurate. Actual results could differ materially from those anticipated due to certain risks inherent in the biotechnology industry and for companies engaged in the development of new products in a regulated market. Among other things, the period for which our cash resources are sufficient could be shortened if expenditures are made earlier or in larger amounts than anticipated or are unanticipated, if anticipated revenue or cost sharing arrangements do not materialize, or if funds are not otherwise available on acceptable terms; plans to initiate new clinical trials may change depending on availability of resources, actions or inactions by our present or future collaboration partners or unanticipated safety issues; and our revenues may be lower than anticipated, and our costs may be higher than expected, due to actions or inactions by our present or future collaboration partners, unanticipated safety issues or unavailability of additional licensing or collaboration opportunities. These and other risks, including those related the generally unstable nature of current economic and financial market conditions; the results of discovery research and preclinical testing; the timing or results of pending and future clinical trials (including the design and progress of clinical trials; safety and efficacy of the products being tested; action, inaction or delay by the FDA, European or other regulators or their advisory bodies; and analysis or interpretation by, or submission to, these entities or others of scientific data); changes in the status of existing collaborative or licensing relationships; the ability of collaborators, licensees and other third parties to meet their obligations and their discretion in decision-making; our ability to meet the demands of the United States government agency with which we have entered our government contracts; competition; market demand for products; scale-up, manufacturing and marketing capabilities; availability of additional licensing or collaboration opportunities; international operations; share price volatility; our financing needs and opportunities; uncertainties regarding the status of biotechnology patents; uncertainties as to the costs of protecting intellectual property; and risks associated with our status as a Bermuda company, are described in more detail in Part II - Item 1A: Risk Factors.



ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Interest Rate Risk

Our exposure to market rate risk for changes in interest rates relates primarily to our investment portfolio and our secured note and loan agreements. By policy, we make our investments in high quality debt securities, limit the amount of credit exposure to any one issuer and limit duration by restricting the term of the instrument. We generally hold investments to maturity, with a weighted average portfolio period of less than twelve months. However, if the need arose to liquidate such securities before maturity, we may experience losses on liquidation.

We hold interest-bearing instruments that are classified as cash and cash equivalents. Fluctuations in interest rates can affect the principal values and yields of fixed income investments. If interest rates in the general economy were to rise rapidly in a short period of time, our fixed income investments could lose value.

The following table presents the amounts and related weighted average interest rates of our cash and cash equivalents at June 30, 2011 and December 31, 2010 (in thousands, except interest rates):

June 30, 2011	Maturity	A	rrying nount ousands)	air Value thousands)	Average Interest Rate
Cash and cash equivalents	Daily to 90 days	\$	51,156	\$ 51,157	0.10%
December 31, 2010 Cash and cash equivalents	Daily to 90 days	\$	37,304	\$ 37,304	0.09%

As of June 30, 2011, we have an outstanding principal balance on our note with Novartis of \$13.9 million, which is due in 2015. The interest rate on this note is charged at a rate of USD six-month LIBOR plus 2%, which was 2.39% at June 30, 2011. No further borrowing is available under this note.

As of June 30, 2011, we have an outstanding principal balance on our loan with Servier of \notin 15.0 million, which converts to approximately \$21.6 million at June 30, 2011. The interest rate on this loan is charged at a floating rate based on a Euro Inter-Bank Offered Rate ("EURIBOR") and subject to a cap. The interest rate was equal to 3.22% at June 30, 2011, and has been reset to 3.83% for six month period from July 2011 through January 2012. No further borrowing is available under this loan.

The variable interest rates related to our long-term debt instruments are based on LIBOR and EURIBOR. We estimate that a hypothetical 100 basis point change in interest rates could increase or decrease our interest expense by approximately \$0.4 million on an annualized basis.

Foreign Currency Risk

We hold debt and may incur expenses denominated in foreign currencies. The amount of debt owed or expenses incurred will be impacted by fluctuations in these foreign currencies. When the U.S. dollar weakens against foreign currencies, the U.S. dollar value of the foreign-currency denominated debt and expense increases, and when the U.S. dollar strengthens against these currencies, the U.S. dollar value of the foreign-currency denominated debt and expense increases, and when the U.S. dollar strengthens against these currencies, the U.S. dollar value of the foreign-currency denominated debt and expense decreases. Consequently, changes in exchange rates will affect the amount we are required to repay on our €15.0 million loan from Servier and may affect our results of operations. Our loan from Servier was fully funded in January of 2011, with the proceeds converting to approximately \$19.5 million using the January 13, 2011 Euro to USD exchange rate. At June 30, 2011, the €15.0 million outstanding principal balance under this loan agreement would have equaled approximately \$21.6 million using the June 30, 2011 Euro to USD exchange rate. In May of 2011, in order to manage our foreign currency exposure relating to our principal and interest payments on our loan from Servier, we entered into two foreign exchange option contracts. Our use of derivative financial instruments represents risk management; we do not enter into derivative financial contracts for trading purposes. Refer to *Item 1, Condensed Consolidated Financial Statements*, *Note 3 of Notes to Consolidated Financial Statements* for additional information of the foreign exchange option contracts. Our derivative financial instruments are recorded in the consolidated balance sheets at fair value as of the balance sheet dates.

77	

ITEM 4. CONTROLS AND PROCEDURES

Evaluation of Controls and Procedures

Under the supervision and with the participation of our management, including our Chairman, Chief Executive Officer and President (our principal executive officer) and our Vice President, Finance and Chief Financial Officer (our principal financial officer), we conducted an evaluation of our disclosure controls and procedures, as such term is defined under Rule 13a-15(e) promulgated under the Securities Exchange Act of 1934, as amended, as of the end of the period covered by this report. Based on this evaluation, our Chairman, Chief Executive Officer and President and our Vice President, Finance and Chief Financial Officer concluded that our disclosure controls and procedures are effective as of the end of the period covered by this report in timely alerting them to material information relating to us and our consolidated subsidiaries required to be included in our periodic SEC filings.

Changes in Internal Control

There have been no changes in our internal controls over financial reporting during our most recent fiscal quarter that have materially affected, or are reasonably likely to materially affect, our internal controls over financial reporting.

PART II - OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

On April 9, 2009, a complaint was filed in the Superior Court of Alameda County, California, in a lawsuit captioned Hedrick et al. v. Genentech, Inc. et al, Case No. 09-446158. The complaint asserts claims against Genentech, us and others for alleged strict liability for failure to warn, strict product liability, negligence, breach of warranty, fraudulent concealment, wrongful death and other claims based on injuries alleged to have occurred as a result of three individuals' treatment with RAPTIVA®. The complaint seeks unspecified compensatory and punitive damages. Since then, additional complaints have been filed in the same court, bringing the total number of pending cases to seventy five. The cases have been consolidated as a coordinated proceeding. All of the complaints allege the same claims and seek the same types of damages based on injuries alleged to have occurred as a result of the plaintiffs' treatment with RAPTIVA®. On July 15, 2011, the Court dismissed with prejudice one of the cases in this coordinated proceeding, White v. Genentech, Inc., et al, Case No. RG-09-484026. Our agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which we believe is applicable to these matters. We believe the claims against us to be without merit and intend to defend against them vigorously.

On August 4, 2010, a petition was filed in the District Court of Dallas County, Texas in a case captioned McCall v. Genentech, Inc., et al., No. 10-09544. The defendants filed a Notice of Removal to the United States District Court for the Northern District of Texas on September 3, 2010. The removed case is captioned McCall v. Genentech, Inc., et al., No. 3:10-cv-01747-B. The parties have fully briefed the plaintiff's Motion to Remand and are awaiting a final ruling from the Court. The petition asserts personal injury claims against Genentech, us and others arising out of the plaintiff's treatment with RAPTIVA®. The petition alleges claims based on negligence, strict liability, misrepresentation and suppression, conspiracy, and actual and constructive fraud. The petition seeks compensatory damages and punitive damages in an unspecified amount. On June 6, 2011, the Court dismissed plaintiff's claims of negligent misrepresentation, fraud, and conspiracy. Our agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which we believe is applicable to this matter. We believe the claims against us to be without merit and intend to defend against them vigorously.

On January 7, 2011, a complaint was filed in the United States District Court for the Northern District of Texas in a case captioned Massa v. Genentech, Inc., et al., No. 4:11CV70. On January 11, 2011, a complaint was filed in the United States District Court for the District of Massachusetts in a case captioned Sylvia, et al. v. Genentech, Inc., et al., No. 1:11-cv-10054-MLW. These two complaints allege the same claims against Genentech, us and others and seek the same types of damages as the complaints filed in the Superior Court of Alameda County, California referenced above. Our agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which we believe is applicable to these matters. We believe the claims against us to be without merit and intend to defend against them vigorously.

On April 8, 2011, four complaints were filed in the United States District Court for the Eastern District of Michigan. The cases are captioned: Muniz v. Genentech, et al., 5:11-cv-11489-JCO-RSW; Tifenthal v. Genentech, et al., 2:11-cv-11489-DPH-LJM; Blair v. Genentech, et al., 2:11-cv-11463-SFC-MJH; and Marsh v. Genentech, et al., 2:11-cv-11462-RHC-MKM. The complaints allege the same claims against Genentech, us and others and seek the same types of damages as the complaints filed in the Superior Court of Alameda County, California referenced above. All four cases have been transferred to the United States District Court for the Western District of Michigan. Genentech and we have filed Motions to Dismiss all four actions and are awaiting a decision from the Court. Our agreement with Genentech provides for an indemnity of XOMA and payment of legal fees by Genentech which we believe is applicable to this matter. We believe the claims against us to be without merit and intend to defend against them vigorously.

ITEM 1A. RISK FACTORS

The following risk factors and other information included in this quarterly report should be carefully considered. The risks and uncertainties described below are not the only ones we face. Additional risks and uncertainties not presently known to us also may impair our business operations. If any of the following risks occur, our business, financial condition, operating results and cash flows could be materially adversely affected.

Because our product candidates are still being developed, we will require substantial funds to continue; we cannot be certain that funds will be available and, if they are not available, we may have to take actions that could adversely affect your investment and may not be able to continue operations.

We will need to commit substantial funds to continue development of our product candidates and we may not be able to obtain sufficient funds on acceptable terms, or at all. If adequate funds are not available, we may have to raise additional funds in a manner that may dilute or otherwise adversely affect the rights of existing shareholders, curtail or cease operations, or file for bankruptcy protection in extreme circumstances. We have spent, and we expect to continue to spend, substantial funds in connection with:

- · research and development relating to our product candidates and production technologies,
- various human clinical trials, and
- · protection of our intellectual property.

We finance our operations primarily through our multiple revenue streams resulting from the licensing of our antibody technologies, discovery and development collaborations, product royalties and biodefense contracts, and sales of our common shares. In September of 2009, we sold our royalty interest in LUCENTIS[®] to Genentech, Inc., a wholly-owned member of the Roche Group ("Genentech") for gross proceeds of \$25.0 million, including royalty revenue from the second quarter of 2009. These proceeds, along with other funds, were used to fully repay our loan from Goldman Sachs Specialty Lending Holdings, Inc. ("Goldman Sachs"). As a result, we no longer have a royalty interest in CIMZIA[®] for gross proceeds of \$4.0 million, including royalty revenue from the second quarter of 2010. As a result, we no longer have a royalty interest in CIMZIA[®]. We received revenue from this royalty interest of \$0.5 million in 2010 and \$0.5 million in 2009.

Based on our cash reserves and anticipated spending levels, revenue from collaborations including our XOMA 052 collaboration agreement with Les Laboratoires Servier ("Servier"), funding from our loan agreement with Servier, biodefense contracts and licensing transactions and other sources of funding that we believe to be available, we believe that we have sufficient cash resources to meet our anticipated net cash needs through the next twelve months. Any significant revenue shortfalls, increases in planned spending on development programs or more rapid progress of development programs than anticipated, as well as the unavailability of anticipated sources of funding, could shorten this period. If adequate funds are not available, we will be required to delay, reduce the scope of, or eliminate one or more of our product development programs and further reduce personnel-related costs. Progress or setbacks by potentially competing products may also affect our ability to raise new funding on acceptable terms. As a result, we do not know when or whether:

- · operations will generate meaningful funds,
- · additional agreements for product development funding can be reached,
- · strategic alliances can be negotiated, or
- adequate additional financing will be available for us to finance our own development on acceptable terms, or at all.

Cash balances and operating cash flow are influenced primarily by the timing and level of payments by our licensees, collaborators and development partners, as well as by our operating costs.

Global credit and financial market conditions may reduce our ability to access capital and cash and could negatively impact the value of our current portfolio of cash equivalents and our ability to meet our financing objectives.

Traditionally, we have funded a large portion of our research and development expenditures through raising capital in the equity markets. Recent events, including failures and bankruptcies among large commercial and investment banks, have led to considerable declines and uncertainties in these and other capital markets and have led to new regulatory and other restrictions that may broadly affect the nature of these markets. These circumstances could severely restrict the raising of new capital by companies such as us in the future.

Volatility in the financial markets has also created liquidity problems in investments previously thought to bear a minimal risk. For example, money market fund investors, including us, have in the past been unable to retrieve the full amount of funds, even in highly-rated liquid money market accounts, upon maturity. Although as of June 30, 2011, we have received the full amount of proceeds from money market fund investments, an inability to retrieve funds from money market fund investments as they mature in the future could have a material and adverse impact on our business, results of operations and cash flows.

Our cash and cash equivalents are maintained in highly liquid investments with remaining maturities of 90 days or less at the time of purchase. While as of the date of this filing, we are not aware of any downgrades, material losses, or other significant deterioration in the fair value of our cash equivalents since June 30, 2011, no assurance can be given that further deterioration in conditions of the global credit and financial markets would not negatively impact our current portfolio of cash equivalents or our ability to meet our financing objectives.

Because all of our product candidates are still being developed, we have sustained losses in the past and we expect to sustain losses in the future.

We have experienced significant losses and, as of June 30, 2011, we had an accumulated deficit of \$867.8 million.

For the three and six months ended June 30, 2011, we had net losses of approximately \$8.1 million or \$0.27 per common share (basic and diluted) and \$14.5 million or \$0.49 per common share (basic and diluted), respectively. For the three and six months ended June 30, 2010, we had net losses of approximately \$15.6 million or \$0.93 per common share (basic and diluted) and \$37.4 million or \$2.24 per common share (basic and diluted), respectively.

Our ability to achieve profitability is dependent in large part on the success of our development programs, obtaining regulatory approval for our product candidates and entering into new agreements for product development, manufacturing and commercialization, all of which are uncertain. Our ability to fund our ongoing operations is dependent on the foregoing factors and on our ability to secure additional funds. Because our product candidates are still being developed, we do not know whether we will ever achieve sustained profitability or whether cash flow from future operations will be sufficient to meet our needs.

We may issue additional equity securities and thereby materially and adversely affect the price of our common shares.

We are authorized to issue, without shareholder approval, 1,000,000 preference shares, of which none were issued and outstanding as of August 2, 2011, which may give other shareholders dividend, conversion, voting, and liquidation rights, among other rights, which may be superior to the rights of holders of our common shares. In April of 2011, the 2,959 Series B convertible preference shares previously issued to Genentech were converted by Genentech into 254,560 common shares. In addition, we are authorized to issue, generally without shareholder approval, up to 92,666,666 common shares, of which 32,554,155 were issued and outstanding as of August 2, 2011. If we issue additional equity securities, the price of our common shares may be materially and adversely affected.

In the third quarter of 2009, we had entered into an At Market Issuance Sales Agreement (the "2009 ATM Agreement"), with Wm Smith & Co. ("Wm Smith"), under which we could sell up to 1.7 million of our common shares from time to time through Wm Smith, as the agent for the offer and sale of the common shares. Wm Smith could sell these common shares by any method permitted by law deemed to be an "at the market" offering as defined in Rule 415 of the Securities Act of 1933, as amended (the "Securities Act"), including but not limited to sales made directly on The NASDAQ Global Market, on any other existing trading market for the common shares or to or through a market maker. Wm Smith could also sell the common shares in privately negotiated transactions, subject to our approval. From the inception of the 2009 ATM Agreement through October 27, 2010, we sold a total of 1,666,666 common shares through Wm Smith, constituting all of the shares available for sale under the agreement, for aggregate gross proceeds of \$12.2 million.

In February of 2010, we completed an underwritten offering of 2.8 million units, with each unit consisting of one of our common shares and a warrant to purchase 0.45 of a common share, for gross proceeds of approximately \$21.0 million, before deducting underwriting discounts and commissions and estimated offering expenses of \$1.7 million. The investors purchased the units at a price of \$7.50 per unit. The warrants, which represent the right to acquire an aggregate of up to 1.26 million common shares, are exercisable beginning six months and one day after issuance and have a five-year term and an exercise price of \$10.50 per share.



In July of 2010, we entered into a common share purchase agreement with Azimuth Opportunity, Ltd. ("Azimuth"), pursuant to which we obtained a committed equity line of credit facility under which we could sell up to \$30.0 million of our registered common shares to Azimuth over a 12-month period, subject to certain conditions and limitations. In August of 2010, we sold a total of 3,421,407 common shares under this facility for aggregate proceeds of \$14.2 million, representing the maximum number of shares that could be sold under this facility.

In October of 2010, we entered into an At Market Issuance Sales Agreement (the "2010 ATM Agreement"), with Wm Smith and McNicoll, Lewis & Vlak LLC (the "Agents"), under which we could sell common shares from time to time through the Agents, as our agents for the offer and sale of the common shares, in an aggregate amount not to exceed the amount that can be sold under our registration statement on Form S-3 (File No. 333-148342) filed with the U.S. Securities and Exchange Commission (the "SEC") on December 26, 2007 and declared effective by the SEC on May 29, 2008. The Agents could sell the common shares by any method permitted by law deemed to be an "at the market" offering as defined in Rule 415 of the Securities Act, including without limitation sales made directly on The NASDAQ Global Market, on any other existing trading market for the common shares or to or through a market maker. The Agents could also sell the common shares in privately negotiated transactions, subject to our prior approval. From the inception of the 2010 ATM Agreement through May of 2011, we sold a total of 7.6 million common shares under this agreement for aggregate gross proceeds of \$4.4 million. In May of 2011, the 2010 ATM Agreement expired by its terms, and there will be no further issuances under this facility.

On February 4, 2011, we entered into an At Market Issuance Sales Agreement with McNicoll, Lewis & Vlak LLC ("MLV"), under which we may sell common shares from time to time through the MLV, as our agent for the offer and sale of the common shares, in an aggregate amount not to exceed the amount that can be sold under our registration statement on Form S-3 (File No. 333-172197) filed with the SEC on February 11, 2011 and amended on March 10, 2011 and June 3, 2011, which was declared effective by the SEC on June 6, 2011. MLV may sell the common shares by any method permitted by law deemed to be an "at the market" offering as defined in Rule 415 of the Securities Act, including without limitation sales made directly on The NASDAQ Global Market, on any other existing trading market for the common shares in privately negotiated transactions, subject to our prior approval. From the inception of the 2011 ATM Agreement through August 2, 2011, we sold a total of 2,751,441 common shares under this agreement for aggregate gross proceeds of \$6.7 million.

The financial terms of future collaborative or licensing arrangements could result in dilution of our share value.

Funding from collaboration partners and others has in the past and may in the future involve issuance by us of our shares. Because we do not currently have any such arrangements, we cannot be certain how the purchase price of such shares, the relevant market price or premium, if any, will be determined or when such determinations will be made. Any such issuance could result in dilution in the value of our issued and outstanding shares.

Our share price may be volatile and there may not be an active trading market for our common shares.

There can be no assurance that the market price of our common shares will not decline below its present market price or that there will be an active trading market for our common shares. The market prices of biotechnology companies have been and are likely to continue to be highly volatile. Fluctuations in our operating results and general market conditions for biotechnology stocks could have a significant impact on the volatility of our common share price. We have experienced significant volatility in the price of our common shares. From January 1, 2011 through August 2, 2011, our share price has ranged from a high of \$7.71 to a low of \$2.07. Factors contributing to such volatility include, but are not limited to:

- · results of preclinical studies and clinical trials,
- · information relating to the safety or efficacy of products or product candidates,
- · developments regarding regulatory filings,
- · announcements of new collaborations,
- · failure to enter into collaborations,
- · developments in existing collaborations,
- · our funding requirements and the terms of our financing arrangements,

- · technological innovations or new indications for our therapeutic products and product candidates,
- · introduction of new products or technologies by us or our competitors,
- · sales and estimated or forecasted sales of products for which we receive royalties, if any,
- · government regulations,
- · developments in patent or other proprietary rights,
- · the number of shares issued and outstanding,
- the number of shares trading on an average trading day,
- · announcements regarding other participants in the biotechnology and pharmaceutical industries, and
- · market speculation regarding any of the foregoing.

If we are unable to continue meet the requirements for continued listing on The NASDAQ Global Market, then we may be de-listed. In March of 2010, we received a Staff Determination letter from The NASDAQ Stock Market LLC ("NASDAQ") indicating that we had not regained compliance with the minimum \$1.00 per share requirement for continued inclusion on The NASDAQ Global Market, pursuant to NASDAQ Listing Rule 5450(a)(1). On August 18, 2010, we effected a reverse split of our common shares in order to regain compliance.

We have received negative results from certain of our clinical trials, and we face uncertain results of other clinical trials of our potential products.

In March of 2011, we announced that our Phase 2b trial of XOMA 052 in Type 2 diabetes in 421 patients did not achieve the primary endpoint of reduction in hemoglobin A1c ("HbA1c") after six monthly treatments with XOMA 052 compared to placebo. In June of 2011, we announced top line trial results from our six-month Phase 2a trial of XOMA 052 in Type 2 diabetes in 74 patients, and there were no differences in glycemic control between the drug and placebo groups as measured by HbA1c levels.

Our potential products, including XOMA 052 and XOMA 3AB, will require significant additional research and development, extensive preclinical studies and clinical trials and regulatory approval prior to any commercial sales. This process is lengthy and expensive, often taking a number of years. As clinical results are frequently susceptible to varying interpretations that may delay, limit or prevent regulatory approvals, the length of time necessary to complete clinical trials and to submit an application for marketing approval for a final decision by a regulatory authority varies significantly. As a result, it is uncertain whether:

- · our future filings will be delayed,
- · our preclinical and clinical studies will be successful,
- · we will be successful in generating viable product candidates to targets,
- · we will be able to provide necessary additional data,
- · results of future clinical trials will justify further development, or
- · we will ultimately achieve regulatory approval for any of these product candidates.

For example, in 2003, we completed two Phase 1 trials of XOMA 629, a BPI-derived topical peptide compound targeting acne, evaluating the safety, skin irritation and pharmacokinetics. In January of 2004, we announced the initiation of Phase 2 clinical testing in patients with mild-to-moderate acne. In August of 2004, we announced the results of a Phase 2 trial with XOMA 629 gel. The results were inconclusive in terms of clinical benefit of XOMA 629 compared with vehicle gel. In 2007, after completing an internal evaluation of this program, we chose to reformulate and focus development efforts on the use of this reformulated product candidate in superficial skin infections, including impetigo and the eradication of staphylococcus aureus. In the fourth quarter of 2008, we decided to curtail all spending on XOMA 629 in response to current economic conditions and to focus our financial resources on XOMA 052.

The timing of the commencement, continuation and completion of clinical trials may be subject to significant delays relating to various causes, including completion of preclinical testing and earlier-stage clinical trials in a timely manner, scheduling conflicts with participating clinicians and clinical institutions, difficulties in identifying and enrolling patients who meet trial eligibility criteria, and shortages of available drug supply. Patient enrollment is a function of many factors, including the size of the patient population, the proximity of patients to clinical sites, the eligibility criteria for the trial, the existence of competing clinical trials and the availability of alternative or new treatments. In addition, we will conduct clinical trials in foreign countries in the future which may subject us to further delays and expenses as a result of increased drug shipment costs, additional regulatory requirements and the engagement of foreign clinical research organizations, as well as expose us to risks associated with foreign currency transactions insofar as we might desire to use U.S. dollars to make contract payments denominated in the foreign currency where the trial is being conducted.

All of our product candidates are prone to the risks of failure inherent in drug development. Preclinical studies may not yield results that would satisfactorily support the filing of an Investigational New Drug application ("IND") (or a foreign equivalent) with respect to our product candidates. Even if these applications would be or have been filed with respect to our product candidates, the results of preclinical studies do not necessarily predict the results of clinical trials. Similarly, early-stage clinical trials in healthy volunteers do not predict the results of later-stage clinical trials including the safety and efficacy profiles of any particular product candidates. In addition, there can be no assurance that the design of our clinical trials is focused on appropriate indications, patient populations, dosing regimens or other variables which will result in obtaining the desired efficacy data to support regulatory approval to commercialize the drug. Preclinical and clinical data can be interpreted in different ways. Accordingly, Food and Drug Administration ("FDA") officials from foreign regulatory authorities could interpret the data in different ways than we or our collaboration or development partners do which could delay, limit or prevent regulatory approval.

Administering any of our products or potential products may produce undesirable side effects, also known as adverse effects. Toxicities and adverse effects that we have observed in preclinical studies for some compounds in a particular research and development program may occur in preclinical studies or clinical trials of other compounds from the same program. Such toxicities or adverse effects could delay or prevent the filing of an IND (or a foreign equivalent) with respect to such products or potential products or cause us to cease clinical trials with respect to any drug candidate. In clinical trials, administering any of our products or product candidates to humans may produce adverse effects. These adverse effects could interrupt, delay or halt clinical trials of our products and product candidates and could result in the FDA or other regulatory authorities denying approval of our products or product candidates for any or all targeted indications. The FDA, other regulatory authorities, our collaboration or development partners or we may suspend or terminate clinical trials at any time. Even if one or more of our product candidates were approved for sale, the occurrence of even a limited number of toxicities or adverse effects when used in large populations may cause the FDA to impose restrictions on, or stop, the further marketing of such drugs. Indications of potential adverse effects or toxicities when a drug has been used in large populations or for extended periods of time. Any failure or significant delay in completing preclinical studies or clinical trials may adverse effects and were effects and where the adverse effects or toxicities when a drug has been used in large populations or for extended periods of time. Any failure or significant delay in completing preclinical studies or clinical trials for our product candidates, may severely harm our reputation and business.

In June of 2011, Novartis announced that an advisory committee of the FDA voted in favor of the overall efficacy but not the overall safety of Ilaris® (canakinumab), a fully-human monoclonal antibody that, like XOMA 052, targets IL-1 beta, to treat gouty arthritis attacks in patients who cannot obtain adequate relief with non-steroidal antiinflammatory drugs or colchicine. Novartis also stated that in two pivotal Phase 3 studies of canakinumab in gouty arthritis patients, a higher percentage of patients had adverse events with canakinumab than with the standard treatment for gouty arthritis, and more serious adverse events were reported by patients treated with canakinumab compared to patients receiving the standard treatment. We have not yet determined what impact, if any, this vote and these findings may have on the development of XOMA 052.

Given that regulatory review is an interactive and continuous process, we maintain a policy of limiting announcements and comments upon the specific details of regulatory review of our product candidates, subject to our obligations under the securities laws, until definitive action is taken.

Our therapeutic product candidates have not received regulatory approval. If these product candidates do not receive regulatory approval, neither our third party collaborators nor we will be able to manufacture and market them.

Our product candidates, including XOMA 052 and XOMA 3AB, cannot be manufactured and marketed in the United States and other countries without required regulatory approvals. The United States government and governments of other countries extensively regulate many aspects of our product candidates, including:



- testing,
- manufacturing,
- · promotion and marketing, and
- exporting.

In the United States, the FDA regulates pharmaceutical products under the Federal Food, Drug, and Cosmetic Act and other laws, including, in the case of biologics, the Public Health Service Act. At the present time, we believe that most of our product candidates will be regulated by the FDA as biologics. Initiation of clinical trials requires approval by health authorities. Clinical trials involve the administration of the investigational new drug to healthy volunteers or to patients under the supervision of a qualified principal investigator. Clinical trials must be conducted in accordance with FDA and International Conference on Harmonisation of Technical Requirements for Registration of Pharmaceuticals for Human Use Good Clinical Practices and the European Clinical Trials Directive under protocols that detail the objectives of the study, the parameters to be used to monitor safety and the efficacy criteria to be evaluated. Other national, foreign and local regulations may also apply. The developer of the drug must provide information relating to the characterization and controls of the product before administration to the patient and during the conduct of the trial. In addition, developers of pharmaceutical products must provide periodic data regarding clinical trials to the FDA and other health authorities, and these health authorities may issue a clinical hold upon a trial if they do not believe, or cannot confirm, that the trial can be conducted without unreasonable risk to the trial participants. We cannot assure you that U.S. and foreign health authorities will not issue a clinical hold with respect to any of our clinical trials in the future.

The results of the preclinical studies and clinical testing, together with chemistry, manufacturing and controls information, are submitted to the FDA and other health authorities in the form of a new drug application for a pharmaceutical product, and in the form of a Biologic License Application ("BLA") for a biological product, requesting approval to commence commercial sales. In responding to a new drug application or an antibody license application, the FDA or foreign health authorities may grant marketing approvals, request additional information or further research, or deny the application if it determines that the application does not satisfy its regulatory approval criteria. Regulatory approval of a new drug application, BLA, or supplement is never guaranteed, and the approval process can take several years and is extremely expensive. The FDA and foreign health authorities have substantial discretion in the drug and biologics approval processes. Despite the time and expense incurred, failure can occur at any stage, and we could encounter problems that cause us to abandon clinical trials or to repeat or perform additional preclinical, clinical or manufacturing-related studies.

Changes in the regulatory approval policy during the development period, changes in, or the enactment of additional regulations or statutes, or changes in regulatory review for each submitted product application may cause delays in the approval or rejection of an application. State regulations may also affect our proposed products. The FDA has substantial discretion in both the product approval process and manufacturing facility approval process and, as a result of this discretion and uncertainties about outcomes of testing, we cannot predict at what point, or whether, the FDA will be satisfied with our or our collaborators' submissions or whether the FDA will raise questions which may be material and delay or preclude product approval or manufacturing facility approval. As we accumulate additional clinical data, we will submit it to the FDA, and such data may have a material impact on the FDA product approval process.

Even once approved, a product may be subject to additional testing or significant marketing restrictions, its approval may be withdrawn or it may be voluntarily taken off the market.

Even if the FDA, the European Commission or another regulatory agency approves a product candidate for marketing, the approval may impose ongoing requirements for post-approval studies, including additional research and development and clinical trials, and the FDA, European Commission or other regulatory agency may subsequently withdraw approval based on these additional trials.

Even for approved products, the FDA, European Commission or other regulatory agency may impose significant restrictions on the indicated uses, conditions for use, labeling, advertising, promotion, marketing and/or production of such product.

Furthermore, a marketing approval of a product may be withdrawn by the FDA, the European Commission or another regulatory agency or such a product may be voluntarily withdrawn by the company marketing it based, for example, on subsequently-arising safety concerns. In February of 2009, the European Medicines Agency ("EMA") announced that it had recommended suspension of the marketing authorization of RAPTIVA[®] in the European Union and that its Committee for Medicinal Products for Human Use ("CHMP") had concluded that the benefits of RAPTIVA[®] no longer outweigh its risks because of safety concerns, including the occurrence of progressive multifocal leukoencephalopathy ("PML") in patients taking the medicine. In the second quarter of 2009, Genentech announced and carried out a phased voluntary withdrawal of RAPTIVA[®] from the U.S. market, based on the association of RAPTIVA[®] with an increased risk of PML.

The FDA, European Commission and other agencies also may impose various civil or criminal sanctions for failure to comply with regulatory requirements, including withdrawal of product approval.

Certain of our technologies are relatively new and are in-licensed from third parties, so our capabilities using them are unproven and subject to additional risks.

We license technologies from third parties. These technologies include but are not limited to phage display technologies licensed to us in connection with our bacterial cell expression technology licensing program. However, our experience with some of these technologies remains relatively limited and, to varying degrees, we are still dependent on the licensing parties for training and technical support for these technologies. In addition, our use of these technologies is limited by certain contractual provisions in the licenses relating to them and, although we have obtained numerous licenses, intellectual property rights in the area of phage display are particularly complex. If the owners of the patent rights underlying the technologies we license do not properly maintain or enforce those patents, our competitive position and business prospects could be harmed. Our success will depend in part on the ability of our licensors may fail to maintain existing patents. They may determine not to pursue litigation against other companies that are infringing these patents, or they may pursue such litigation less aggressively than we would. Our licensors may also seek to terminate our license, which could cause us to lose the right to use the license intellectual property and adversely affect our ability to commercialize our technologies, products or services.

We do not know whether there will be, or will continue to be, a viable market for the products in which we have an ownership or royalty interest.

Even if products in which we have an interest receive approval in the future, they may not be accepted in the marketplace. In addition, we or our collaborators or licensees may experience difficulties in launching new products, many of which are novel and based on technologies that are unfamiliar to the healthcare community. We have no assurance that healthcare providers and patients will accept such products, if developed. For example, physicians and/or patients may not accept a product for a particular indication because it has been biologically derived (and not discovered and developed by more traditional means) or if no biologically derived products are currently in widespread use in that indication. Similarly, physicians may not accept a product if they believe other products to be more effective or are more comfortable prescribing other products.

Safety concerns may also arise in the course of on-going clinical trials or patient treatment as a result of adverse events or reactions. For example, in February of 2009, the EMA announced that it had recommended suspension of the marketing authorization of RAPTIVA® in the European Union and EMD Serono Inc., the company that marketed RAPTIVA® in Canada ("EMD Serono") announced that, in consultation with Health Canada, the Canadian health authority ("Health Canada"), it would suspend marketing of RAPTIVA® in Canada. In March of 2009, Merck Serono Australia Pty Ltd, the company that marketed RAPTIVA® in Australia ("Merck Serono Australia"), following a recommendation from the Therapeutic Goods Administration, the Australian health authority ("TGA"), announced that it was withdrawing RAPTIVA® from the Australian market. In the second quarter of 2009, Genentech announced and carried out a phased voluntary withdrawal of RAPTIVA® from the U.S. market, based on the association of RAPTIVA® with an increased risk of PML. As a result, sales of RAPTIVA® ceased in the second quarter of 2009.

Furthermore, government agencies, as well as private organizations involved in healthcare, from time to time publish guidelines or recommendations to healthcare providers and patients. Such guidelines or recommendations can be very influential and may adversely affect the usage of any products we may develop directly (for example, by recommending a decreased dosage of a product in conjunction with a concomitant therapy or a government entity withdrawing its recommendation to screen blood donations for certain viruses) or indirectly (for example, by recommending a competitive product over our product). Consequently, we do not know if physicians or patients will adopt or use our products for their approved indications.

We or our third party collaborators or licensees may not have adequate manufacturing capacity sufficient to meet market demand.

If any of our product candidates are approved, because we have never commercially introduced any pharmaceutical products, we do not know whether the capacity of our existing manufacturing facilities can be increased to produce sufficient quantities of our products to meet market demand. Also, if we or our third party collaborators or licensees need additional manufacturing facilities to meet market demand, we cannot predict that we will successfully obtain those facilities because we do not know whether they will be available on acceptable terms. In addition, any manufacturing facilities acquired or used to meet market demand must meet the FDA's quality assurance guidelines.

Our agreements with third parties, many of which are significant to our business, expose us to numerous risks.

Our financial resources and our marketing experience and expertise are limited. Consequently, our ability to successfully develop products depends, to a large extent, upon securing the financial resources and/or marketing capabilities of third parties.

- In April of 1996, we entered into an agreement with Genentech whereby we agreed to co-develop Genentech's humanized monoclonal antibody product RAPTIVA[®]. In April of 1999, March of 2003, and January of 2005, the companies amended the agreement. In October of 2003, RAPTIVA[®] was approved by the FDA for the treatment of adults with chronic moderate-to-severe plaque psoriasis who are candidates for systemic therapy or phototherapy and, in September of 2004, Merck Serono announced the product's approval in the European Union. In January of 2005, we entered into a restructuring of our collaboration agreement with Genentech which ended our existing cost and profit sharing arrangement related to RAPTIVA[®] in the United States and entitled us to a royalty interest on worldwide net sales. In February of 2009, the EMA announced that it had recommended suspension of the marketing authorization of RAPTIVA[®] in the European Union and EMD Serono announced that, in consultation with Health Canada, it would suspend marketing of RAPTIVA[®] in Canada. In March of 2009, Merck Serono Australia, following a recommendation from the TGA, announced that it was withdrawing RAPTIVA[®] from the Australian market. In the second quarter of 2009, Genentech announced and carried out a phased voluntary withdrawal of RAPTIVA[®] from the U.S. market, based on the association of RAPTIVA[®] with an increased risk of PML. As a result, sales of RAPTIVA[®] ceased in the second quarter of 2009.
- In March of 2004, we announced we had agreed to collaborate with Chiron Corporation (now Novartis) for the development and commercialization of antibody products for the treatment of cancer. In April of 2005, we announced the initiation of clinical testing of the first product candidate out of the collaboration, HCD122, an anti-CD40 antibody, in patients with advanced chronic lymphocytic leukemia. In October of 2005, we announced the initiation of the second clinical trial of HCD122 in patients with multiple myeloma. In November of 2008, we announced the restructuring of this product development collaboration, which involves six development programs including the ongoing HCD122 and LFA102 programs. In exchange for cash and debt reduction on our existing loan facility with Novartis, Novartis has control over the HCD122 and LFA102 programs and the additional ongoing program, as well as the right to expand the development of these programs into additional indications outside of oncology.
- In March of 2005, we entered into a contract with the National Institute of Allergy and Infectious Diseases ("NIAID") to produce three monoclonal antibodies designed to protect United States citizens against the harmful effects of botulinum neurotoxin used in bioterrorism. In July of 2006, we entered into an additional contract with NIAID for the development of an appropriate formulation for human administration of these three antibodies in a single injection. In September of 2008, we announced that we were awarded an additional contract with NIAID to support our on-going development of drug candidates toward clinical trials in the treatment of botulism poisoning.
- In December of 2010, we entered into a license and collaboration agreement with Servier, to jointly develop and commercialize XOMA 052 in multiple indications. Under the terms of the agreement, Servier has worldwide rights to diabetes and cardiovascular disease indications and rights outside the U.S. and Japan to Behcet's uveitis and other inflammatory and oncology indications. We retain development and commercialization rights for Behcet's uveitis and other inflammatory and oncology indications. We retain development and commercialization rights for Behcet's uveitis and other inflammatory disease and oncology indications in the U.S. and Japan, and has an option to reacquire rights to diabetes and cardiovascular disease indications from Servier in these territories. Should we exercise this option, we will be required to pay Servier an option fee and partially reimburse their incurred development expenses. The agreement contains customary termination rights relating to matters such as material breach by either party, safety issues and patents. Servier also has a unilateral right to terminate the agreement on a country-by-country basis or in its entirety on six months' notice.
- In December of 2010, we also entered into a loan agreement with Servier, which provides for an advance of up to €15.0 million and was fully funded in January of 2011 with the proceeds converting to approximately \$19.5 million using the January 13, 2011 Euro to USD exchange rate. This loan is secured by an interest in our intellectual property rights to all XOMA 052 indications worldwide, excluding the U.S. and Japan. The loan has a final maturity date in 2016; however, after a specified period prior to final maturity, the loan is required to be repaid (i) at Servier's option, by applying up to a significant percentage of any milestone or royalty payments owed by Servier under our collaboration agreement and (ii) using a significant percentage of any upfront, milestone or royalty payments we receive from any third party collaboration or development partner for rights to XOMA 052 in the U.S. and/or Japan. In addition, the loan becomes immediately due and payable upon certain customary events of default. At June 30, 2011, the €15.0 million outstanding principal balance under this loan agreement would have equaled approximately \$21.6 million using the June 30, 2011 Euro to USD exchange rate.

We have licensed our bacterial cell expression technology, an enabling technology used to discover and screen, as well as develop and manufacture, recombinant antibodies and other proteins for commercial purposes, to over 50 companies. As of June 30, 2011, we were aware of two antibody products manufactured using this technology that have received FDA approval, Genentech's LUCENTIS[®] (ranibizumab injection) for treatment of neovascular wet age-related macular degeneration and UCB's CIMZIA[®] (certolizumab pegol) for treatment of Crohn's disease and rheumatoid arthritis. In the third quarter of 2009, we sold our LUCENTIS[®] royalty interest to Genentech. In the third quarter of 2010, we sold our CIMZIA[®] royalty interest.

Because our collaborators and licensees are independent third parties, they may be subject to different risks than we are and have significant discretion in, and different criteria for, determining the efforts and resources they will apply related to their agreements with us. If these collaborators and licensees do not successfully develop and market these products, we may not have the capabilities, resources or rights to do so on our own. We do not know whether we, our collaborators or licensees will successfully develop and market any of the products that are or may become the subject of any of our collaboration or licensing arrangements. In some cases these arrangements provide for funding solely by our collaborators or licensees, and in other cases, such as our arrangement with Servier, all of the funding for certain projects and a significant portion of the funding for other projects is to be provided by our collaborator or licensee. In addition, third party arrangements such as ours also increase uncertainties in the related decision-making processes and resulting progress under the arrangements, as we and our collaborators or licensees may reach different conclusions, or support different paths forward, based on the same information, particularly when large amounts of technical data are involved. Furthermore, our contracts with NIAID contain numerous standard terms and conditions provided for in the applicable federal acquisition regulations and customary in many government contracts. Uncertainty exists as to whether we will be able to comply with these terms and conditions in a timely manner, if at all. In addition, we are uncertain as to the extent of NIAID's demands and the flexibility that will be granted to us in meeting those demands.

Even when we have a collaborative relationship, other circumstances may prevent it from resulting in successful development of marketable products.

- In September of 2004, we entered into a collaboration arrangement with Aphton Corporation ("Aphton") for the treatment of gastrointestinal and other gastrinsensitive cancers using anti-gastrin monoclonal antibodies. In January of 2006, Aphton announced that its common stock had been delisted from NASDAQ. In May of 2006, Aphton filed for bankruptcy protection under Chapter 11, Title 11 of the United States Bankruptcy Code.
- In September of 2006, we entered into an agreement with Taligen Therapeutics, Inc. ("Taligen") which formalized an earlier letter agreement, which was signed in May of 2006, for the development and cGMP manufacture of a novel antibody fragment for the potential treatment of inflammatory diseases. In May of 2007, we and Taligen entered into a letter agreement which provided that we would not produce a cGMP batch at clinical scale pursuant to the terms of the agreement entered into in September of 2006. In addition, the letter agreement provided that we would conduct and complete the technical transfer of the process to Avecia Biologics Limited or its designated affiliate ("Avecia"). The letter agreement also provided that, subject to payment by Taligen of approximately \$1.7 million, we would grant to Avecia a non-exclusive, worldwide, paid-up, non-transferable, non-sublicensable, perpetual license under our owned project innovations. We received \$0.6 million as the first installment under the payment terms of the letter agreement but not the two additional payments totaling approximately \$1.1 million to which we were entitled upon fulfillment of certain obligations. In May of 2009, the matter was resolved by agreement of the parties in a manner that had no further impact on our financial position.

Although we continue to evaluate additional strategic alliances and potential partnerships, we do not know whether or when any such alliances or partnerships will be entered into.

Products and technologies of other companies may render some or all of our products and product candidates noncompetitive or obsolete.

Developments by others may render our products, product candidates, or technologies obsolete or uncompetitive. Technologies developed and utilized by the biotechnology and pharmaceutical industries are continuously and substantially changing. Competition in the areas of genetically engineered DNA-based and antibody-based technologies is intense and expected to increase in the future as a number of established biotechnology firms and large chemical and pharmaceutical companies advance in these fields. Many of these competitors may be able to develop products and processes competitive with or superior to our own for many reasons, including that they may have:

- · significantly greater financial resources,
- · larger research and development and marketing staffs,
- · larger production facilities,
- · entered into arrangements with, or acquired, biotechnology companies to enhance their capabilities, or
- extensive experience in preclinical testing and human clinical trials.

These factors may enable others to develop products and processes competitive with or superior to our own or those of our collaborators. In addition, a significant amount of research in biotechnology is being carried out in universities and other non-profit research organizations. These entities are becoming increasingly interested in the commercial value of their work and may become more aggressive in seeking patent protection and licensing arrangements. Furthermore, many companies and universities tend not to announce or disclose important discoveries or development programs until their patent position is secure or, for other reasons, later; as a result, we may not be able to track development of competitive products, particularly at the early stages. Positive or negative developments in connection with a potentially competing product may have an adverse impact on our ability to raise additional funding on acceptable terms. For example, if another product is perceived to have a competitive advantage, or another product's failure is perceived to increase the likelihood that our product will fail, then investors may choose not to invest in us on terms we would accept or at all.

The examples below pertain to competitive events in the market which we review quarterly and are not intended to be representative of all existing competitive events.

XOMA 052

We, in collaboration with Servier, are conducting clinical testing of XOMA 052, a potent anti-inflammatory monoclonal antibody targeting IL-1 beta, in Behcet's uveitis patients, Type 2 diabetes patients and cardiovascular disease patients. Other companies are developing other products based on the same or similar therapeutic targets as XOMA 052 and these products may prove more effective than XOMA 052. We are aware that:

- In June of 2009, Novartis announced it had received U.S. marketing approval for Ilaris® (canakinumab), a fully-human monoclonal antibody targeting IL-1 beta, to treat children and adults with Cryopyrin-Associated Periodic Syndromes ("CAPS"). In October of 2009, Novartis announced that Ilaris® had been approved in the European Union for CAPS. Canakinumab is also in clinical trials for cardiovascular risk reduction, and in Type 2 diabetes, certain forms of gout and systemic juvenile rheumatoid arthritis, among other conditions. In January of 2011, Novartis announced that it had filed for EMA approval of Ilaris® for the treatment and prevention of gout. In June of 2011, Novartis announced that an advisory committee of the FDA voted in favor of the overall efficacy but not the overall safety of canakinumab to treat gouty arthritis attacks in patients who cannot obtain adequate relief with non-steroidal anti-inflammatory drugs or colchicine.
- Eli Lilly and Company ("Lilly") is developing LY2189102, an investigational IL-1 beta antibody, for subcutaneous injection for the treatment of Type 2 diabetes. In June of 2011, Lily disclosed at a scientific conference that, in a double-blind, placebo controlled Phase 2 study of 106 patients with Type 2 diabetes, a significant (p<0.05), early reduction in C reactive protein occurred, HbA1c was moderately reduced and anti-inflammatory effects were shown.
- In 2008, Biovitrum AB (now called Swedish Orphan Biovitrum, "Biovitrum") obtained a worldwide exclusive license to Amgen Inc.'s ("Amgen") Kineret[®] (anakinra) for its current approved indication. Kineret[®] is an IL-1 receptor antagonist (IL-1ra) currently marketed to treat rheumatoid arthritis and has been evaluated over the years in multiple IL-1 mediated diseases, including Type 2 diabetes and other indications we are considering for XOMA 052. In addition to other on-going studies, a proof-of-concept clinical trial in the United Kingdom investigating Kineret[®] in patients with a certain type of myocardial infarction, or heart attack, has been completed. In August of 2010, Biovitrum announced that the FDA had granted orphan drug designation to Kineret[®] for the treatment of CAPS.

- In February of 2008, Regeneron Pharmaceuticals, Inc. ("Regeneron") announced it had received marketing approval from the FDA for ARCALYST[®] (rilonacept) Injection for Subcutaneous Use, an interleukin-1 blocker or IL-1 Trap, for the treatment of CAPS, including Familial Cold Auto-inflammatory Syndrome and Muckle-Wells Syndrome in adults and children 12 and older. In September of 2009, Regeneron announced that rilonacept was approved in the European Union for CAPS. In June of 2010 and February of 2011, Regeneron announced positive results of two Phase 3 clinical trials of rilonacept in gout. In March of 2011, Regeneron disclosed that it intends to file a supplemental BLA for ARCALYST[®] for the prevention and treatment of gout.
- Amgen has been developing AMG 108, a fully-human monoclonal antibody that targets inhibition of the action of IL-1. In April of 2008, Amgen discussed
 results from its recently completed Phase 2 study in rheumatoid arthritis. AMG 108 showed statistically significant improvement in the signs and symptoms of
 rheumatoid arthritis and was well tolerated. In January of 2011, MedImmune, the worldwide biologics unit for AstraZeneca PLC, announced that Amgen granted
 it rights to develop AMG 108 worldwide except in Japan.
- In June of 2009, Cytos Biotechnology AG announced the initiation of an ascending dose Phase 1/2a study of CYT013-IL1bQb, a therapeutic vaccine targeting IL-1 beta, in Type 2 diabetes. In 2010, this study was extended to include two additional groups of patients.

XOMA 3AB

We are also developing XOMA 3AB, a combination, or cocktail, of antibodies designed to neutralize the most potent of botulinum toxins. Other companies are developing other products targeting botulism poisoning and these products may prove more effective than XOMA 3AB. We are aware that:

- In May of 2006, the U.S. Department of Health & Human Services ("DHHS") awarded Cangene Corporation ("Cangene") a five-year, \$362.0 million contract under Project Bioshield. The contract requires Cangene to manufacture and supply 200,000 doses of an equine heptavalent botulism anti-toxin to treat individuals who have been exposed to the toxins that cause botulism. In May of 2008, Cangene announced significant product delivery under this contract. In March of 2010, this contract was extended for an additional two years, until May of 2013. In June of 2011, Cangene announced that DHHS will exercise options under the supply contract which are expected to increase the total contract to \$423.0 million and to extend the delivery schedule to 2018.
- Emergent BioSolutions, Inc. ("Emergent") is currently in development of a botulism immunoglobulin candidate that may compete with our anti-botulinum neurotoxin monoclonal antibodies.
- We are aware of additional companies that are pursuing biodefense-related antibody products. PharmAthene, Inc. and Human Genome Sciences, Inc. are developing anti-anthrax antibodies. Cangene and Emergent are developing anti-anthrax immune globulin products. These products may compete with our efforts in the areas of other monoclonal antibody-based biodefense products and the manufacture of antibodies to supply strategic national stockpiles.

Manufacturing risks and inefficiencies may adversely affect our ability to manufacture products for ourselves or others.

To the extent we continue to provide manufacturing services for our own benefit or to third parties, we are subject to manufacturing risks. Additionally, unanticipated fluctuations in customer requirements have led and may continue to lead to manufacturing inefficiencies, which if significant could lead to an impairment on our long-lived assets or restructuring activities. We must utilize our manufacturing operations in compliance with regulatory requirements, in sufficient quantities and on a timely basis, while maintaining acceptable product quality and manufacturing costs. Additional resources and changes in our manufacturing processes may be required for each new product, product modification or customer or to meet changing regulatory or third party requirements, and this work may not be successfully or efficiently completed. In addition, to the extent we continue to provide manufacturing services, our fixed costs, such as facility costs, would be expected to increase, as would necessary capital investment in equipment and facilities.

Manufacturing and quality problems may arise in the future to the extent we continue to perform these services for our own benefit or for third parties. Consequently, our development goals or milestones may not be achieved in a timely manner or at a commercially reasonable cost, or at all. In addition, to the extent we continue to make investments to improve our manufacturing operations, our efforts may not yield the improvements that we expect.

Because many of the companies we do business with are also in the biotechnology sector, the volatility of that sector can affect us indirectly as well as directly.

As a biotechnology company that collaborates with other biotechnology companies, the same factors that affect us directly can also adversely impact us indirectly by affecting the ability of our collaborators, partners and others we do business with to meet their obligations to us and reduce our ability to realize the value of the consideration provided to us by these other companies.

For example, in connection with our licensing transactions relating to our bacterial cell expression technology, we have in the past and may in the future agree to accept equity securities of the licensee in payment of license fees. The future value of these or any other shares we receive is subject both to market risks affecting our ability to realize the value of these shares and more generally to the business and other risks to which the issuer of these shares may be subject.

As we do more business internationally, we will be subject to additional political, economic and regulatory uncertainties.

We may not be able to successfully operate in any foreign market. We believe that, because the pharmaceutical industry is global in nature, international activities will be a significant part of our future business activities and that, when and if we are able to generate income, a substantial portion of that income will be derived from product sales and other activities outside the United States. Foreign regulatory agencies often establish standards different from those in the United States, and an inability to obtain foreign regulatory approvals on a timely basis could put us at a competitive disadvantage or make it uneconomical to proceed with a product or product candidate's development. International operations and sales may be limited or disrupted by:

- · imposition of government controls,
- · export license requirements,
- · political or economic instability,
- · trade restrictions,
- · changes in tariffs,
- · restrictions on repatriating profits,
- exchange rate fluctuations,
- · withholding and other taxation, and
- · difficulties in staffing and managing international operations.

We are subject to foreign currency exchange rate risks.

We are subject to foreign currency exchange rate risks because substantially all of our revenues and operating expenses are paid in U.S. dollars, but we pay interest and principal obligations with respect to our loan from Servier in Euros. To the extent that the U.S. dollar declines in value against the Euro, the effective cost of servicing our Euro-denominated debt will be higher. Changes in the exchange rate result in foreign currency gains or losses. Although we have managed some of our exposure to changes in foreign currency exchange rates by entering into foreign exchange option contracts, there can be no assurance that foreign currency fluctuations will not have a material adverse effect on our business, financial condition, liquidity or results of operations. In addition, our foreign exchange option contracts are re-valued at each financial reporting period, which may also result in gains or losses from time to time.

If we and our partners are unable to protect our intellectual property, in particular our patent protection for our principal products, product candidates and processes, and prevent its use by third parties, our ability to compete in the market will be harmed, and we may not realize our profit potential.

We rely on patent protection, as well as a combination of copyright, trade secret, and trademark laws to protect our proprietary technology and prevent others from duplicating our products or product candidates. However, these means may afford only limited protection and may not:

- · prevent our competitors from duplicating our products,
- · prevent our competitors from gaining access to our proprietary information and technology, or
- · permit us to gain or maintain a competitive advantage.



Because of the length of time and the expense associated with bringing new products to the marketplace, we and our collaboration and development partners hold and are in the process of applying for a number of patents in the United States and abroad to protect our product candidates and important processes and also have obtained or have the right to obtain exclusive licenses to certain patents and applications filed by others. However, the mere issuance of a patent is not conclusive as to its validity or its enforceability. The United States Federal Courts or equivalent national courts or patent offices elsewhere may invalidate our patents or find them unenforceable. In addition, the laws of foreign countries may not protect our intellectual property rights effectively or to the same extent as the laws of the United States. If our intellectual property rights are not adequately protected, we may not be able to commercialize our technologies, products, or services, and our competitors could commercialize our technologies, which could result in a decrease in our sales and market share that would harm our business and operating results. Specifically, the patent position of biotechnology companies generally is highly uncertain and involves complex legal and factual questions. The legal standards governing the validity of biotechnology patents are in transition, and current defenses as to issued biotechnology patents may not be adequate in the future. Accordingly, there is uncertainty as to:

- whether any pending or future patent applications held by us will result in an issued patent, or that if patents are issued to us, that such patents will provide meaningful protection against competitors or competitive technologies,
- whether competitors will be able to design around our patents or develop and obtain patent protection for technologies, designs or methods that are more effective than those covered by our patents and patent applications, or
- the extent to which our product candidates could infringe on the intellectual property rights of others, which may lead to costly litigation, result in the payment of substantial damages or royalties, and/or prevent us from using technology that is essential to our business.

We have established an extensive portfolio of patents and applications, both United States and foreign, related to our BPI-related product candidates, including novel compositions, their manufacture, formulation, assay and use. We have also established a portfolio of patents, both United States and foreign, related to our bacterial cell expression technology, including claims to novel promoter sequences, secretion signal sequences, compositions and methods for expression and secretion of recombinant proteins from bacteria, including immunoglobulin gene products. Most of the more important European patents in our bacterial cell expression patent portfolio expired in July of 2008.

If certain patents issued to others are upheld or if certain patent applications filed by others issue and are upheld, we may require licenses from others in order to develop and commercialize certain potential products incorporating our technology or we may become involved in litigation to determine the proprietary rights of others. These licenses, if required, may not be available on acceptable terms, and any such litigation may be costly and may have other adverse effects on our business, such as inhibiting our ability to compete in the marketplace and absorbing significant management time.

Due to the uncertainties regarding biotechnology patents, we also have relied and will continue to rely upon trade secrets, know-how and continuing technological advancement to develop and maintain our competitive position. All of our employees have signed confidentiality agreements under which they have agreed not to use or disclose any of our proprietary information. Research and development contracts and relationships between us and our scientific consultants and potential customers provide access to aspects of our know-how that are protected generally under confidentiality agreements. These confidentiality agreements may be breached or may not be enforced by a court. To the extent proprietary information is divulged to competitors or to the public generally, such disclosure may adversely affect our ability to develop or commercialize our products by giving others a competitive advantage or by undermining our patent position.

Litigation regarding intellectual property can be costly and expose us to risks of counterclaims against us.

We may be required to engage in litigation or other proceedings to protect our intellectual property. The cost to us of this litigation, even if resolved in our favor, could be substantial. Such litigation could also divert management's attention and resources. In addition, if this litigation is resolved against us, our patents may be declared invalid, and we could be held liable for significant damages. In addition, we may be subject to a claim that we are infringing another party's patent. If such claim is resolved against us, we or our collaborators may be enjoined from developing, manufacturing, selling or importing products, processes or services unless we obtain a license from the other party. Such license may not be available on reasonable terms, thus preventing us from using these products, processes or services and adversely affecting our revenue.

Even if we or our third party collaborators or licensees bring products to market, we may be unable to effectively price our products or obtain adequate reimbursement for sales of our products, which would prevent our products from becoming profitable.

If we or our third party collaborators or licensees succeed in bringing our product candidates to the market, they may not be considered cost-effective, and reimbursement to the patient may not be available or may not be sufficient to allow us to sell our products on a competitive basis. In both the United States and elsewhere, sales of medical products and treatments are dependent, in part, on the availability of reimbursement to the patient from third-party payors, such as government and private insurance plans. Third-party payors are increasingly challenging the prices charged for pharmaceutical products and services. Our business is affected by the efforts of government and third-party payors to contain or reduce the cost of healthcare through various means. In the United States, there have been and will continue to be a number of federal and state proposals to implement government controls on pricing. In addition, the emphasis on managed care in the United States has increased and will continue to increase the pressure on the pricing of pharmaceutical products. We cannot predict whether any legislative or regulatory proposals will be adopted or the effect these proposals or managed care efforts may have on our business.

Healthcare reform measures and other statutory or regulatory changes could adversely affect our business.

In both the United States and certain foreign jurisdictions, there have been a number of legislative and regulatory proposals to change the healthcare system in ways that could impact our business. In March of 2010, the U.S. Congress enacted and President Obama signed into law the Patient Protection and Affordable Care Act, which includes a number of healthcare reform provisions. Assuming the new law survives recent calls for its repeal, the reforms imposed by the new law would significantly impact the pharmaceutical industry, most likely in the area of pharmaceutical product pricing; however, the full effects of new law cannot be known until these provisions are implemented and the relevant federal and state agencies issue applicable regulations or guidance.

The pharmaceutical and biotechnology industries are subject to extensive regulation, and from time to time legislative bodies and governmental agencies consider changes to such regulations that could have significant impact on industry participants. For example, in light of certain highly-publicized safety issues regarding certain drugs that had received marketing approval, the U.S. Congress has considered various proposals regarding drug safety, including some which would require additional safety studies and monitoring and could make drug development more costly. We are unable to predict what additional legislation or regulation, if any, relating to safety or other aspects of drug development may be enacted in the future or what effect such legislation or regulation would have on our business.

The business and financial condition of pharmaceutical and biotechnology companies are also affected by the efforts of governments, third-party payors and others to contain or reduce the costs of healthcare to consumers. In the United States and various foreign jurisdictions there have been, and we expect that there will continue to be, a number of legislative and regulatory proposals aimed at changing the healthcare system, such as proposals relating to the reimportation of drugs into the U.S. from other countries (where they are then sold at a lower price) and government control of prescription drug pricing. The pendency or approval of such proposals could result in a decrease in our share price or limit our ability to raise capital or to obtain strategic collaborations or licenses.

We are exposed to an increased risk of product liability claims, and a series of related cases is currently pending against us.

The testing, marketing and sales of medical products entails an inherent risk of allegations of product liability. In the event of one or more large, unforeseen awards of damages against us, our product liability insurance may not provide adequate coverage. A significant product liability claim for which we were not covered by insurance would have to be paid from cash or other assets. To the extent we have sufficient insurance coverage, such a claim would result in higher subsequent insurance rates. In addition, product liability claims can have various other ramifications including loss of future sales opportunities, increased costs associated with replacing products, and a negative impact on our goodwill and reputation, which could also adversely affect our business and operating results.

On April 9, 2009, a complaint was filed in the Superior Court of Alameda County, California, in a lawsuit captioned Hedrick et al. v. Genentech, Inc. et al, Case No. 09-446158. The complaint asserts claims against Genentech, us and others for alleged strict liability for failure to warn, strict product liability, negligence, breach of warranty, fraudulent concealment, wrongful death and other claims based on injuries alleged to have occurred as a result of three individuals' treatment with RAPTIVA®. The complaint seeks unspecified compensatory and punitive damages. Since then, additional complaints have been filed in the same court, bringing the total number of pending cases to seventy five. The cases have been consolidated as a coordinated proceeding. All of the complaints allege the same claims and seek the same types of damages based on injuries alleged to have occurred as a result of the plaintiffs' treatment with RAPTIVA®. On July 15, 2011, the Court dismissed with prejudice one of the cases in this coordinated proceeding, White v. Genentech, Inc., et al, Case No. RG-09-484026. Even though Genentech has agreed to indemnify us in connection with these matters, there can be no assurance that these or other products liability lawsuits will not result in liability to us or that our insurance or contractual arrangements will provide us with adequate protection against such liabilities.

On August 4, 2010, a petition was filed in the District Court of Dallas County, Texas in a case captioned McCall v. Genentech, Inc., et al., No. 10-09544. The defendants filed a Notice of Removal to the United States District Court for the Northern District of Texas on September 3, 2010. The removed case is captioned McCall v. Genentech, Inc., et al., No. 3:10-cv-01747-B. The parties have fully briefed the plaintiff's Motion to Remand and are awaiting a final ruling from the Court. The petition asserts personal injury claims against Genentech, us and others arising out of the plaintiff's treatment with RAPTIVA®. The petition alleges claims based on negligence, strict liability, misrepresentation and suppression, conspiracy, and actual and constructive fraud. The petition seeks compensatory damages and punitive damages in an unspecified amount. On June 6, 2011, the Court dismissed plaintiff's claims of negligent misrepresentation, fraud, and conspiracy. Even though Genentech has agreed to indemnify us in connection with this matter, there can be no assurance that this or other products liability lawsuits will not result in liability to us or that our insurance or contractual arrangements will provide us with adequate protection against such liabilities.

On January 7, 2011, a complaint was filed in the United States District Court for the Northern District of Texas in a case captioned Massa v. Genentech, Inc., et al., No. 4:11CV70. On January 11, 2011, a complaint was filed in the United States District Court for the District of Massachusetts in a case captioned Sylvia, et al. v. Genentech, Inc., et al., No. 1:11-cv-10054-MLW. These two complaints allege the same claims against Genentech, us and others and seek the same types of damages as the complaints filed in the Superior Court of Alameda County, California referenced above. Even though Genentech has agreed to indemnify us in connection with these matters, there can be no assurance that these or other products liability lawsuits will not result in liability to us or that our insurance or contractual arrangements will provide us with adequate protection against such liabilities.

On April 8, 2011, four complaints were filed in the United States District Court for the Eastern District of Michigan. The cases are captioned: Muniz v. Genentech, et al., 5:11-cv-11489-JCO-RSW; Tifenthal v. Genentech, et al., 2:11-cv-11489-DPH-LJM; Blair v. Genentech, et al., 2:11-cv-11463-SFC-MJH; and Marsh v. Genentech, et al., 2:11-cv-11462-RHC-MKM. The complaints allege the same claims against Genentech, us and others and seek the same types of damages as the complaints filed in the Superior Court of Alameda County, California referenced above. All four cases have been transferred to the United States District Court for the Western District of Michigan. Genentech and we have filed Motions to Dismiss all four actions and are awaiting a decision from the Court. Even though Genentech has agreed to indemnify us in connection with these matters, there can be no assurance that these or other products liability lawsuits will not result in liability to us or that our insurance or contractual arrangements will provide us with adequate protection against such liabilities.

The loss of key personnel, including our Chief Executive Officer, could delay or prevent achieving our objectives.

Our research, product development and business efforts could be adversely affected by the loss of one or more key members of our scientific or management staff, particularly our executive officers: Steven B. Engle, our Chairman, Chief Executive Officer and President; Patrick J. Scannon, M.D., Ph.D., our Executive Vice President and Chief Scientific Officer; Fred Kurland, our Vice President, Finance and Chief Financial Officer; and Christopher J. Margolin, our Vice President, General Counsel and Secretary. We currently have no key person insurance on any of our employees.

A U.S. holder of our common shares or warrants could be subject to material adverse U.S. federal income tax consequences if we were considered to be a PFIC at any time during the U.S. holder's holding period.

A non-U.S. corporation generally will be a "passive foreign investment company," or PFIC, for U.S. federal income tax purposes in any taxable year in which, after applying the relevant look-through rules with respect to the income and assets of its subsidiaries, either 75% or more of its gross income is "passive income" (generally including (without limitation) dividends, interest, annuities and certain royalties and rents not derived in the active conduct of a business) or the average value of its assets that produce passive income or are held for the production of passive income is at least 50% of the total value of its assets. In determining whether we meet the 50% test, cash is considered a passive asset and the total value of our assets generally will be treated as equal to the sum of the aggregate fair market value of our outstanding common shares plus our liabilities. If we own at least 25% (by value) of the stock of another corporation, we will be treated, for purposes of the PFIC tests, as owning our proportionate share of the other corporation's income.

We believe that we were not a PFIC for the 2010 taxable year. However, because PFIC status is determined annually and depends on the composition of a company's income and assets and the fair market value of its assets (including goodwill), which may be volatile in our industry, there can be no assurance that we will not be considered a PFIC for 2011 or any subsequent year. For example, taking into account our existing cash balances, if the value of our common shares were to decline materially, it is possible that we could become a PFIC in 2011 or a subsequent year. Additionally, due to the complexity of the PFIC provisions and the limited authority available to interpret such provisions, there can be no assurance that our determination regarding our PFIC status could not be successfully challenged by the Internal Revenue Service ("IRS").

If we were found to be a PFIC for any taxable year in which a U.S. holder (as defined below) held common shares or warrants, certain adverse U.S. federal income tax consequences could apply to such U.S. holder, including a recharacterization of any capital gain recognized on a sale or other disposition of common shares or warrants as ordinary income, ineligibility for any preferential tax rate otherwise applicable to any "qualified dividend income," a material increase in the amount of tax that such U.S. holder would owe and the possible imposition of interest charges, an imposition of tax earlier than would otherwise be imposed and additional tax form filing requirements.

For purposes of this discussion, the term "U.S. holder" means a beneficial owner of common shares or warrants that is, for U.S. federal income tax purposes, (i) an individual who is a U.S. citizen or resident, (ii) a corporation (or other entity treated as a corporation for U.S. federal income tax purposes) created or organized in or under the laws of the United States, any state thereof or the District of Columbia, (iii) an estate the income of which is includable in gross income for U.S. income tax purposes regardless of its source, or (iv) a trust, if a U.S. court is able to exercise primary supervision over the administration of the trust and one or more U.S. fiduciaries have the authority to control all substantial decisions of the trust, or if the trust has a valid election in effect under applicable U.S. Treasury Regulations to be treated as a U.S. person. Special rules apply to a U.S. investor who owns our common shares or warrants through an entity treated as a partnership for U.S. federal income tax purposes.

A U.S. holder owning shares in a PFIC (or a corporation that might become a PFIC) might be able to mitigate the adverse tax consequences of PFIC status by making certain elections, including "qualified electing fund" (a "QEF") or "mark-to-market" elections, if deemed appropriate based on guidance provided by the U.S. holder's tax advisor. However, it should be noted that (1) the beneficial effect of a QEF election or a mark-to-market election may be substantially diminished if such election is not made from the inception of a U.S. holder's holding period (a "Year One Election"), (2) neither a QEF election nor a mark-to-market election can be made with respect to warrants, (3) a Year One Election generally cannot be made for any common shares received upon exercise of warrants ("Warrant Shares") because the holding period of Warrant Shares is deemed, for QEF election and mark-to-market election or mark-to-market election or mark-to-market election will not be effective until the underlying warrants are exercised, and (4) a QEF election or a mark-to-market election is made on a shareholder-by-shareholder basis and, once made, can only be revoked with the consent of the IRS.

The PFIC rules are very complex, as are the requirements and effects of the various elections designed to mitigate the adverse consequences of the PFIC rules. A U.S. holder should consult its own tax advisor regarding the PFIC rules, including the foregoing limitations on the ability to make a QEF election or a mark-to-market election (or to qualify either such election as a Year One Election), the timing requirements with respect to the various elections and the irrevocability of certain elections (absent the consent of the IRS).

As a result of a recent legislative change, a U.S. holder generally will be required to file IRS Form 8621 if the U.S. holder holds our common shares or warrants in any taxable year in which we are classified as a PFIC (whether or not a QEF or mark-to-market election is made).

Our ability to use our net operating loss carry-forwards and other tax attributes will be substantially limited by Section 382 of the Internal Revenue Code.

Section 382 of the Internal Revenue Code of 1986, as amended, generally limits the ability of a corporation that undergoes an "ownership change" to utilize its net operating loss carry-forwards ("NOLs") and certain other tax attributes against any taxable income in taxable periods after the ownership change. The amount of taxable income in each taxable year after the ownership change that may be offset by pre-change NOLs and certain other pre-change tax attributes is generally equal to the product of (a) the fair market value of the corporation's outstanding shares immediately prior to the ownership change and (b) the long-term tax exempt rate (i.e., a rate of interest established by the IRS that fluctuates from month to month). In general, an "ownership change" occurs whenever the percentage of the shares of a corporation owned, directly or indirectly, by "5-percent shareholders" (within the meaning of Section 382 of the Internal Revenue Code) increases by more than 50 percentage points over the lowest percentage of the shares of such corporation owned, directly or indirectly, by such "5-percent shareholders" at any time over the preceding three years.

In 2009, we experienced an ownership change under Section 382, which subjects the amount of pre-change NOLs and certain other pre-change tax attributes that can be utilized to an annual limitation, which will substantially limit the future use of our pre-change NOLs and certain other pre-change tax attributes per year. We may have recently experienced a second ownership change (or may be close to the threshold for a second ownership change), which would result in similar limitations on our NOLs and certain other tax attributes that arose since the 2009 ownership change.

Recently proposed legislation, if enacted, could subject us to U.S. federal income taxation as if we were a U.S. corporation.

A bill recently introduced in the House of Representatives provides in certain instances that a foreign corporation would, for any taxable year beginning on or after the second anniversary of the bill's enactment, be treated as a U.S. corporation for U.S. federal income taxes purposes if such corporation were managed and controlled primarily in the United States. If this bill were enacted in 2011 in its present form and we were to make no changes to our current management structure, we would likely be treated, beginning in 2014, as a U.S. corporation subject to U.S. federal income taxation on our worldwide income. There can be no assurance that the foregoing bill or another similar legislative proposal will not become law.

Because we are a relatively small biopharmaceutical company with limited resources, we may not be able to attract and retain qualified personnel.

Our success in developing marketable products and achieving a competitive position will depend, in part, on our ability to attract and retain qualified scientific and management personnel, particularly in areas requiring specific technical, scientific or medical expertise. We had approximately 245 employees as of August 2, 2011. We may require additional experienced executive, accounting, research and development, legal, administrative and other personnel from time to time in the future. There is intense competition for the services of these personnel, especially in California. Moreover, we expect that the high cost of living in the San Francisco Bay Area, where our headquarters and manufacturing facilities are located, may impair our ability to attract and retain employees in the future. If we do not succeed in attracting new personnel and retaining and motivating existing personnel, our operations may suffer and we may be unable to implement our current initiatives or grow effectively.

Calamities, power shortages or power interruptions at our Berkeley headquarters and manufacturing facility could disrupt our business and adversely affect our operations, and could disrupt the businesses of our customers.

Our principal operations are located in Northern California, including our corporate headquarters and manufacturing facility in Berkeley, California. In addition, many of our collaborators and licensees are located in California. All of these locations are in areas of seismic activity near active earthquake faults. Any earthquake, terrorist attack, fire, power shortage or other calamity affecting our facilities or our customers' facilities may disrupt our business and could have material adverse effect on our business and results of operations.

We may be subject to increased risks because we are a Bermuda company.

Alleged abuses by certain companies that have changed their legal domicile from jurisdictions within the United States to Bermuda have created an environment where, notwithstanding that we changed our legal domicile in a transaction that was approved by our shareholders and fully taxable to our company under United States law, we may be exposed to various prejudicial actions, including:

- · "blacklisting" of our common shares by certain pension funds,
- · legislation restricting certain types of transactions, and
- · punitive tax legislation.

We do not know whether any of these things will happen, but if implemented one or more of them may have an adverse impact on our future operations or our share price.

It may be difficult to enforce a judgment obtained against us because we are a foreign entity.

We are a Bermuda company. All or a substantial portion of our assets, including substantially all of our intellectual property, may be located outside the United States. As a result, it may be difficult for shareholders and others to enforce in United States courts judgments obtained against us. We have irrevocably agreed that we may be served with process with respect to actions based on offers and sales of securities made hereby in the United States by serving Christopher J. Margolin, c/o XOMA Ltd., 2910 Seventh Street, Berkeley, California 94710, our United States agent appointed for that purpose. Uncertainty exists as to whether Bermuda courts would enforce judgments of United States courts obtained in actions against us or our directors and officers that are predicated upon the civil liability provisions of the United States securities laws or entertain original actions brought in Bermuda against us or such persons predicated upon the United States securities laws. There is no treaty in effect between the United States and Bermuda providing for such enforcement, and there are grounds upon which Bermuda courts may not enforce judgments of United States courts. Certain remedies available under the United States federal securities laws may not be allowed in Bermuda courts as contrary to that nation's policy.

Our shareholder rights agreement or bye-laws may prevent transactions that could be beneficial to our shareholders and may insulate our management from removal.

In February of 2003, we adopted a new shareholder rights agreement (to replace the shareholder rights agreement that had expired), which could make it considerably more difficult or costly for a person or group to acquire control of us in a transaction that our Board of Directors opposes.

Our bye-laws:

- require certain procedures to be followed and time periods to be met for any shareholder to propose matters to be considered at annual meetings of shareholders, including nominating directors for election at those meetings;
- authorize our Board of Directors to issue up to 1,000,000 preference shares without shareholder approval and to set the rights, preferences and other designations, including voting rights, of those shares as the Board of Directors may determine; and
- contain provisions, similar to those contained in the Delaware General Corporation Law that may make business combinations with interested shareholders more difficult.

These provisions of our shareholders rights agreement and our bye-laws, alone or in combination with each other, may discourage transactions involving actual or potential changes of control, including transactions that otherwise could involve payment of a premium over prevailing market prices to holders of common shares, could limit the ability of shareholders to approve transactions that they may deem to be in their best interests, and could make it considerably more difficult for a potential acquirer to replace management.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

None.

ITEM 3. DEFAULTS UPON SENIOR SECURITIES

None.

ITEM 4. RESERVED

ITEM 5. OTHER INFORMATION

At the Company's 2011 annual general meeting of shareholders held on May 26, 2011, in a non-binding, advisory vote of the Company's shareholders regarding the desired frequency of a non-binding, advisory vote to approve the Company's named executive officer compensation, the proposed frequency that received the highest number of votes was once every three years. In light of this result and the other factors considered by the Board of Directors of the Company (the "Board") in making its original recommendation to the shareholders, the Board has determined that the Company will hold a non-binding, advisory vote on the Company's compensation of its named executive officers as disclosed in the proxy statement every three years until the Board determines that a different frequency for such advisory vote is in the Company's best interest or the next required vote on the frequency of such vote.

Table of Contents

ITEM 6. EXHIBITS

Exhibit Number

<u>10.1</u>	Foreign Exchange and Options Master Agreement (FEOMA) dated as of May 16, 2011 between Royal Bank of Canada and XOMA Ltd., with letter agreement dated May 17, 2011
	agreenent dated way 17, 2011
<u>31.1</u>	Certification of Steven B. Engle, filed pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
<u>31.2</u>	Certification of Fred Kurland, filed pursuant to Section 302 of the Sarbanes-Oxley Act of 2002
<u>32.1</u>	Certification of Steven B. Engle, furnished pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
32.2	Certification of Fred Kurland, furnished pursuant to Section 906 of the Sarbanes-Oxley Act of 2002
<u>99.1</u>	Press Release dated August 4, 2011, furnished herewith
101.INS	XBRL Instance Document
101 0011	
101.SCH	XBRL Schema Document
101.CAL	XBRL Calculation Linkbase Document
101.LAB	XBRL Label Linkbase Document
101.LAD	ABRE Later Linkoase Document
101.PRE	XBRL Presentation Linkbase Document
101.DEF	XBRL Definition Linkbase Document
TOTEDEL	

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

	XOMA Ltd.	
Date: August 4, 2011	By: /s/ STEVEN B. ENGLE Steven B. Engle Chairman, Chief Executive Officer and President (principal executive officer)	-
Date: August 4, 2011	By: /s/ FRED KURLAND Fred Kurland Vice President, Finance and Chief Financial Officer (principal financial officer and chief accounting officer)	-
	43	

THE FOREIGN EXCHANGE COMMITTEE

in association with

THE BRITISH BANKERS' ASSOCIATION

and

THE CANADIAN FOREIGN EXCHANGE COMMITTEE

and

THE TOKYO FOREIGN EXCHANGE MARKET PRACTICES COMMITTEE

THE 1997

INTERNATIONAL FOREIGN EXCHANGE AND OPTIONS MASTER AGREEMENT

(FEOMA)

8 The Foreign Exchange Committee 1997 All Rights Reserved.

International Foreign Exchange and Options Master Agreement (1997)

TABLE OF CONTENTS

1.	Definitions	1
2.	FX Transactions and Options	11
2.1	Scope of the Agreement	11
2.2	Single Agreement	12
2.3	Confirmations	12
2.4	Inconsistencies	12
3.	Option Premium	12
3.1	Payment of Premium	12
3.2	Late Payment or Non-Payment of Premium	12
5.2		15
4.	Discharge and Termination of Options; Netting of Option Premiums	13
4.1	Discharge and Termination	13
4.2	Netting of Option Premiums	14
5.	Exercise and Settlement of Options	14
5.1	Exercise of Options	14
5.2	No Partial Exercise	15
5.3	Automatic Exercise	15
5.4	Settlement of Exercised Options	15
5.5	Settlement at In-the-Money Amount	16
6.	Settlement and Netting of FX Transactions	16
6.1	Settlement of FX Transactions	16
6.2	Settlement Netting	16
6.3	Novation Netting	16
6.4	General	17
7.	Representations. Warranties and Covenants	17
7.1	Representations and Warranties	17
7.2	Covenants	18
0		10
8.	Close-out and Liquidation Manner of Close-Out and Liquidation	18
8.1		18
8.2.	Set-Off Against Credit Support	22 22
8.3	Other Foreign Exchange Transactions and Currency Options	22 23
8.4	Payment and Late Interest	
8.5	Suspension of Obligations	23
8.6	Expenses	23
8.7	Reasonable Pre-Estimate	24
8.8	No Limitation of Other Rights; Set-Off	24

0		24
9.	Force Majeure, Act of State, Illegality and Impossibility	24
9.1	Force Majeure, Act of State, Illegality and Impossibility	24
9.2	Transfer to Avoid Force Majeure, Act of State, Illegality or Impossibility	25
10.	Parties to Rely on Their Own Expertise	25
11.	Miscellaneous	26
11.1	Currency Indemnity	26
11.2	Assignment	26
11.3	Telephonic Recording	27
11.4	Notices	27
11.5	Termination	27
11.6	Severability	27
11.7	No Waiver	27
11.8	Master Agreement	27
11.9	Time of Essence, Etc	28
11.10	Headings	28
11.11	Payments Generally	28
11.12	Amendments	28
11.13	Credit Support	28
11.14	Adequate Assurances	28
11.15	Correction of Confirmations	29
12.	Law and Jurisdiction	29
12.1	Governing Law	29
12.2	Consent to Jurisdiction	29
12.3	Waiver of Jury Trial	29
	Waiver of Immunities	29
Schedule		i
Seneatic		

FOREIGN EXCHANGE AND OPTIONS MASTER AGREEMENT (FEOMA)

, 2011, by and between Royal Bank of MASTER AGREEMENT dated as of May 16 <u>a n a d a</u> ,a federally chartered canadian bank

and XOMA Ltd , a Bermuda company

SECTION 1. DEFINITIONS

Unless otherwise required by the context, the following terms shall have the following meanings in the Agreement:

"Agreement'1 has the meaning given to it in Section 2.2.

"American Style Option" means an Option which may be exercised on any Business Day up to and including the Expiration Time.

"Base Currency", as to a Party, means the Currency agreed to as such in relation to it in Part VII of the Schedule.

"Business Day" means for purposes of: (i) Section 3.2. a day which is a Local Banking Day for the applicable Designated Office of the Buyer; (ii) Section 5.1 and the definition of American Style Option, a day which is a Local Banking Day for the applicable Designated Office of the Seller; (iii) clauses (i), (viii) and (xii) of the definition of Event of Default, a day which is a Local Banking Day for the Non-Defaulting Party: (iv) solely in relation to delivery of a Currency, a day which is a Local Banking Day in relation to that Currency; and (v) any other provision of the Agreement, a day which is a Local Banking Day for the applicable Designated Offices of both Parties; provided, however, that neither Saturday nor Sunday shall be considered a Business Day for any purpose.

"Buyer" means the owner of an Option.

"Call" means an Option entitling, but not obligating (except upon exercise), the Buyer to purchase from the Seller at the Strike Price a specified quantity of the Call Currency.

"Call Currency" means the Currency agreed to as such at the time an Option is entered into, as evidenced in a Confirmation.

"Close-Out Amount" has the meaning given to it in Section 8.1.

"Close-Out Date" means a day on which, pursuant to the provisions of Section 8.1. the Non- Defaulting Party closes out Currency Obligations and/or Options or such close-out occurs automatically.

"Closing Gain", as to the Non-Defaulting Party, means the difference described as such in relation to a particular Value Dale under the provisions of Section 8.1.

"Closing Loss", as to the Non-Defaulting Party, means the difference described as such in relation to a particular Value Date under the provisions of Section 8.1.

"Confirmation" means a writing (including telex, facsimile or other electronic means from which it is possible to produce a hard copy) evidencing an FX Transaction or an Option, and specifying:

- (A) in the case of an FX Transaction, the following information:
 - (i) the Parties thereto and the Designated Offices through which they are respectively acting,
 - (ii) the amounts of the Currencies being bought or sold and by which Party,
 - (iii) the Value Date, and
 - (iv) any other term generally included in such a writing in accordance with the practice of the relevant foreign exchange market; and
- (B) in the case of an Option, the following information:
 - (i) the Parties thereto and the Designated Offices through which they are respectively acting,
 - (ii) whether the Option is a Call or a Put,
 - (iii) the Call Currency and the Put Currency that are the subject of the Option and their respective quantities,
 - (iv) which Party is the Seller and which is the Buyer,
 - (v) the Strike Price,

- (vi) the Premium and the Premium Payment Date,
- (vii) the Expiration Date,
- (viii) the Expiration Time,
- (ix) whether the Option is an American Style Option or a European Style Option, and
- (x) such other matters, if any, as the Parties may agree.

"Credit Support" has the meaning given to it in Section 8.2.

"Credit Support Document", as to a Party (the "first Party"), means a guaranty, hypothecation agreement, margin or security agreement or document, or any other document containing an obligation of a third party ("Credit Support Provider") or of the first Party in favor of the other Party supporting any obligations of the first Party under the Agreement.

"Credit Support Provider" has the meaning given to it in the definition of Credit Support Document.

"Currency" means money denominated in the lawful currency of any country or the Ecu.

"Currency Obligation" means any obligation of a Party to deliver a Currency pursuant to an FX Transaction, the application of Section 6.3(a) or (b). or an exercised Option (except, for the purposes of Section 8.1 only, one that is to be settled at its In-the-Money Amount under Section 5.5).

"Currency Pair" means the two Currencies which potentially may be exchanged in connection with an FX Transaction or upon the exercise of an Option, one of which shall be the Put Currency and the other the Call Currency.

"Custodian" has the meaning given to it in the definition of Insolvency Proceeding.

"Defaulting Party" has the meaning given to it in the definition of Event of Default.

"Designated Office(s)", as to a Party, means the office or offices specified in Part II of the Schedule.

"Effective Date" means the date of this Master Agreement.

"European Style Option" means an Option for which Notice of Exercise may be given only on the Option's Expiration Date up to and including the Expiration Time, unless otherwise agreed.

"Event of Default" means the occurrence of any of the following with respect to a Party (the "Defaulting Party", the other Party being the "Non-Defaulting Party"):

(i) the Defaulting Party shall (A) default in any payment when due under the Agreement (including, but not limited to, a Premium payment) to the Non-Defaulting Party with respect to any Currency Obligation or Option and such failure shall continue for two (2) Business Days after the Non-Defaulting Party has given the Defaulting Party written notice of non-payment, or (B) fail to perform or comply with any other obligation assumed by it under the Agreement and such failure is continuing thirty (30) days after the Non-Defaulting Party has given the Defaulting Party written notice thereof;

(ii) the Defaulting Party shall commence a voluntary Insolvency Proceeding or shall take any corporate action to authorize any such Insolvency Proceeding;

(iii) a governmental authority or self-regulatory organization having jurisdiction over either the Defaulting Party or its assets in the country of its organization or principal office (A) shall commence an Insolvency Proceeding with respect to the Defaulting Party or its assets or (B) shall take any action under any bankruptcy, insolvency or other similar law or any banking, insurance or similar law or regulation governing the operation of the Defaulting Part)- which may prevent the Defaulting Party from performing its obligations under the Agreement as and when due;

(iv) an involuntary Insolvency Proceeding shall be commenced with respect to the DefaultingParty or its assets by a person other than a governmental authority or self-regulatory organization having jurisdiction over either the Defaulting Party or its assets in the country of its organization or principal office and such Insolvency Proceeding (A) results in the appointment of a Custodian or a judgment of insolvency or bankruptcy or the entry of an order for winding-up, liquidation, reorganization or other similar relief, or (B) is not dismissed within five (5) days of its institution or presentation;

- (v) the Defaulting Party is bankrupt or insolvent, as defined under any bankruptcy or insolvency law applicable to it;
- (vi) the Defaulting Party fails, or shall otherwise be unable, to pay its debts as they becomedue;

(vii) the Defaulting Party or any Custodian acting on behalf of the Defaulting Party shall disaffirm, disclaim or repudiate any Currency Obligation or Option;

(viii) any representation or warranty made or given or deemed made or given by the Defaulting Party pursuant to the Agreement or any Credit Support Document shall prove to have been false or misleading in any material respect as at the time it was made or given or deemed made or given and one (1) Business Day has elapsed after the Non-Defaulting Party has given the Defaulting Party written notice thereof;

(ix) the Defaulting Party consolidates or amalgamates with or merges into or transfers all or substantially <u>all</u> its assets to another entity and (A) the creditworthiness of the resulting, surviving or transfere entity is materially weaker than that of the Defaulting Party prior to such action, or (B) at the time of such consolidation, amalgamation, merger or transfer the resulting, surviving or transferee entity fails to assume all the obligations of the Defaulting Party under the Agreement by operation of law or pursuant to an agreement satisfactory to the Non-Defaulting Party;

(x) by reason of any default, or event of default or other similar condition or event, anySpecified Indebtedness (being Specified Indebtedness of an amount which, when expressed in the Currency of the Threshold Amount, is in aggregate equal to or in excess of the Threshold Amount) of the Defaulting Party or any Credit Support Provider in relation to it: (A) is not paid on the due date therefor and remains unpaid after any applicable grace period has elapsed, or (B) becomes, or becomes capable at any time of being declared, due and payable under agreements or instruments evidencing such Specified Indebtedness before it would otherwise have been due and payable;

(xi) the Defaulting Part)' is in breach of or default under any Specified Transaction and any applicable grace period has elapsed, and there occurs any liquidation or early termination of, or acceleration of obligations under, that Specified Transaction or the Defaulting Party (or any Custodian on its behalf) disaffirms, disclaims or repudiates the whole or any part of a Specified Transaction;

(xii) (A) any Credit Support Provider of the Defaulting Party or the Defaulting Party itself fails to comply with or perform any agreement or obligation to be complied with or performed by it in accordance with the applicable Credit Support Document and such failure is continuing after any applicable grace period has elapsed; (B) any Credit Support Document relating to the Defaulting Party expires or ceases to be in full force and effect prior to the satisfaction of all obligations of the Defaulting Party under the Agreement, unless otherwise agreed in writing by the Non-Defaulting Party; (C) the Defaulting Party or any Credit Support Provider of the Defaulting Party (or, in either case, any Custodian acting on its behalf) disaffirms, disclaims or repudiates, in whole or in part, or challenges the validity of, any Credit Support Documents; (D) any representation or misleading in any material respect as at the time it was made or given or deemed made or given and one (1) Business Day has elapsed after the Non-Defaulting Party has given the Defaulting Party written notice thereof; or (E) any event set out in (ii) to (vii) or (ix) to (xi) above occurs in respect of any Credit Support Provider of the Defaulting Party; or

(xiii) any other condition or event specified in Part IX of the Schedule or in Section 11.14 ifmade applicable to the Agreement in Part XI of the Schedule.

"Exercise Date", in respect of any Option, means the day on which a Notice of Exercise received by the applicable Designated Office of the Seller becomes effective pursuant to Section 5.1.

"Expiration Date", in respect of any Option, means the date agreed to as such at the time the Option is entered into, as evidenced in a Confirmation.

"Expiration Time", in respect of any Option, means the latest time on the Expiration Date on which the Seller must accept a Notice of Exercise as agreed to at the time the Option is entered into, as evidenced in a Confirmation.

"FX Transaction" means any transaction between the Parties for the purchase by one Party of an agreed amount in one Currency against the sale by it to the other of an agreed amount in another Currency, both such amounts either being deliverable on the same Value Date or, if the Parties have so agreed in Part VI of the Schedule, being cashsettled in a single Currency, which is or shall become subject to the Agreement and in respect of which transaction the Parties have agreed (whether orally, electronically or in writing): the Currencies involved, the amounts of such Currencies to be purchased and sold, which Party will purchase which Currency and the Value Date.

"In-the-Money Amount" means (i) in the case of a Call, the excess of the Spot Price over the Strike Price, multiplied by the aggregate amount of the Call Currency to be purchased under the Call, where both prices are quoted in terms of the amount of the Put Currency to be paid for one unit of the Call Currency; and (ii) in the case of a Put, the excess of the Strike Price over the Spot Price, multiplied by the aggregate amount of the Put Currency to be sold under the Put, where both prices are quoted in terms of the amount of the Call Currency to be paid for one unit of the Put Currency.

"Insolvency Proceeding" means a case or proceeding seeking a judgment of or arrangement for insolvency, bankruptcy, composition, rehabilitation, reorganization, administration, winding-up, liquidation or other similar relief with respect to the Defaulting Party or its debts or assets, or seeking the appointment of a trustee, receiver, liquidator, conservator, administrator, custodian or other similar official (each, a "Custodian") of the Defaulting Party or any substantial part of its assets, under any bankruptcy, insolvency or other similar law or any banking, insurance or similar law governing the operation of the Defaulting Party.

"LIBOR", with respect to any Currency and date, means the average rate at which deposits in the Currency for the relevant amount and time period are offered by major banks in the London interbank market as of 11:00 a.m. (London time) on such date, or, if major banks do not offer deposits in such Currency in the London interbank market on such date, the average rate at which deposits in the Currency for the relevant amount and time period are offered by major banks in the relevant foreign exchange market at such time on such date as may be determined by the Party making the determination.

"Local Banking Day" means (i) for any Currency, a day on which commercial banks effect deliveries of that Currency in accordance with the market practice of the relevant foreign exchange market, and (ii) for any Party, a day in the location of the applicable Designated Office of such Part) on which commercial banks in that location are not authorized or required by law to close.

"Master Agreement" means the terms and conditions set forth in this Master Agreement, including the Schedule.

"Matched Pair Novation Netting Office(s)". in respect of a Party, means the Designated Office(s) specified in Part V of the Schedule.

"Non-Defaulting Party" has the meaning given to it in the definition of Event of Default.

"Notice of Exercise" means telex, telephonic or other electronic notification (excluding facsimile transmission) providing assurance of receipt, given by the Buyer prior to or at the Expiration Time, of the exercise of an Option, which notification shall be irrevocable.

"Novation Netting Office(s)", in respect of a Party, means the Designated Office(s) specified in Part V of the Schedule.

"Option" means a currency option which is or shall become subject to the Agreement.

"Parties" means the parties to the Agreement, including their successors and permitted assigns <u>(but without prejudice to the application of clause (ix) of the definition of</u> Event of Default): and the term "Party" shall mean whichever of the Parties is appropriate in the context in which such expression may be used.

"Premium", in respect of any Option, means the purchase price of the Option as agreed upon by the Parties, and payable by the Buyer to the Seller thereof.

"Premium Payment Date", in respect of any Option, means the date on which the Premium is due and pax able, as agreed to at the time the Option is entered into, as evidenced in a Confirmation.

"Proceedings" means any suit, action or other proceedings relating to the Agreement, any FX Transaction or any Option.

"Put" means an Option entitling, but not obligating (except upon exercise), the Buyer to sell to the Seller at the Strike Price a specified quantity of the Put Currency.

"Put Currency" means the Currency agreed to as such at the time an Option is entered into, as evidenced in a Confirmation.

"Schedule" means the Schedule attached to and part of this Master Agreement, as it may be amended from time to time by agreement of the Parties.

"Seller" means the Party granting an Option.

"Settlement Date" means, in respect of: (i) an American Style Option, the Spot Date of the Currency Pair on the Exercise Date of such Option, and (ii) a European Style Option, the Spot Date of the Currency Pair on the Expiration Date of such Option; and. where market practice in the relevant foreign exchange market in relation to the two Currencies involved provides for delivery of one Currency on one date which is a Local Banking Day in relation to that Currency but not to the other Currency and for delivery of the other Currency on the next Local Banking Day in relation to that other Currency, "Settlement Date" means such two (2) Local Banking Days.

"Settlement Netting Office(s)". in respect of a Part)', means the Designated Office(s) specified in Part V of the Schedule.

"Specified Indebtedness" means any obligation (whether present or future, contingent or otherwise, as principal or surety or otherwise) in respect of borrowed money, other than in respect of deposits received.

"Specified Transaction" means any transaction (including an agreement with respect thereto) between one Party to the Agreement (or any Credit Support Provider of such Party) and the other Party to the Agreement (or any Credit Support Provider of such Party) which is a rate swap transaction, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity linked swap, equity or equity index option, bond option, interest rate option, foreign exchange transaction, cap transaction, floor transaction, currency swap transaction, cross- currency rate swap transaction, currency option or any other similar transaction (including any option with respect to any of these transactions) or any combination of any of the foregoing.

"Spot Date" means the spot delivery day for the relevant Currency Pair as generally used by the relevant foreign exchange market.

"Spot Price" means the rate of exchange at the time at which such price is to be determined for foreign exchange transactions in the relevant Currency Pair for value on the Spot Date, as determined in good faith: (i) by the Seller, for purposes of Section 5, and (ii) by the Non-Defaulting Party, for purposes of Section 8.

"Strike Price", in respect of any Option, means the price at which the Currency Pair may be exchanged, as agreed to at the time the Option is entered into, as evidenced in a Confirmation.

"Threshold Amount" means the amount specified as such for each Party in Part VIII of the Schedule.

"Value Date" means, with respect to any FX Transaction, the Business Day (or where market practice in the relevant foreign exchange market in relation to the two Currencies involved provides for deliven of one Currency on one date which is a Local Banking Day in relation to that Currency but not to the other Currency and for delivery of the other Currency on the next Local Banking Day in relation to that other Currency ("Split Settlement") the two (2) Local Banking Days in accordance with that market practice) agreed by the Parties for delivery of the Currencies to be purchased and sold pursuant to such FX Transaction, and. with respect to any Currency Obligation, the Business Day (or, in the case of Split Settlement. Local Banking Day) upon which the obligation to deliver Currency pursuant to such Currency Obligation is to be performed.

SECTION 2. FX TRANSACTIONS AND OPTIONS

2.1 <u>Scope of the Agreement.</u> The Parties (through their respective Designated Offices) may enter into (i) FX Transactions, for such quantities of such Currencies, as may be agreed subject to the terms of the Agreement, and (ii) Options, for such Premiums, with such Expiration Dates, at such Strike Prices and for the purchase or sale of such quantities of such Currencies, as may be agreed subject to the terms of the Agreement; <u>provided</u> that neither Party shall be required to enter into any FX Transaction or Option with the other Party (other than in connection with an exercised Option). Unless otherwise agreed in writing by the Parties, each FX Transaction and Option entered into between Designated Offices of the Parties on or after the Effective Date shall be governed by the Agreement. Each FX Transaction and Option between any two Designated Offices of the Parties outstanding on the Effective Date which is identified in Part 1 of the Schedule shall also be governed by the Agreement.

2.2 <u>Single Agreement</u>. This Master Agreement, the terms agreed between the Parties with respect to each FX Transaction and Option (and, to the extent recorded in a Confirmation, each such Confirmation), and all amendments to any of such items shall together form the agreement between the Parties (the "Agreement") and shall together constitute a single agreement between the Parties. The Parties acknowledge that all FX Transactions and Options are entered into in reliance upon such fact, it being understood that the Parties would not otherwise enter into any FX Transaction or Option.

2.3 <u>Confirmations</u>. FX Transactions and Options shall be promptly confirmed by the Parties by Confirmations exchanged by mail, telex, facsimile or other electronic means from which it is possible to produce a hard copy. The failure by a Party to issue a Confirmation shall not prejudice or invalidate the terms of any FX Transaction or Option.

2.4 <u>Inconsistencies</u>. In the event of any inconsistency between the provisions of the Schedule and the other provisions of the Agreement, the Schedule will prevail. In the event of any inconsistency between the terms of a Confirmation and the other provisions of the Agreement, (i) in the case of an FX Transaction, the other provisions of the Agreement shall prevail, and the Confirmation shall not modify the other terms of the Agreement and (ii) in the case of an Option, the terms of the Confirmation shall prevail, and the other terms of the Agreement shall be deemed modified with respect to such Option, except for the manner of confirmation under Section 2.3 and, if applicable, discharge of Options under Section 4.

SECTION 3. <u>OPTION PREMIUM</u>

3.1 <u>Payment of Premium</u>. Unless otherwise agreed in writing by the Parties, the Buyer shall be obligated to pay the Premium related to an Option no later than its Premium Payment Date.

3.2 Late Payment or Non-Payment of Premium. If any Premium is not received on or before the Premium Payment Date, the Seller may elect: (i) to accept a late payment of such Premium; (ii) to give written notice of such non-payment and, if such payment shall not be received within two (2) Business Days of such notice, treat the related Option as void; or (iii) to give written notice of such non-payment and, if such payment shall not be received within two (2) Business Days of such notice, treat such non-payment as an Event of Default under clause (i) of the definition of Event of Default. If the Seller elects to act under either clause (i) or (ii) of the preceding sentence, the Buyer shall pay all out-of-pocket costs and actual damages incurred in connection with such unpaid or late Premium or void Option, including, without limitation, interest on such Premium from and including the Premium Payment Date to but excluding the late payment date in the same Currency as such Premium at overnight LIBOR and any other losses, costs or expenses incurred by the Seller in connection with such terminated Option, for the loss of its bargain, its cost of funding, or the loss incurred as a result of terminating, liquidating, obtaining or re-establishing a delta hedge or related trading position with respect to such Option.

SECTION 4. DISCHARGE AND TERMINATION OF OPTIONS: NETTING OF OPTION PREMIUMS

4.1 <u>Discharge and Termination.</u> If agreed in Part V of the Schedule, any Call or any Put written by a Party will automatically be discharged and terminated, in whole or in part, as applicable, against a Call or a Put, respectively, written by the other Party, such discharge and termination to occur automatically upon the payment in full of the last Premium payable in respect of such Options: provided that such discharge and termination may only occur in respect of Options:

- (i) each being with respect to the same Put Currency and the same Call Currency;
- (ii) each having the same Expiration Date and Expiration Time;
- (iii) each being of the same style, i.e. either both being American Style Options or both being European Style Options;
- (iv) each having the same Strike Price;

- (v) each being transacted by the same pair of Designated Offices of Buyer and Seller; and
- (vi) neither of which shall have been exercised by delivery of a Notice of Exercise:

and, upon the occurrence of such discharge and termination, neither Party shall have any further obligation to the other Party in respect of the relevant Options or, as the case may be, parts thereof so discharged and terminated. Such discharge and termination shall be effective notwithstanding that either Party may fail to record such discharge and termination in its books. In the case of a partial discharge and termination (i.e., where the relevant Options are for different amounts of the Currency Pair), the remaining portion of the Option which is partially discharged and terminated shall continue to be an Option for all purposes of the Agreement, including this Section 4.1.

4.2 <u>Netting of Option Premiums</u>. If agreed in Part V of the Schedule and if, on any date, Premiums would otherwise be payable under the Agreement in the same Currency between the same respective Designated Offices of the Parties, then, on such date, each Party's obligation to make payment of any such Premium will be automatically satisfied and discharged and, if the aggregate Premium(s) that would otherwise have been payable by such Designated Office of one Party exceeds the aggregate Premium(s) that would otherwise have been payable by an obligation upon the Party by whom the larger aggregate Premium(s) would have been payable to pay the other Party the excess of the larger aggregate Premium(s) over the smaller aggregate Premium(s) and, if the aggregate Premiums are equal, no payment shali be made.

SECTION 5. EXERCISE AND SETTLEMENT OF OPTIONS

5.1 <u>Exercise of Options</u>. The Buyer may exercise an Option by delivery to the Seller of a Notice of Exercise. Subject to Section 5.3, if a Notice of Exercise with respect to an Option has not been received by the Seller prior to or at the Expiration Time, the Option shall expire and become void and of no effect. Any Notice of Exercise shall (unless otherwise agreed):

(i) in respect of an American Style Option, (A) if received at or prior to 3:00 p.m. on a Business Day, be effective upon receipt thereof by the Seller, and (B) if received after 3:00 p.m. on a Business Day, be effective only as of the opening of business of the Seller on the first Business Day subsequent to its receipt; and

(ii) in respect of a European Style Option, if received on or, if the parties have so agreed, before the Expiration Date, prior to or at the Expiration Time, be effective upon receipt thereof by the Seller.

5.2 <u>No Partial Exercise</u>. Unless otherwise agreed by the Parties, an Option may be exercised only in whole.

5.3 Automatic Exercise. Unless otherwise agreed in Part VI of the Schedule or unless the Seller is otherwise instructed by the Buyer, if an Option has an In-the-Money Amount at its Expiration Time that equals or exceeds the product of (x) 1% of the Strike Price (or such other percentage or amount as may have been agreed by the Parties) and (y) the amount of the Call Currency or Put Currency, as appropriate, then the Option shall be deemed automatically exercised. In such case, the Seller may elect to settle such Option either in accordance with Section 5.4 or by payment to the Buyer on the Settlement Date for such Option of the In-the-Money Amount, as determined at the Expiration Time or as soon thereafter as practicable. In the latter case, the sole obligations of the Parties with respect to settlement of such Option shall be to deliver or receive the In-the-Money Amount of such Option on the Settlement Date. The Seller shall notify the Buyer of its election of the method of settlement of an automatically exercised Option as soon as practicable after the Expiration Time.

5.4 <u>Settlement of Exercised Options</u>. An exercised Option shall settle on its Settlement Date. Subject to Section 5.3 and 5.5, on the Settlement Date, the Buyer shall pay the Put Currency to the Seller for value on the Settlement Date and the Seller shall pay the Call Currency to the Buyer for value on the Settlement Date. An exercised Option shall be treated as an FX Transaction and a Currency Obligation (except, for the purposes of Section 8.1 only, if it is to be settled at its In-the-Money Amount), and for this purpose the relevant Settlement Date shall be treated as the Value Date of the FX Transaction.

5.5 <u>Settlement at In-the-Money Amount</u>. An Option shall be settled at its In-the-Money Amount if so agreed by the Parties at the time such Option is entered into. In such case, the In-the- Money Amount shall be determined based upon the Spot Price at the time of exercise or as soon thereafter as practicable. The sole obligations of the Parties with respect to settlement of such Option shall be to deliver or receive the In-the-Money Amount of such Option on the Settlement Date.

SECTION 6. SETTLEMENT AND NETTING OF FX TRANSACTIONS

6.1 <u>Settlement of FX Transactions</u>. Subject to Sections 6.2 and 6.3, each Party shall deliver to the other Party the amount of the Currency to be delivered by it under each Currency Obligation on the Value Date for such Currency Obligation.

6.2 <u>Settlement Netting</u>. If, on any date, more than one delivery of a particular Currency under Currency Obligations is to be made between a pair of Settlement Netting Offices, then each Party shall aggregate the amounts of such Currency deliverable by it and only the difference between these aggregate amounts shall be delivered by the Party owing the larger aggregate amount to the other Party, and, if the aggregate amounts are equal, no delivery of the Currency shall be made.

6 . 3 <u>Novation Netting</u>, (a) <u>By Currency</u>. If the Parties enter into an FX Transaction through a pair of Novation Netting Offices giving rise to a Currency Obligation for the same Value Date and in the same Currency as a then existing Currency Obligation between the same pair of Novation Netting Offices, then immediately upon entering into such FX Transaction, each such Currency Obligation shall automatically and without further action be individually canceled and simultaneously replaced by a new Currency Obligation for such Value Date determined as follows: the amounts of such Currency Obligation to deliver to the other Party on such Value Date shall be aggregate amount shall have a new Currency Obligation to deliver to the other Part}' the amount of such Currency by which its aggregate amount exceeds the other Party's aggregate amount, <u>provided</u> that if the aggregate amounts are equal, no new Currency Obligation shall arise. This Section 6.3 shall not affect any other Currency Obligation of a Party to deliver any different Currency on the same Value Date.

(b) <u>By Matched Pair</u>. If the Parties enter into an FX Transaction between a pair of Matched Pair Novation Netting Offices then the provisions of Section 6.3(a) shall apply only in respect of Currency Obligations arising by virtue of FX Transactions entered into between such pair of Matched Pair Novation Netting Offices and involving the same pair of Currencies and the same Value Date.

6.4 <u>General (a) Inapplicability of Sections 6.2 and 6.3</u>. The provisions of Sections 6.2 and 6.3 shall not apply if a Close-Oul Date has occurred or a voluntary or involuntary Insolvency Proceeding or action of the kind described in clause (ii), (iii) or (iv) of the definition of Event of Default has occurred without being dismissed in relation to either Party.

(b) Failure to Record. The provisions of Section 6.3 shall apply notwithstanding that either Party may fail to record the new Currency Obligation in its books.

(c) <u>Cut-off Date and Time</u>. The provisions of Section 6.3 are subject to any cut-off date and cut-off time agreed between the applicable Novation Netting Offices and Matched Pair Novation Netting Offices of the Parties.

SECTION 7. <u>REPRESEN TATION'S. WARRANTIES AND COVENANTS</u>

7.1 Representations and Warranties. Each Party represents and warrants to the other Party as of the Effective Date and as of the date of each FX Transaction and each Option that: (i) it has authority to enter into the Agreement (including such FX Transaction or Option, as the case may be); (ii) the persons entering into the Agreement (including such FX Transaction or Option, as the case may be) on its behalf have been duly authorized to do so; (iii) the Agreement (including such FX Transaction or Option, as the case may be) is binding upon it and enforceable against it in accordance with its terms (subject to applicable bankruptcy, reorganization, insolvency, moratorium or similar laws affecting creditors' rights generally and applicable principles of equity) and does not and will not violate the terms of any agreements to which such Party is bound; (iv) no Event of Default, or event which, with notice or lapse of time or both, would constitute an Event of Default, has occurred and is continuing with respect to it; (v) it acts as principal in entering into each FX Transaction and Option and exercising each and every Option; and (vi) if the Parties have so specified in Part XV of the Schedule, it makes the representations and warranties set forth in such Part XV.

7.2 <u>Covenants</u>. Each Party covenants to the other Party that: (i) it will at all times obtain and comply with the terms of and do all that is necessary to maintain in full force and effect all authorizations, approvals, licenses and consents required to enable it lawfully to perform its obligations under the Agreement; (ii) it will promptly notify the other Party of the occurrence of any Event of Default with respect to itself or any Credit Support Provider in relation to it; and (iii) if the Parties have set forth additional covenants in Part XVI of the Schedule, it makes the covenants set forth in such Part XVI.

SECTION 8. <u>CLOSE-OUT AND LIQUIDATION</u>

8.1 <u>Manner of Close-Out and Liquidation</u>, (a) <u>Close-Out</u>. If an Event of Default has occurred and is continuing, then the Non-Defaulting Party shall have the right to close out all. but not less than all, outstanding Currency Obligations (including any Currency Obligation which has not been performed and in respect of which the Value Date is on or precedes the Close-Out Date) and Options, except to the extent that in the good faith opinion of the Non-Defaulting Party certain of such Currency Obligations or Options may not be closed out under applicable law. Such close-out shall be effective upon receipt by the Defaulting Party of notice that the Non-Defaulting Party is terminating such Currency Obligations and Options. Notwithstanding the foregoing, unless otherwise agreed by the Parties in Part X of the Schedule, in the case of an Event of Default in clause (ii), (iii) or (iv) of the definition thereof with respect to a Party and, if agreed by the Parties in Part IX of the Schedule, in the case of any other Event of Default specified and so agreed in Part IX with respect to a Party, close-out shall be automatic as to all outstanding Currency Obligations and Options, as of the time immediately preceding the institution of the relevant Insolvency Proceeding or action. The Non-Defaulting Party shall have the right to liquidate such closed-out Currency Obligations and Options as provided below.

1	0
1	8

- (b) <u>Liquidation of Currency Obligations</u>. Liquidation of Currency Obligations terminated by close-out shall be effected as follows:
- (i) <u>Calculating Closing Gain or Loss</u>. The Non-Defaulting Party shall calculate in good faith, with respect to each such terminated Currency Obligation, except to the extent that in the good faith opinion of the Non-Defaulting Party certain of such Currency Obligations may not be liquidated as provided herein under applicable law, as of the Close-Out Date or as soon thereafter as reasonably practicable, the Closing Gain. or. as appropriate, the Closing Loss, as follows:
 - (A) for each Currency Obligation calculate a "Close-Out Amount" as follows:
 - (1) in the case of a Currency Obligation whose Value Date is the same as or is later than the Close-Out Date, the amount of such Currency Obligation; or
 - (2) in the case of a Currency Obligation whose Value Date precedes the Close-Out Date, the amount of such Currency Obligation increased, to the extent permitted by applicable law, by adding interest thereto from and including the Value Date to but excluding the Close-Out Date at overnight LIBOR; and
 - (3) for each such amount in a Currency other than the Non-Defaulting

Party=s Base Currency, convert such amount into the Non-Defaulting Party=s Base Currency at the rate of exchange at which, at the time of the calculation, the Non-Defaulting Party can buy such Base Currency with or against the Currency of the relevant Currency Obligation for delivery (x) if the Value Date of such Currency Obligation is on or after the Spot Date as of such time of calculation for the Base Currency, on the Value Date of that Currency Obligation or (y) if such Value Date precedes such Spot Date, for delivery on such Spot Date (or, in either case, if such rate of exchange is not available, conversion shall be accomplished by the Non-Defaulting Party using any commercially reasonable method); and

- (B) determine in relation to each Value Date: (1) the sum of all Close-Out Amounts relating to Currency Obligations under which the Non-Defaulting Party would otherwise have been entitled to receive the relevant amount on that Value Date; and (2) the sum of all Close-Out Amounts relating to Currency Obligations under which the Non-Defaulting Party would otherwise have been obliged to deliver the relevant amount to the Defaulting Party on that Value Date; and
- (C) if the sum determined under (B)(1) is greater than the sum determined under (B)(2), the difference shall be the Closing Gain for such Value Date; if the sum determined under (B)(1) is less than the sum determined under (B)(2), the difference shall be the Closing Loss for such Value Date.
- (ii) <u>Determining Present Value</u>. To the extent permitted by applicable law, the Non- Defaulting Party shall adjust the Closing Gain or Closing Loss for each Value Date falling after the Close-Out Date to present value by discounting the Closing Gain or Closing Loss from and including the Value Date to but excluding the Close-Out Date, at LIBOR with respect to the Non-Defaulting Party's Base Currency as at the Close-Out Date or at such other rate as may be prescribed by applicable law.



(iii) <u>Netting</u>. The Non-Defaulting Party shall aggregate the following amounts so that all such amounts are netted into a single liquidated amount payable to or by the Non- Defaulting Party: (x) the sum of the Closing Gains for all Value Dates (discounted to present value, where appropriate, in accordance with the provisions of Section 8.1(b)(ii)) (which for the purposes of the aggregation shall be a positive figure); and (y) the sum of the Closing Losses for all Value Dates (discounted to present value, where appropriate, in accordance with the provisions of Section 8.1(b)(ii)) (which for the purposes of the aggregation shall be a negative figure).

(c) Liquidation of Options. To liquidate unexercised Options and exercised Options to be settled at their In-the-Money Amounts that have been tenninated by close-out, the Non-Defaulting Party shall:

- (i) <u>Calculating Settlement Amount</u>. Calculate in good faith with respect to each such terminated Option, except to the extent that in the good faith opinion of the Non- Defaulting Party certain of such Options may not be liquidated as provided herein under applicable law, as of the Close-Out Date or as soon as reasonably practicable thereafter a settlement amount for each Party equal to the aggregate of:
 - (A) with respect to each Option purchased by such Party, and which the other Party has not elected to treat as void pursuant to Section 3.2(ii) for lack of payment of the Premium, the current market premium for such Option;
 - (B) with respect to each Option sold by such Party and which such Party has not elected to treat as void pursuant to Section 3.2(ii) for lack of payment of the Premium, any unpaid Premium, provided that, if the Close-Out Date occurs before the Premium Payment Date, such amount shall be discounted from and including the Premium Payment Date to but excluding the Close-Out Date at a rate equal to LIBOR on the Close-Out Date and, if the Close-Out Date occurs after the Premium Payment Date, to the extent permitted by applicable law, the settlement amount shall include interest on any unpaid Premium from and including the Premium Payment Date to but excluding the Close-Out Date in the same Currency as such Premium at overnight LIBOR;

- (C) with respect to any exercised Option to be settled at its In-the-Money Amount (whether or not the Close-Out Date occurs before the Settlement Date for such Option), any unpaid amount due to such Party in settlement of such Option and, if the Close-Out Date occurs after the Settlement Date for such Option, to the extent permitted by applicable law, interest thereon from and including the applicable Settlement Date to but excluding the Close-Out Date at overnight LIBOR; and
- (D) without duplication, the amount that the Non-Defaulting Party reasonably determines in good faith, as of the Close-Out Date or as of the earliest date thereafter that is reasonably practicable, to be its additional losses, costs and expenses in connection with such terminated Option, for the loss of its bargain, its cost of funding, or the loss incurred as a result of terminating, liquidating, obtaining or re-establishing a delta hedge or related trading position with respect to such Option;
- (ii) <u>Converting to Base Currency</u>. Convert any settlement amount calculated in accordance with clause (i) above in a Currency other than the Non-Defaulting Party's Base Currency into such Base Currency at the Spot Price at which, at the time of the calculation, the Non-Defaulting Party could enter into a contract in the foreign exchange market to buy the Non-Defaulting Party's Base Currency in exchange for such Currency (or, if such Spot Price is not available, conversion shall be accomplished by the Non-Defaulting Party using any commercially reasonable method); and



(iii) <u>Netting</u>. Net such settlement amounts with respect to each Party so that all such amounts are netted to a single liquidated amount payable by one Party to the other Party.

(d) <u>Final Netting</u>. The Non-Defaulting Party shall net (or, if both are payable by one Party, add) the liquidated amounts payable under Sections 8.1(b) and 8.1(c) with respect to each Party' so that such amounts are netted (or added) to a single liquidated amount payable by one Party to the other Party as a settlement payment.

8.2 <u>Set-Off Against Credit Support</u> Where close-out and liquidation occurs in accordance with Section 8.1, the Non-Defaulting Party' shall also be entitled (i) to set off the net payment calculated in accordance with Section 8.1(d) which the Non-Defaulting Party owes to the Defaulting Party, if any, against any credit support or other collateral ("Credit Support") held by the Defaulting Party pursuant to a Credit Support Document or otherwise (including the liquidated value of any non-cash Credit Support) in respect of the Non-Defaulting Party's obligations under the Agreement or (ii) to set off the net payment calculated in accordance with Section 8.1(d) which the Defaulting Party' owes to the Non-Defaulting Party, if any, against any Credit Support held by the Non-Defaulting Party' (including the liquidated value of any non-cash Credit Support) in respect of the Defaulting Party's obligations under the Agreement; provided that, for purposes of either such setoff, any Credit Support denominated in a Currency other than the Non-Defaulting Party's Base Currency shall be converted into such Base Currency at the rate specified in Section 8.1(c)(ii).

8.3 <u>Other Foreign Exchange Transactions and Currency Options</u> Where close-out and liquidation occurs in accordance with Section 8.1, the Non-Defaulting Party shall also be entitled to close-out and liquidate, to the extent permitted by applicable law, any other foreign exchange transaction or currency option entered into between the Parties which is then outstanding in accordance with the provisions of Section 8.1, with each obligation of a Party to deliver a Currency under such a foreign exchange transaction being treated as if it were a Currency Obligation (including exercised options, provided that cash-settled options shall be treated analogously to Options to be settled at their In-the-Money Amount) and each unexercised option being treated as if it were an

Option under the Agreement.

8.4 <u>Payment and Late Interest</u>. The net amount payable by one Party to the other Party pursuant to the provisions of Sections 8.1 and 8.3 above shall be paid by the close of business on the Business Day following the receipt by the Defaulting Party of notice of the Non-Defaulting Party's settlement calculation, with interest at overnight LIBOR from and including the Close-Out Date to but excluding such Business Day (and converted as required by applicable law into any other Currency, any costs of conversion to be borne by, and deducted from any payment to, the Defaulting Party). To the extent permitted by applicable law. any amounts owed but not paid when due under this Section 8 shall bear interest at overnight LIBOR (or, if conversion is required by applicable law into some other Currency, either overnight LIBOR with respect to such other Currency or such other rate as may be prescribed by such applicable law) for each day for which such amount remains unpaid. Any addition of interest or discounting required under this Section 8 shall be calculated on the basis of a year of such number of days as is customary for transactions involving the relevant Currency in the relevant foreign exchange market.

8.5 <u>Suspension of Obligations</u>. Without prejudice to the foregoing, so long as a Party shall be in default in payment or performance to the other Party under the Agreement and the other Party has not exercised its rights under this Section 8, or, if "Adequate Assurances" is specified as applying to the Agreement in Part XI of the Schedule, during the pendency of a reasonable request to a Party for adequate assurances of its ability to perform its obligations under the Agreement, the other Party may, at its election and without penalty, suspend its obligation to perform under the Agreement.

8.6 <u>Expenses</u>. The Defaulting Party shall reimburse the Non-Defaulting Party in respect of all out-of-pocket expenses incurred by the Non-Defaulting Party (including fees and disbursements of counsel, including attorneys who may be employees of the Non-Defaulting Party) in connection with any reasonable collection or other enforcement proceedings related to the payments required under the Agreement.

8.7 <u>Reasonable Pre-Estimate</u>. The Parties agree that the amounts recoverable under this Section 8 are a reasonable pre-estimate of loss and not a penalty. Such amounts are payable for the loss of bargain and the loss of protection against future risks and, except as otherwise provided in the Agreement, neither Party will be entitled to recover any additional damages as a consequence of such losses.

8.8 <u>No Limitation of Other Rights; Set-Off</u>. The Non-Defaulting Party's rights under this Section 8 shall be in addition to, and not in limitation or exclusion of, any other rights which the Non- Defaulting Party may have (whether by agreement, operation of law or otherwise), and, to the extent not prohibited by law, the Non-Defaulting Party shall have a general right of set-off with respect to all amounts owed by each Party to the other Party, whether due and payable or not due and payable (prov ided that any amount not due and payable at the time of such set-off shall, if appropriate, be discounted to present value in a commercially reasonable manner by the Non-Defaulting Party). The Non-Defaulting Party's rights under this Section 8.8 are subject to Section 8.7.

SECTION 9. FORCE MAJEURE. ACT OF STATE. ILLEGALITY AND IMPOSSIBILITY

9.1 Force Majeure. Act of State. Illegality and Impossibility. If either Party is prevented from or hindered or delayed by reason of force majeure or act of state in the delivery or receipt of any Currency in respect of a Currency Obligation or Option or if it becomes or, in the good faith judgment of one of the Parties, may become unlawful or impossible for either Party to make or receive any payment in respect of a Currency Obligation or Option, then the Party for whom such performance has been prevented, hindered or delayed or has become illegal or impossible shall promptly give notice thereof to the other Party and either Party may, by notice to the other Party, require the close-out and liquidation of each affected Currency Obligation and Option in accordance with the provisions of Section 8.1 and, for such purposes, the Parts' unaffected by such force majeure, act of state, illegality or impossibility (or, if both Parties are so affected, whichever Party gave the relevant notice) shall perform the calculation required under Section 8.1 as if it were the Non- Defaulting Party. Nothing in this Section 9.1 shall be taken as indicating that the Party treated as the

Defaulting Party for the purpose of calculations required by Section 8.1 has committed any breach or default.

9.2 Transfer to Avoid Force Majeure. Act of State. Illegality or Impossibility. If Section 9.1 becomes applicable, unless prohibited by law, the Party which has been prevented, hindered or delayed from performing shall, as a condition to its right to designate a close-out and liquidation of any affected Currency Obligation or Option, use all reasonable efforts (which will not require such Part}' to incur a loss, excluding immaterial, incidental expenses) to transfer as soon as practicable, and in any event before the earlier to occur of the expiration date of the affected Options or twenty (20) days after it gives notice under Section 9.1, all its rights and obligations under the Agreement in respect of the affected Currency Obligations to another of its Designated Offices so that such force ma jeure, act of state, illegality or impossibility ceases to exist. Any such transfer will be subject to the prior written consent of the other Party, which consent will not be withheld if such other Party's policies in effect at such time would permit it to enter into transactions with the transferee Designated Office on the tenns proposed, unless such transfer would cause the other Party to incur a material tax or other cost.

SECTION 10. PAR TIES TO RELY ON THEIR OWN EXPERTISE

Each Party will be deemed to represent to the other Party on the date on which it enters into an FX Transaction or Option that (absent a written agreement between the Parties that expressly imposes affirmative obligations to the contrary for that FX Transaction or Option): (i)(A) it is acting for its own account, and it has made its own independent decisions to enter into that FX Transaction or Option and as to whether that FX Transaction or Option is appropriate or proper for it based upon its own judgment and upon advice from such advisors as it has deemed necessary; (B) it is not relying on any communication (written or oral) of the other Party as investment advice or as a recommendation to enter into that FX Transaction or Option, it being understood that information and explanations related to the terms and conditions of an FX Transaction or Option shall not be considered investment advice or a recommendation to enter into that FX Transaction or Option; (ii) it is capable of evaluating and understanding (on its own behalf or through independent professional advice), and understands and accepts, the terms, conditions and risks of that FX Transaction or Option; and (iii) the other Party is not acting as a fiduciary or an advisor for it in respect of that FX Transaction or Option.

SECTION 11. MISCELLANEOUS

11.1 <u>Currency Indemnity</u>. The receipt or recover)' by either Party (the "first Party") of an> amount in respect of an obligation of the other Party (the "second Party") in a Currency other than that in which such amount was due, whether pursuant to a judgment of any court or pursuant to Section 8 or 9, shall discharge such obligation only to the extent that, on the first day on which the first Party is open for business immediately following such receipt or recovery, the first Party shall be able, in accordance with normal banking practice, to purchase the Currency in which such amount was due with the Currency received or recovered. If the amount so purchasable shall be less than the original amount of the Currency in which such amount was due, the second Party shall, as a separate obligation and notwithstanding any judgment of any court, indemnify the first Party against any loss sustained by it. The second Party shall in any event indemnify the first Party against any costs incurred by it in making any such purchase of Currency.

11.2 <u>Assignment</u>. Neither Party may assign, transfer or charge or purport to assign, transfer or charge its rights or obligations under the Agreement to a third party without the prior written consent of the other Party and any purported assignment, transfer or charge in violation of this Section 11.2 shall be void.

11.3 <u>Telephonic Recording</u>. The Parties agree that each may electronically record all telephonic conversations between them and that any such recordings may be submitted in evidence to any court or in any Proceedings for the purpose of establishing any matters pertinent to the Agreement.

1 1 . 4 <u>Notices.</u> Unless otherwise agreed, all notices, instructions and other communications to be given to a Part)' under the Agreement shall be given to the address, telex (if confirmed by the appropriate answerback), facsimile (confirmed if requested) or telephone number and to the individual or department specified by such Party in Part III of the Schedule. Unless otherwise specified, any notice, instruction or other communication given in accordance with this Section 11.4 shall be effective upon receipt.

11.5 <u>Termination</u>. Each of the Parties may terminate the Agreement at any time by seven (7) days' prior written notice to the other Party delivered as prescribed in Section 11.4, and termination shall be effective at the end of such seventh day; provided, however, that any such termination shall not affect any outstanding Currency Obligations or Options, and the provisions of the Agreement shall continue to apply until all the obligations of each Party to the other under the Agreement have been fully performed.

11.6 <u>Severability</u>. In the event any one or more of the provisions contained in the Agreement should be held invalid, illegal or unenforceable in any respect under the law of any jurisdiction, the validity, legality and enforceability of the remaining provisions contained in the Agreement under the law of such jurisdiction, and the validity, legality and enforceability of such and any other provisions under the law of any other jurisdiction shall not in any way be affected or impaired thereby. The Parties shall endeavor in good faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions the economic effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.

11.7 <u>No Waiver</u>. No indulgence or concession granted by a Patty' and no omission or delay on the part of a Party in exercising any right, power or privilege under the Agreement shall operate as a waiver thereof, nor shall any single or partial exercise of any such right, power or privilege preclude any other or further exercise thereof or the exercise of any other right, power or privilege.

11.8 <u>Master Agreement</u>. Where one of the Parties to the Agreement is domiciled in the United States, the Parties intend that the Agreement shall be a master agreement, as referred to in 11 U.S.C. Section 101(53B)(C) and 12 U.S.C. Section 1821(e)(8)(D)(vii).

11.9 <u>Time of Essence. Etc.</u> Time shall be of the essence in the Agreement. Unless otherwise agreed, the times referred to in the Agreement with respect to Options shall in each case refer to the local time of the relevant Designated Office of the Seller of the relevant Option.

11.10 <u>Headings</u>. Headings in the Agreement are for ease of reference only.

11.11 <u>Payments Generally</u>. All payments to be made under the Agreement shall be made in same day (or immediately available) and freely transferable funds and, unless otherwise specified, shall be delivered to such office of such bank, and in favor of such account as shall be specified by the Part) entitled to receive such payment in Part IV of the Schedule or in a notice given in accordance with Section 11.4.

11.12 <u>Amendments</u>. No amendment, modification or waiver of the Agreement will be effective unless in writing executed by each of the Parties<u>prov ided</u> that the Parties may agree in a Confirmation that complies with Section 2.3 to amend the Agreement solely with respect to the Option that is the subject of the Confirmation.

11.13 <u>Credit Support</u> A Credit Support Document between the Parties may apply to obligations governed by the Agreement. If the Parties have executed a Credit Support Document, such Credit Support Document shall be subject to the terms of the Agreement and is hereby incorporated by reference in the Agreement. In the event of any conflict between a Credit Support Document and the Agreement, the Agreement shall prevail, except for any provision in such Credit Support Document in respect of governing law.

11.14 <u>Adequate Assurances</u>. If the Parties have so agreed in Part XI of the Schedule, the failure by a Party to give adequate assurances of its ability to perform any of its obligations under the Agreement within two (2) Business Days of a written request to do so when the other Party has reasonable grounds for insecurity shall be an Event of Default under the Agreement.

1 1 . 1 5 <u>Correction of Confirmations</u>. Unless either Party objects to the terms contained in any Confirmation sent by the other Party or sends a corrected Confirmation within three (3) Business Days of receipt of such Confirmation, or such shorter time as may be appropriate given the Value Date of an FX Transaction, the terms of such Confirmation shall be deemed correct and accepted absent manifest error. If the Party receiving a Confirmation sends a corrected Confirmation within such three (3) Business Days, or shorter period, as appropriate, then the Party receiving such corrected Confirmation shall have three (3) Business Days, or shorter period, as appropriate, after receipt thereof to object to the terms contained in such corrected Confirmation.

SECTION 12. LAW AND JURISDICTION

12.1 <u>Governing Law</u>. The Agreement shall be governed by, and construed in accordance with, the laws of the jurisdiction set forth in Part XII of the Schedule without giving effect to conflict of laws principles.

1 2 . 2 <u>Consent to Jurisdiction</u>, (a) With respect to any Proceedings, each Party irrevocably (i) submits to<u>the</u> non-exclusive jurisdiction of the courts of the jurisdiction set forth in Part XIII of the Schedule and (ii) waives any objection which it may have at any time to the laying of venue of any Proceedings brought in any such court, waives any claim that such Proceedings have been brought in an inconvenient forum and further waives the right to object, with respect to such Proceedings, that such court does not have jurisdiction over such Party. Nothing in the Agreement precludes either Party from bringing Proceedings in any other jurisdiction nor will the bringing of Proceedings in any one or more jurisdictions preclude the bringing of Proceedings in any other jurisdiction.

(b) Each Party irrevocably appoints the agent for service of process (if any) specified with respect to it in Part XIV of the Schedule. If for any reason any Party's process agent is unable to act as such, such Party will promptly notify the other Party and within thirty (30) days will appoint a substitute process agent acceptable to the other Party.

12.3 Waiver of Jury Trial. Each Party irrevocably waives any and all right to trial by jury in any Proceedings.

12.4 <u>Waiver of Immunities</u>. Each Party irrevocably waives, to the fullest extent permitted by applicable law, with respect to itself and its revenues and assets (irrespective of their use or intended use), all immunity on the grounds of sovereignty or other similar grounds from (i) suit, (ii) jurisdiction of any court, (iii) relief by way of injunction, order for specific performance or for recovery of property, (iv) attachment of its assets (whether before or after judgment) and (v) execution or enforcement of any judgment to which it or its revenues or assets might otherwise be entitled in any Proceedings in the courts of any jurisdiction and irrevocably agrees, to the extent permitted by applicable law, that it will not claim any such immunity in any Proceedings.

By /s/ IRENE KLAUSMANN Name: IRENE KLAUSMANN Title: AUTHORIZED SIGNATORY

XOMA Ltd.

By /s/ Christopher J. Margolin

Name: Christopher J. Margolin

Title: Vice President, General Counsel and Secretary



17 May 2011

XOMA LTD. 2910 SEVENTH STREET BERKELEY CALIFORNIA, UNITED STATES 94710-2700

Attention:

Dear Sirs:

1. The purpose of this letter agreement (this "Confirmation") is to confirm the terms and conditions of the transaction entered into between us on the Trade Date specified below (the "Transaction"). This Confirmation constitutes a "Confirmation" as referred to in the Agreement specified below.

This Confirmation evidences a complete binding agreement between you and us entered into in the telephone trade on the Trade Date and the terms of the Transaction to which this Confirmation relates. This Confirmation shall supplement, form a part of, and be subject to an agreement (the "Master Agreement") in the form of the 2002 ISDA Master Agreement published by the International Swaps and Derivatives Association, Inc. as if we had executed an agreement in such form (but without any Schedule, except for the election of the laws of the State of New York as the governing law and USD as the Termination Currency) on the Trade Date of the first such Transaction between us. In the event of any inconsistency between the provisions of the Master Agreement and this Confirmation, this Confirmation will prevail for the purpose of this Transaction.

The definitions and provisions contained in the 1996 FX and Currency Option Definitions (as published by the International Swaps and Derivatives Association, Inc., the Emerging Markets Traders Association and The Foreign Exchange Committee) are incorporated into this Confirmation. In the event of any inconsistency between those definitions and the provisions of this Confirmation, this Confirmation shall govern.

2. The terms of the particular Transaction to which this Confirmation relates are as follows:

Contract No.:	647781
Transaction Date:	May 17, 2011
QpenIng'Closing Trade:	OPENING
Buyer:	XOMA LTD.
Seller:	ROYAL BANK OF CANADA
Style:	EUROPEAN
Type of Option:	CALL
Call Currency and amount:	15,000,000.00EUR
Put Currency and amount:	22,650,000.00USD
Strike Price:	1.510Q
Expiration Date:	January 11, 2016
Expiration Tims:	10:00AM New York Time
Settlement Date:	January 13, 2016
Broker:	DIRECT
Option Buyers Premium Payment:	1,370,000.00USD
Premium Payment Date:	May 19, 2011

3.	Account Details:			
	Payments to Buyer: Payments to Seller:			
4.	Offices			
	(a) (b)	The office of the Buyer for the currency Option Transaction Is: The office of the Seller for the Currency Option Transaction is:		BERKELEY TORONTO
mess		executed and delivered in counterparts (including by facsimile transm messaging system, which in each case upon your confirmation in the rransaction.		
to us	within two (2) busine	regoing correctly sets forth the terms of the Option by executing this C ess days a letter by facsimile transmission, telex or electronic messagin and which indicates your agreement to these terms and conditions.		
1	hone: mile No.:	416-642-4582 416-642-4246		
ROY	AL BANK OF CAN	ADA	Confirmed as of the date first written: XOMA LTD.	
Erica	ica Lei Lei gn Exchange Derivat	ive Operations	/s/ Tom Burns Name: Tom Burns Title: Director, FP&A	
			Name: Title:	

Certification Pursuant To Section 302 Of The Sarbanes-Oxley Act Of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) And (B))

I, Steven B. Engle, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of XOMA Ltd.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light
 of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f))) for the registrant and we have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
 material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the
 period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 4, 2011

/ STEVEN B. ENGLE

Steven B. Engle Chairman, Chief Executive Officer and President (principal executive officer)

Certification Pursuant To Section 302 Of The Sarbanes-Oxley Act Of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) And (B))

I, Fred Kurland, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of XOMA Ltd.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light
 of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f))) for the registrant and we have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that
 material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the
 period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 4, 2011

/ FRED KURLAND

Fred Kurland Vice President, Finance and Chief Financial Officer (principal financial officer)

Certification Pursuant To Section 906 Of The Sarbanes-Oxley Act Of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) And (B))

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(a) and (b)), the undersigned hereby certifies in his capacity as an officer of XOMA Ltd. (the "Company") that the quarterly report of the Company on Form 10-Q for the period ended June 30, 2011, fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934, as amended, and that the information contained in such report fairly presents, in all material respects, the financial condition and results of operations of the Company at the end of and for the periods covered by such report.

Date: August 4, 2011

/s/ STEVEN B. ENGLE

Steven B. Engle Chairman, Chief Executive Officer and President (principal executive officer)

This certification will not be deemed filed for purposes of Section 18 of the Exchange Act (15 U.S.C. 78), or otherwise subject to the liability of that section. Such certification will not be deemed to be incorporated by reference into any filing under the Securities Act or the Exchange Act, except to the extent that the registrant specifically incorporates it by reference.

Certification Pursuant To Section 906 Of The Sarbanes-Oxley Act Of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(A) And (B))

Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Chapter 63, Title 18 U.S.C. Section 1350(a) and (b)), the undersigned hereby certifies in his capacity as an officer of XOMA Ltd. (the "Company") that the quarterly report of the Company on Form 10-Q for the period ended June 30, 2011, fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934, as amended, and that the information contained in such report fairly presents, in all material respects, the financial condition and results of operations of the Company at the end of and for the periods covered by such report.

Date: August 4, 2011

/s/ FRED KURLAND

Fred Kurland Vice President, Finance and Chief Financial Officer (principal financial officer)

This certification will not be deemed filed for purposes of Section 18 of the Exchange Act (15 U.S.C. 78), or otherwise subject to the liability of that section. Such certification will not be deemed to be incorporated by reference into any filing under the Securities Act or the Exchange Act, except to the extent that the registrant specifically incorporates it by reference.



XOMA Reports Second Quarter 2011 Financial Results

BERKELEY, Calif., August 4, 2011 -- XOMA Ltd. (Nasdaq: XOMA), a leader in the discovery and development of therapeutic antibodies, today announced its financial results for the second quarter and six months ended June 30, 2011 and provided a general business update.

XOMA had total revenues of \$16.5 million in the second quarter of 2011 compared with \$5.9 million in the second quarter of 2010. The increase in revenues in the 2011 period compared with the 2010 period was due primarily to funding from the company's collaborative partner Les Laboratoires Servier (Servier) for XOMA 052 development and increased funding under the company's contracts with the U.S. government for XOMA 3AB development.

XOMA had a net loss of \$8.1 million, or \$0.27 per share, for the second quarter of 2011, compared with a net loss of \$15.6 million, or \$0.93 per share, for the second quarter of 2010. Research and development expenses in the second quarter of 2011 were \$18.3 million as compared with \$19.3 million in the second quarter of 2010. The reduction in R&D expenses in the 2011 second quarter compared with the 2010 second quarter was primarily due to lower expenses for the XOMA 052 Phase 2 clinical program. Selling, general and administrative expenses were \$6.1 million in the second quarter of 2011 as compared with \$5.0 million in the second quarter of 2010, primarily reflecting higher non-cash share-based compensation expense in the 2011 period.

At June 30, 2011, XOMA had cash and cash equivalents of \$51.2 million, compared with \$37.3 million at December 31, 2010. Since April 1, 2011, XOMA received gross proceeds of \$5.9 million from the sale of common shares under its 2011 At Market Issuance Sales Agreement, of which \$3.7 million in cash was received in the second quarter with the remaining \$2.2 million in cash received in the third quarter of 2011.

"Our financial results for the second quarter of 2011 as compared with the same period last year reflect the favorable impact of funding received from our XOMA 052 development collaboration with Servier and increased revenue from our U.S. government biodefense contracts," said Steven B. Engle, Chairman and Chief Executive Officer of XOMA. "In the second half of this year, we expect to annoucne the details of our XOMA 052 Phase 3 program in Behcet's uveitis which we plan to initiate this year with Servier. Servier continues to make progress toward initiation of a Phase 2 trial for XOMA 052 in cardiovascular disease."

Recent Highlights

XOMA 3AB Phase 1 trial initiated by NIAID: The National Institute of Allergy and Infectious Diseases, part of the National Institutes of Health, initiated a Phase 1 trial of XOMA 3AB, a novel formulation of three antibodies designed to prevent and treat botulism poisoning, among the most deadly bioterror threats. This doubleblind, dose-escalation study in approximately 24 healthy volunteers, is designed to assess the safety and tolerability and determine the pharmacokinetic profile of XOMA 3AB. More information is available at http://www.clinicaltrials.gov/ct2/show/NCT01357213?term=XOMA+3AB&rank=1

- Discovery of two new classes of insulin receptor-regulating antibodies reported at American Diabetes Association 7¹⁴ Scientific Sessions: Insulin is the key metabolic hormone for regulating blood sugar and exerts its action on cells by signaling through the insulin receptor. Insulin receptor-activating antibodies such as XOMA's XMetA antibody are designed to provide long-acting insulin-like activity to diabetic patients who cannot make sufficient insulin, potentially reducing the number of insulin injections needed to control their blood glucose levels. In contrast, insulin receptor-sensitizing antibodies such as XOMA's XMetS are designed to reduce insulin resistance and could enable diabetes patients to more effectively use their own insulin to control blood glucose levels.
- New U.S. patent issued covering XOMA 052 use in certain interleukin-1 beta-related coronary conditions: On August 2, the U.S. Patent and Trademark Office issued XOMA a new patent covering methods of treating certain coronary conditions including myocardial infarction, or heart attack, using XOMA 052 and interleukin-1 beta antibodies with similar binding properties. This is the tenth U.S. patent from the XOMA 052 program to have been issued, in addition to numerous pending applications and granted patents outside of the United States.
- Six month top line results from 74 patient XOMA 052 Phase 2a trial support safety and biological activity: XOMA 052 was well tolerated with no significant differences between the XOMA 052 and placebo groups in observations of adverse events. XOMA 052 continued to show evidence of biological activity as shown by a reduction in levels of C-reactive protein, a biomarker of cardiovascular risk. There were no differences in glycemic control between the drug groups and placebo as measured by hemoglobin A1c levels. This Phase 2a trial was designed as an exploratory trial focused on overall safety and kinetics and was not designed to show statistically significant differences in measures of biological activity. The results were as expected based on data from the Phase 2a three-month interim review and the 421 patient Phase 2b trial.

Guidance

XOMA will not be providing specific guidance on overall revenues or cash receipts for 2011 so as to best manage its ongoing business development discussions and other activities. The company currently expects that cash used in operating activities in 2011 may range from \$30 million to cash neutral.

Investor Conference Call and Webcast

XOMA will host a conference call and webcast today, August 4, 2011, at 4:30 p.m. ET. The webcast can be accessed via the Investors section of XOMA's website at http://investors.xoma.com/events.cfm and will be available for replay until close of business on November 4, 2011. Telephone numbers for the live audio cast are 877-369-6589 (U.S./Canada) and 408-337-0122 (international). A telephonic replay will be available beginning approximately two hours after the conclusion of the call until close of business on August 11, 2011. Telephone numbers for the replay are 855-859-2056 (U.S./Canada) and 404-537-3406 (international), passcode 87169081.

About XOMA

XOMA is a leader in the discovery and development of novel antibody therapeutics. The company's proprietary product pipeline includes:

- XOMA 052, a potentially best-in-class antibody that binds to the inflammatory cytokine interleukin-1 beta, or IL-1 beta. Les Laboratoires Servier is XOMA's
 development and commercialization partner for XOMA 052. XOMA and Servier plan to enter XOMA 052 into Phase 3 clinical development for Behcet's uveitis, an
 orphan indication, and Phase 2 development for cardiovascular disease.
- XOMA 3AB, a novel combination of three antibodies in one product under development to prevent and treat botulism poisoning caused by exposure to botulinum neurotoxin Type A, among the most deadly bioterror threats. XOMA 3AB is in a Phase 1 clinical trial sponsored by the National Institute of Allergy and Infectious Diseases (NIAID) of the National Institutes of Health (NIH). XOMA receives funding for development of XOMA 3AB under NIAID Contract # HHSN26620060008C.

· A preclinical pipeline with candidates in development for autoimmune, cardio-metabolic, inflammatory and oncologic diseases.

XOMA has a premier antibody discovery and development platform that incorporates an unmatched collection of antibody phage display libraries and proprietary optimization and expression and manufacturing technologies that it uses for its own pipeline and in collaborations with pharmaceutical and biotechnology companies. XOMA technologies have contributed to the success of marketed antibody products including LUCENTIS® for wet age-related macular degeneration and CIMZIA® for rheumatoid arthritis and Crohn's disease. XOMA's fully integrated product development infrastructure extends from preclinical science to approval and is located in Berkeley, California. For more information, please visit www.xoma.com.

The XOMA Ltd. logo is available at www.globenewswire.com/newsroom/prs/?pkgid=5960

Forward-Looking Statements

Certain statements contained herein concerning anticipated levels of cash utilization, initiation of clinical trials, or interim or other results of early-stage clinical trials, or that otherwise relate to future periods are forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934. These statements are based on assumptions that may not prove accurate. Actual results could differ materially from those anticipated due to certain risks inherent in the biotechnology industry and for companies engaged in the development of new products in a regulated market.

Among other things, the anticipated levels of cash utilization may be other than as expected due to unanticipated changes in XOMA's research and development programs, unavailability of additional arrangements or higher than anticipated transaction costs; the initiation of clinical trials may be delayed or may never occur as a result of actions or inaction by our present or future collaboration partners, complications in the design, implementation or third-party approval of clinical trials or unanticipated safety issues; results of early-stage clinical trials may not be supported by later findings, larger trials and/or other actions required for regulatory approval may not be economically feasible, and final results of clinical trials may in any event not be consistent with preclinical or interim results.

These and other risks, including those related to the generally unstable nature of current economic and financial market conditions; the results of discovery and pre-clinical testing; the timing or results of pending and future clinical trials (including the design and progress of clinical trials; safety and efficacy of the products being tested; action, inaction or delay by the FDA, European or other regulators or their advisory bodies; and analysis or interpretation by, or submission to, these entities or others of scientific data); changes in the status of existing collaborative or licensing relationships; the ability of collaborators, licensees and other third parties to meet their obligations and their discretion in decision-making; XOMA's ability to meet the demands of the United States government agency with which it has entered into its government contracts; competition; market demand for products; scale-up, manufacturing and marketing capabilities; availability of additional licensing or collaboration opportunities; international operations; share price volatility; XOMA's financing needs and opportunities; uncertainties regarding the status of biotechnology patents; uncertainties as to the costs of protecting intellectual property; and risks carefully when considering XOMA's prospects.

** Tables Follow **

XOMA Ltd. Company and Investor Contact: Carol DeGuzman 510-204-7270 deguzman@xoma.com

Canale Communications Media Contact: Carolyn Hawley 619-849-5375 carolyn@canalecomm.com

XOMA Ltd. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (unaudited) (in thousands, except per share amounts)

	Three months ended June 30,			Six months ended June 30,				
		2011		2010		2011		2010
Revenues:								
License and collaborative fees	\$	6,039	\$	150	\$	11,866	\$	339
Contract and other revenue		10,467		5,481		20,128		12,292
Royalties		19		311		126		513
Total revenues		16,525		5,942		32,120		13,144
Operating expenses:								
Research and development		18,281		19,346		35,628		36,933
Selling, general and administrative		6,113		5,026		11,483		10,579
Total operating expenses		24,394		24,372		47,111		47,512
Loss from operations		(7,869)		(18,430)		(14,991)		(34,368)
Other income (expense):								
Investment and interest income		18		6		28		9
Interest expense		(634)		(90)		(1,166)		(177)
Other income (expense)		355		2,950		1,678		(2,813)
Net loss before taxes		(8,130)		(15,564)		(14,451)		(37,349)
Provision for income tax expense		-		(16)		(15)		(16)
Net loss	\$	(8,130)	\$	(15,548)	\$	(14,436)	\$	(37,333)
Basic and diluted net loss per common share	\$	(0.27)	\$	(0.93)	\$	(0.49)	\$	(2.24)
Shares used in computing basic and diluted net loss per common share		29,889		16,695		29,536		16,695

XOMA Ltd. CONDENSED CONSOLIDATED BALANCE SHEETS (in thousands)

	June 30, 2011 (unaudited)	December 31, 2010
ASSETS	(unautiteu)	
Current assets:		
Cash and cash equivalents	\$ 51,157	\$ 37,304
Trade and other receivables, net	11,059	20,864
Prepaid expenses and other current assets	807	712
Total current assets	63,023	58,880
Property and equipment, net	14,410	14,869
Other assets	2,165	503
Total assets	\$ 79,598	\$ 74,252
LIABILITIES AND SHAREHOLD	ERS' EQUITY	
Current liabilities:		
Accounts payable	\$ 5,043	\$ 3,581
Accrued liabilities	7,087	10,658
Deferred revenue	5,982	17,044
Warrant liabilities	1,395	4,245
Total current liabilities	19,507	35,528
Deferred revenue – long-term	8,665	1,086
Interest bearing obligation – long-term	27,031	13,694
Other long-term liabilities	257	353
Total liabilities	55,460	50,661
Shareholders' equity	24,138	23,591
Total liabilities and shareholders' equity	79,598	74,252